SECURITIES AND EXCHANGE COMMISSION

FORM 10-Q/A

Quarterly report pursuant to sections 13 or 15(d) [amend]

Filing Date: **2000-11-16** | Period of Report: **2000-09-30** SEC Accession No. 0000905895-00-500009

(HTML Version on secdatabase.com)

FILER

HOLLYWOOD ENTERTAINMENT CORP

CIK:905895| IRS No.: 930981138 | State of Incorp.:OR | Fiscal Year End: 1231

Type: 10-Q/A | Act: 34 | File No.: 000-21824 | Film No.: 771670

SIC: 7841 Video tape rental

Mailing Address 9275 SW PEYTON LANE WILSONVILLE OR 97070

Business Address 9275 SW PEYTON LANE WILSONVILLE OR 97070 5035701600

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q/A

X QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d)
OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended September 30, 2000

OR

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d)
OF THE SECURITIES EXHANGE ACT OF 1934

For the transition period from to

Commission File Number 0-21824

HOLLYWOOD ENTERTAINMENT CORPORATION (Exact name of registrant as specified in charter)

OREGON 93-0981138

(State or other jurisdiction of incorporation or organization)

(I.R.S. Employer Identification No.)

9275 S.W. Peyton Lane, Wilsonville, Oregon 97070 (Address of principal executive office, including zip code)

(503) 570-1600

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such short period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days.

Yes X No

As of November 9, 2000 there were 46,246,086 shares of the registrant's Common Stock outstanding.

HOLLYWOOD ENTERTAINMENT CORPORATION September 30, 2000

		Page
PART I.	FINANCIAL INFORMATION	3
Item 1.	Financial Statements	3
	Consolidated Statements of Operations	3
	Consolidated Balance Sheets	4
	Consolidated Statements of Cash Flows	5
	Notes to Consolidated Financial Statements	6
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	12
PART II.	OTHER INFORMATION	19
Item 6.	Exhibits and Reports on Form 8-K	19
Signatur	es	20

Hollywood Entertainment Corporation is filing this amendment to its report on Form 10-Q for the quarter ended September 30, 2000 because the financial statements and the other tables are illegible as a result of difficulties encountered in the report's electronic conversion for submission to the Commission via EDGAR. This amendment contains the same information as the report on Form 10-Q as originally filed but presents the contents in a readable format.

PART I. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

<TABLE> <CAPTION>

HOLLYWOOD ENTERTAINMENT CORPORATION CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited)

(In thousands, except per share amounts)

	ths Ended ber 30,	Nine Mont Septem	hs Ended ber 30,
2000	1999	2000	1999
<c></c>	<c></c>	<c></c>	<c></c>
\$262 , 077	\$221 , 292	\$796 , 429	\$648,235

Dago

REVENUE:

<S>

Rental revenue

Product sales	44,455	45,027	168,061	134,981
COCH OF CALEG	306,532	266,319	964,490	783,216
	-	67,440 33,702	148,074	97,022
		101,142	401,630	296,208
GROSS MARGIN	192,083	165,177	562,860	487,008
OPERATING COSTS AND EXPENSES:				
Operating and selling General and administrative Restructuring charge for closure				
of internet business Amortization of intangibles		- 14,388		
		164,161	611,760	466,895
INCOME (LOSS) FROM OPERATIONS	18,439	1,016	(48,900)	20,113
Non-operating income (expense): Interest income Interest expense		4 (11,999)		
Income (loss) before income taxes	1,204	(10,979)	(95 , 650)	(12,919)
Benefit (provision) for income taxes	(488)	(643)	21,159	(9,809)
Income (loss) before cumulative effec of a change in accounting principle		(11,622)	(74,491)	(22,728)
Cumulative effect of a change in accounting principle (net of income tax benefit of \$983)	-	_	-	(1,444)
NET INCOME (LOSS)		\$ (11,622)		
Net income (loss) per share before cumulative effect of a change in accounting principle				
Basic Diluted	\$0.02 \$0.02	\$(0.25) \$(0.25)		
Net income (loss) per share: Basic Diluted	\$0.02 \$0.02	\$(0.25) \$(0.25)		
Weighted average shares outstanding: Basic Diluted	46,207 46,483	45,743 45,743		

<TABLE> <CAPTION>

HOLLYWOOD ENTERTAINMENT CORPORATION CONSOLIDATED BALANCE SHEETS (In thousands, except share amounts)

	September 30,	December 31,
	2000	1999
<s> ASSETS</s>	(Unaudited) <c></c>	<c></c>
Current assets:		
Cash and cash equivalents Receivables Merchandise inventories Income tax receivable Prepaid expenses and other current assets		1,797 10,377
Total current assets	140,042	146,167
Rental inventory, net Property and equipment, net Goodwill, net Deferred income tax Other assets, net	398,712 102,852 79,551 24,495	52,691
	\$ 1,119,836	\$ 1,080,177 =======
LIABILITIES AND SHAREHOLDERS' EQUITY Current liabilities:		
Current maturities of long-term obligations Accounts payable Accrued expenses Accrued revenue sharing	148,099 66,114	\$ 14,493 111,578 64,566 9,402
Accrued interest	5,650	
Total current liabilities	407,059	211,904
Long-term obligations, less current portion Other liabilities	428,851 50,755	519,413 44,331
Shareholders' equity:	886,665 =====	775 , 648
Preferred stock, 19,500,000 shares authorized; no shares issued and outstanding Common stock, 100,000,000 shares authorized; 46,243,526 and 45,821,537 shares issued	-	-
and outstanding, respectively Retained deficit	365,440 (132,269)	362,307 (57,778)

233,171	304,529
\$ 1,119,836	\$ 1,080,177
========	========

</TABLE>

The accompanying notes are an integral part of this financial statement.

HOLLYWOOD ENTERTAINMENT CORPORATION CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited) (In thousands)

<TABLE> <CAPTION>

CAI I I ON Z	Nine Month Septembe	
	2000	1999
<s></s>	<c></c>	<c></c>
OPERATING ACTIVITIES:		
Net loss	\$ (74,491)	\$ (24,172)
Adjustments to reconcile net loss to cash provided by operating activities:		
Cumulative effect of a change in accounting		1 444
principle	160 707	1,444
Depreciation and amortization		162,683
Amortization of deferred financing costs		1,383
Change in deferred rent		2,286
Change in deferred income taxes	(22, 227)	(557)
Net asset write down for closure of internet		
business	40,087	-
Net change in operating assets and liabilities:		
Receivables	1,181	269
Merchandise inventories		(3,292)
Accounts payable		(23,331)
Accrued interest		(5 , 382)
Accrued revenue sharing		(2 , 888)
Other current assets and liabilities	516	2,242
Cash provided by operating activities	155,880	
INVESTING ACTIVITIES:		
Purchases of rental inventory, net	(130,331)	(125 , 855)
Purchases of property and equipment, net	(71,464)	(71,273)
Investment in businesses acquired	_	(15 , 976)
Increase in intangibles and other assets	(13,187)	(4,259)
Cash used in investing activities	(214,982)	(217,363)
FINANCING ACTIVITIES:		
Issuance of long-term obligations	12,511	50,000
Repayments of long-term obligations	(11,106)	

Tax benefit from exercise of stock options Proceeds from exercise of stock options Increase in revolving loan, net	63 782 60,000	5,151 4,056 55,000
Cash provided by financing activities	62,250	106,652
INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS	3,148	(26)
CASH AND CASH EQUIVALENTS AT BEGINNING OF YEAR	6 , 941	3 , 975
CASH AND CASH EQUIVALENTS AT END OF THE THIRD QUARTER	\$ 10,089 ======	\$ 3,949 =======
NON-CASH FINANCING ACTIVITIES Issuance of common stock as part of a legal settlement agreement	\$ 2,288	\$ -

</TABLE>

The accompanying notes are an integral part of this financial statement.

HOLLYWOOD ENTERTAINMENT CORPORATION

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

The unaudited consolidated financial statements have been prepared pursuant to the rules and regulations of the Securities and Exchange Commission. Certain information and note disclosures normally included in annual financial statements prepared in accordance with generally accepted accounting principles have been condensed or omitted pursuant to those rules and regulations, although the Company believes that the disclosures made are adequate to make the information presented not misleading. The information furnished reflects all adjustments which are, in the opinion of management, necessary for a fair statement of the results for the interim periods presented, and which are of a normal, recurring nature. These financial statements should be read in conjunction with the financial statements and the notes thereto included in the Company's annual report on Form 10-K for the year ended December 31, 1999, filed with the Securities and Exchange Commission.

(1) Accounting Policies

The consolidated financial statements included herein have been prepared in accordance with the accounting policies described in Note 1 to the December 31, 1999 audited consolidated financial statements included in the Company's annual report on Form 10-K. Certain prior year amounts have been reclassified to conform to the presentation used for the current year. These reclassifications had no impact to previously reported net loss or shareholders' equity.

(2) Statements of Changes in Shareholders' Equity

An analysis of the shareholders' equity amounts for the three quarters ended September 30, 2000 is as follows:

	Common Stock						
			Retained				
	Shares	Amount	Deficit	Total			
<\$>	<c></c>	<c> <c< td=""><td>> <c></c></td><td></td></c<></c>	> <c></c>				
(In thousands, except share amount	_						
Balance at December 31, 1999	45,821,537	\$ 362,307	\$ (57,778)	\$ 304,529			
Issuance of common stock under option plan	221,989	782		782			
Tax benefit from exercise of stock							
options		63		63			
Issuance of common stock as part							
of a legal settlement agreement	200,000	2,288		2,288			
Net loss			(74,491)	(74,491)			
Balance at September 30, 2000	46,243,526	\$ 365,440	\$(132,269)	\$ 233,171			
	========	=======	========	=======			

 | | | |

(3) Operating Leases

The Company leases all of its stores, corporate offices, distribution centers and zone offices under non-cancelable operating leases. All of the Company's stores have an initial operating lease term of five to fifteen years and most have options to renew for between five and fifteen additional years. Most operating leases require payment of property taxes, utilities, common area maintenance and insurance. Total rent expense, including related lease-required cost was \$62.0 million and \$181.3 million for the current year third quarter and three quarters, respectively, compared with \$51.0 million and \$146.8 million for the corresponding periods of the prior year.

(4) Long-term Obligations

The Company had the following long-term obligations as of September 30, 2000 and as of December 31, 1999 (in thousands).

<table></table>		
<\$>	<c></c>	<c></c>
	September 30,	December 31,
	2000	1999
Senior subordinated notes (1)	\$250 , 000	\$ 250,000
Borrowings under revolving credit facility	300,000	240,000
Obligations under capital leases	45,302	43,886
Other	9	20
	595,311	533,906
Current portions:	333,311	333,300
Credit facility	150,000	-
Capital leases	16,460	14,493

		166,460		14,493
Total long-term obligations				
net of current portion	\$	428,851	\$	519,413
	===	=======	==	======

</TABLE>

(1) 10.625% coupon payments are due semi-annually in February and August of each year

At September 30, 2000, maturities on long-term obligations for the next five years were as follows (in thousands):

<table></table>				
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>
			Capital	
Year Ending	Subordinated	Credit	Leases	
December, 31	Notes	Facility	& Other	Total
2000	\$ -	\$ 37 , 500	\$ 6 , 086	\$ 43,586
2001	_	150,000	14,721	164,721
2002	_	112,500	18,415	130,915
2003	_	_	6 , 089	6,089
2004	250 , 000	_	_	250,000
Thereafter	_	_	_	-
	\$ 250,000	\$ 300,000	\$ 45,311	\$ 595,311

(5) Earnings per Share

Basic earnings per share are calculated based on income available to common shareholders and the weighted-average number of common shares outstanding during the reported period. Diluted earnings per share includes additional dilution from the effect of potential issuances of common stock, such as stock issuable pursuant to the exercise of stock options, warrants outstanding and the conversion of debt.

The following tables are a reconciliation of the basic and diluted earnings per share computations:

<TABLE>

</TABLE>

Three Months Ended
September 30,
(In thousands, except per share amounts)

		2000			1999	
<s></s>	<c></c>	<c></c>	<c> Per Share</c>	<c></c>	<c></c>	<c> Per Share</c>
	Net Income	Shares	Amounts	Loss	Shares	Amounts

Basic income (loss)

<pre>per share: Effect of dilutive securities:</pre>	\$	716	46,207	\$	0.02	\$(11,622)	45,743	\$ (0.25)
Stock options		-	276			_	-	
Diluted income (loss)								
per share:	\$	716	46,483	\$	0.02	\$(11,622)	45,743	\$ (0.25)
	===	=====	======	===	=====	=======	======	=======

</TABLE>

Due to the Company's net loss in 1999, stock options accounted for using the treasury stock method would be antidilutive. Accordingly, 1.5 million shares

have been excluded from the diluted net loss per share calculation.

<TABLE> <CAPTION>

Nine Months Ended September 30, (In thousands, except per share amounts)

		2000			1999	
<\$>	<c></c>	<c></c>	<c> Per Share</c>	<c></c>	<c></c>	<c> Per Share</c>
	Loss	Shares	Amounts	Loss	Shares	Amounts
Basic loss per share: Effect of dilutive securities:	\$ (74,491)	46,119	\$ (1.62)	\$(24,172)	45,522	\$ (0.53)
Stock options	-	_		_	-	
Diluted loss per share	e: \$(74,491)	46,119	\$ (1.62)	\$(24,172)	45,522	\$ (0.53)

</TABLE>

Due to the Company's net loss, stock options accounted for using the treasury Stock method would be antidilutive. Accordingly, .3 million shares and 2.3 million shares have been excluded from the diluted net loss per share calculation for the nine months ended September 30, 2000 and 1999, respectively.

(6) Store Preopening Cost

In April 1998, SOP 98-5, "Reporting on the Cost of Start-up Activities," was finalized, which required that cost incurred for start-up activities, such as store openings, be expensed as incurred. The Company adopted SOP 98-5 effective January 1, 1999. The cumulative effect of the change was to increase net loss in the first quarter of 1999 by \$1.4 million, net of tax benefit.

(7) Segment Reporting

In June 1997, the Financial Accounting Standards Board (FASB) issued Statement No. 131, "Disclosures about Segments of an Enterprise and Related Information," effective for fiscal years beginning after December 15, 1997. The Company adopted Statement No. 131 in 1998.

The Company identifies its segments based on management responsibility. The Hollywood Video segment, which consists of the Company's 1,796 retail stores located in 47 states was the only segment operated by the Company in the current year third quarter. The Company operated two segments during the first six months of the current year, the Hollywood Video segment and the Reel.com segment, primarily an e-commerce company specializing in movies. During the current year second quarter, the Company announced the discontinuation of e-commerce operations at Reel.com. This resulted in a restructuring charge detailed below in Note 8. All assets of Reel.com were transferred to the Hollywood Video segment on June 12, 2000. The Company measures segment profit as operating profit, which is defined as income (loss) before interest expense and income taxes. Information on segments and a reconciliation to operating income (loss) are as follows (in thousands):

<TABLE>

Three Months Ended September 30,

		2000			1999	
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>	<c></c>
	Hollywood			Hollywoo	d	
	Video	Reel.com	Total	Video	Reel.com	Total
Revenue	\$306 , 532		306,532	\$256,201	\$ 10,118	\$ 266,319
	•	Y	•	•	•	•
Tape amortization	30,733		30,733	22,231	118	22,349
Other depreciation						
and amortization	20,187		20,187	17,408	13,446	30 , 854
Gross Margin	192,083		192,083	164,606	571	165,177
Operating income						
(loss) *	18,439		18,439	23,703	(22,687)	1,016
Interest expense,						
net	17,235		17,235	10,733	1,262	11,995
Total assets	1,119,836	1,	119,836	933 , 316	64,609	997 , 925
Purchases of						
property and						
equipment, net	15 , 727		15 , 727	23,741	2,060	25,801

</TABLE>

<TABLE> <CAPTION>

^{*}Reel.com's operating loss for the three months ended September 30, 1999 includes \$12.5 million in goodwill amortization. Excluding goodwill amortization, Reel.com's operating loss would have been \$10.2 million.

Nine Months Ended September 30,

		2000			1999	
<s></s>	<c> Hollywood</c>	<c></c>	<c></c>	<c> Hollywood</c>	<c></c>	<c></c>
	Video	Reel.com	Total	Video	Reel.com	Total
Revenue	\$938,210	\$ 26,280	\$964,490	\$758 , 352	\$ 24,864	\$ 783,216
Tape amortization	88,621	110	88,731	76 , 668	354	77,022
Other depreciation						
and amortization	58,438	23,492	81,930	48 , 277		·-
Gross Margin	581 , 834	(18 , 974)	562 , 860	484,995	2,013	487,008
Operating income						
(loss) *	73 , 869	(122,769)	(48,900)	84,027	(63,914)	20,113
Interest expense,						
net	42,484	4,266	46,750	30,314	2,720	33,034
Total assets	1,119,836	- 1	1,119,836	933,316	64,609	997 , 925
Purchases of property and						
equipment, net	69,946	1,518	71,464	67 , 124	4,149	71,273

</TABLE>

*Reel.com's operating loss includes a restructuring charge of \$69.3 million for discontinued e-commerce operations. The loss also includes \$22.7 million in goodwill amortization. Excluding the restructuring charge and goodwill amortization, Reel.com's operating loss would have been \$30.8 million for the six months ended June 30, 2000. The Company did not operate Reel.com as a segment in the current year third quarter. The operating loss for the nine months ended September 30, 1999 would have been \$26.3 million excluding \$37.6 million in goodwill amortization.

(8) Reel.com Discontinued E-commerce Operations

On June 12, 2000, the Company announced that it would close down the e-commerce business at Reel.com. The Company developed a leading web-site over the seven quarters since Reel.com was purchased in October of 1998, but its business model of rapid customer acquisition led to large operating losses and required significant cash funding. Due to market conditions, the Company was unable to obtain outside financing for Reel.com, and could not justify continued funding from its video store cash flow. On June 13, 2000, the Company terminated employment of approximately 200 of Reel.com's 240 employees, and paid \$1.9 million in involuntary termination benefits. The remaining employees have since been terminated or integrated into Hollywood Entertainment.

The Company plans to maintain the web-site as a content-only site to minimize any negative effect the Reel.com shutdown may have on existing Hollywood Video store customers for a period of time not expected to exceed twelve months from the e-commerce shutdown. During this time, point of purchase materials promoting Reel.com as an e-commerce destination will need to be removed from the stores. To offset the costs of maintaining the web-

site during this period, the Company has entered into an agreement with Buy.com to direct Reel.com visitors to Buy.com to make purchases. Revenues associated with the Buy.com agreement and the expenses of maintaining the web-site will be recognized as earned and incurred, respectively.

As a result of the discontinuation of e-commerce operations, the Company recorded a total charge of \$69.3 million in the current year second quarter, of which \$48.5 million was classified as a restructuring charge on the consolidated statement of operations and \$20.8 million was included in cost of product sales.

The restructuring charge line item includes \$1.9 million of severance and benefits paid on June 13, 2000, \$19.3 million of asset write downs, and \$27.3 million of accrued liabilities. The assets written down include the remaining \$14.9 million of goodwill associated with the acquisition of Reel.com, and \$4.4 million to write down equipment, leasehold improvements, prepaid expenses and accounts receivable to their net realizable values. Amounts accrued include \$21.5 million for contractual obligations, including lease commitments and anticipated legal claims against the Company, and \$5.8 million for legal, financial, and other professional services incurred as a direct result of the closure of Reel.com.

The Company used some of the equipment from Reel.com at the new distribution center in Nashville, Tennessee, and at the corporate offices in Wilsonville, Oregon. Equipment not utilized by the Company was sold. Proceeds of \$250,000 were received in the current year third quarter.

Most of Reel.com's business agreements have been terminated under the provisions of each agreement. Many of these agreements contain minimum guarantee provisions, some of which cannot be waived by terminating the agreement. The Company paid \$0.5 million of the \$21.5 million total accrual in the current year third quarter and anticipates paying between \$0.5 million and \$2.0 million in the current year fourth quarter with the remaining amount paid some time in 2001.

Charged to cost of goods sold was the write down of Reel.com inventory, primarily DVD's, to net realizable value. This represents excess product for the Hollywood Video segment because the stores are currently fully stocked for both rental and sell through. The Company plans to liquidate this inventory by the fourth quarter of 2001 through special tent sales or sidewalk sales at select stores, vendor returns, grand opening promotions, and outlet centers.

(9) Related Party Transactions

On May 8, 2000, the Board of Directors approved a long-term compensation package for the Company's Chief Executive Officer, which included a \$15 million loan. The five year loan bears interest at an annual rate equal to the greater of ten percent or the minimum rate required under the Internal Revenue Code and regulations for loans to affiliated persons, and is due in five annual installments of \$3 million beginning on June 1, 2001.

ITEM 2 MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

RESULTS OF OPERATIONS

Summary Results of Operations

The Company's net income for the current year third quarter was \$.7 million compared with a net loss of \$11.6 million in the third quarter of 1999. The improvement in net income was the result of closing down e-commerce operations at Reel.com. The Company's net loss for the current year three quarters was \$74.5 million compared with a net loss before cumulative effect of a change in accounting principle of \$22.7 million in the corresponding period of the prior year. The increase in net loss was primarily due to a restructuring charge for discontinued e-commerce operations of Reel.com recorded in the current year second quarter.

The following table sets forth (i) selected results of operations data, expressed as a percentage of total revenue; (ii) other financial data; and (iii) selected operating data.

<TABLE>

			Nine Months Ended September 30,		
		1999	2000		
	(Unaud	ited)	(Unaudi	ted)	
<s> REVENUE:</s>	<c></c>	<c></c>	<c></c>	<c></c>	
Rental revenue Product sales		16.9	82.6% 17.4		
		100.0	100.0		
GROSS MARGIN	62.7	62.0	58.4	62.2	
OPERATING COSTS AND EXPENSES: Operating and selling General and administrative			49.8		
Restructuring charge for closure of internet business	-	_	5.0	_	
Amortization of intangibles	0.6	5.3	2.9	5.5	
	56.7	61.6	63.4	59.6 	
INCOME (LOSS) FROM OPERATIONS Non-operating income (expense),	6.0	0.4	(5.0)	2.6	
net	(5.6)		(4.9)	(4.2)	

Income (loss) before income

taxes	0.4	(4.1)	(9.9)	(1.6)
Benefit (provision) for income taxes	(0.2)	(0.3)	2.2	(1.3)
Income (loss) before cumulative effect of a change in accounting				
<pre>principle Cumulative effect of a change in accounting principle</pre>	0.2	(4.4)	(7.7)	(2.9)
NET INCOME (LOSS)	0.2%	(4.4)%	(7.7)%	
NET TROOTE (LOSS)				
OTHER FINANCIAL DATA:				
Rental gross margin (1) Product gross margin (2)		69.5% 25.2%		69.3% 28.1%
EBITDA (3) (In thousands) Hollywood Video segment EBITDA Reconciliation to Adjusted EBITDA (4)	\$ 68,681	\$ 62,806	\$218 , 991	\$206,174
Add special charges (5) Add non-cash expenses(6)	- 15,924	2,335 14,143		3,779 37,821
Less existing store investment in new release inventory (7)				
ADJUSTED EBITDA HOLLYWOOD VIDEO	40,014	46,669	142,484	151,545
Reel.com segment EBITDA (8)	_	(9,124)	(29,834)	(24,792)
CONSOLIDATED ADJUSTED EBITDA	\$ 40,014	\$ 37,545	\$112 , 650	\$126 , 753
CASH FLOW FROM: (In thousands) Operating activities Investing activities Financing activities	\$ 54,890 (56,093) 6,921	\$ 49,814 (71,282) 19,269	\$155,880 (214,982) 62,250	
OPERATING DATA:	1 706	1 474	1 706	1 474
Number of stores at quarter end Weighted average stores open	1,796	1,474	1,796	1,474
during the period Comparable store revenue	1,786	1,431	1,729	1,360
increase (9)	1%	11%	4%	16%

</TABLE>

- (1) Rental gross margin as a percentage of rental revenue.
- (2) Product gross margin as a percentage of product revenue.
- (3) EBITDA consists of operating income before interest, tax, depreciation and amortization. EBITDA should not be viewed as a measure of financial performance under Generally Accepted Accounting Principles (GAAP) or as a substitute for GAAP measurements such as net income or cash flow from operations. This calculation of EBITDA is not necessarily comparable to

reported EBITDA of other companies due to the lack of a uniform definition of EBITDA.

- (4) Adjusted EBITDA represents income from operations before depreciation and amortization plus non-cash expenses that reduce EBITDA, less the cost of acquiring new release rental inventory which is capitalized. Adjusted EBITDA should not be viewed as a measure of financial performance under GAAP and should not be considered as an alternative to cash flows from operating activities (as determined in accordance with GAAP) as a measure of liquidity. Our calculation of adjusted EBITDA is not necessarily comparable to amounts reported by other companies due to the lack of a uniform definition of EBITDA.
- (5) The special charge for the prior year third quarter relates to the settlement between the Company and Twentieth Century Fox Home Entertainment ("Fox"). The prior year three quarters included the settlement with Fox and a cumulative effect of a change in accounting principle resulting from the adoption of SOP 98-5.
- (6) Expenses which were non-cash in nature including tape loss and the book cost of previously viewed movies sold.
- (7) This represents existing store purchases of new release tapes, games and DVD's that are considered investing activities on the statement of cash flows. This is in addition to the cost of product represented by revenue sharing expense already included in the statement of operations.
- (8) Excludes the restructuring charge of \$69.3 million for the discontinuation of e-commerce operations.
- (9) A store is comparable after it has been open and owned by the Company for 12 full months. An acquired store converted to the Hollywood Video name and store design is removed from the comparable store base when the conversion process is initiated and returned 12 full months after reopening.

REVENUE

Revenue increased by \$40.2 million, or 15%, in the current year third quarter and \$181.3 million, or 23%, in the current year three quarters compared with the corresponding periods of the prior year. The increase was primarily due to the addition of 360 new superstores in the twelve months ended September 30, 2000. An increase in comparable store revenue of 1% and 4% in the current year third quarter and three quarters, respectively, also favorably impacted revenue.

GROSS MARGIN

Rental margin as a percentage of rental revenue was 68.4% and 68.2% in the current year third quarter and three quarters compared to 69.5% and 69.3% in the corresponding periods of the prior year. The decrease in margin primarily relates to a decrease in new release videocassettes purchased at sell through pricing. Prior to revenue sharing, studios had two primary options for pricing new release titles. The first, rental pricing, was to price each cassette at approximately \$65 and focus sales to video rental

outlets. The second, sell through pricing, was to price each cassette under \$20 and sell to all distributors of videocassette movies. The biggest hits each year were priced for sell through and generally came with a substantial marketing push from the studio. The low unit price, and studio advertising, made these types of purchases the Company's best performers in terms of margin, including revenue sharing purchases. Over the last two years, as revenue sharing has gained acceptance, the percentage of titles released at sell through pricing has substantially decreased.

Product margin as a percentage of product sales was 28.6% and 11.9% in the current year third quarter and three quarters compared to 25.2% and 28.1% in the corresponding periods of the prior year. The increase in margin in the current year third quarter is primarily due to Reel.com's discontinuation of e-commerce sales, which were at a significantly lower margin than the Hollywood Video stores. The decrease in margin in the current year three quarters is primarily due to a \$20.8 million write down of inventory associated with the restructuring of Reel.com in the current year second quarter.

OPERATING COSTS AND EXPENSES

Operating and Selling

Total operating expenses for the current year third quarter and three quarters increased by \$23 million and \$107 million, respectively, when compared to the prior year corresponding periods. Operating and selling expenses increased as a percentage of total revenue to 50.9% and 49.8% in the current year third quarter and three quarters ended, respectively, compared to 49.9% and 47.7% in the corresponding periods of the prior year primarily due to an increase in advertising expenses. For the current year third quarter, the increase in total operating expenses resulted from an increase in store labor cost of \$10 million, occupancy costs of \$11 million, other store operating expenses of \$11 million, offset by the closure of Reel.com which had operating expenses for the prior year third quarter of \$9 million. For the current year three quarters, the increase in total operating expenses resulted from an increase in store labor cost of \$36 million, occupancy costs of \$34 million, other store operating expenses of \$30 million and an increase in Reel.com operating costs of \$6 million. With the exception of Reel.com, operating expenses increased due to the growth in the number of superstores operated by the Company.

General and Administrative

General and administrative expenses in the current year third quarter decreased \$1.1 million to \$15.8 million due to the closure of the e-commerce business at Reel.com. The general and administrative expenses for the current year three quarters increased \$4.2 million to \$54.4 million. The increase was primarily due to higher payroll and related costs partially off set by the closure of the e-commerce business at Reel.com.

Restructuring Charge for Closure of Internet Business

On June 12, 2000, the Company announced that it would close down the e-commerce business at Reel.com. The Company developed a leading web-site over the seven quarters since Reel.com was purchased in October of 1998, but its

business model of rapid customer acquisition led to large operating losses and required significant cash funding. Due to market conditions, the Company was unable to obtain outside financing for Reel.com, and could not justify continued funding from its video store cash flow.

The Company plans to maintain the web-site as a content-only site to minimize any negative effect the Reel.com shutdown may have on existing Hollywood Video store customers for a period of time not expected to exceed twelve months from the e-commerce shutdown. During this time, point of purchase materials promoting Reel.com as an e-commerce destination will need to be removed from the stores. To offset the costs of maintaining the web-site during this period, the Company has entered into an agreement with Buy.com to direct Reel.com visitors to Buy.com to make purchases. Revenues associated with the Buy.com agreement and the expenses of maintaining the web-site will be recognized as earned and incurred, respectively.

The restructuring charge of \$48.5 million includes \$1.9 million of severance and benefits paid on June 13, 2000, \$19.3 million of asset write downs, and \$27.3 million of accrued liabilities. The assets written down include the remaining \$14.9 million of goodwill associated with the acquisition of Reel.com, and \$4.4 million to write down equipment, leasehold improvements, prepaid expenses and accounts receivable to their net realizable values. Amounts accrued include \$21.5 million for contractual obligations, including lease commitments and anticipated legal claims against the Company, and \$5.8 million for legal, financial, and other professional services incurred as a direct result of the closure of Reel.com.

The Company used some of the equipment from Reel.com at the new distribution center in Nashville, Tennessee, and at the corporate offices in Wilsonville, Oregon. Equipment not utilized by the Company was sold. Proceeds of \$250,000 were received in the current year third quarter.

Most of Reel.com's business agreements have been terminated under the provisions of each agreement. Many of these agreements contain minimum guarantee provisions, some of which cannot be waived by terminating the agreement. The Company paid \$0.5 million of the \$21.5 million total accrual in the current year third quarter and anticipates paying between \$0.5 million and \$2.0 million in the current year fourth quarter with the remaining amount paid some time in 2001.

Amortization of Intangibles

Amortization of intangibles decreased to \$1.8 million and \$28 million in the current year third quarter and three quarters from \$14 million and \$43 million in the corresponding periods of the prior year. The decrease was caused by a write down of Reel.com's goodwill as e-commerce operations discontinued in the current year second quarter. The goodwill write down of \$14.9 million was included in the restructuring charge for closure of Internet business line item on the Company's consolidated statement of operations.

NONOPERATING INCOME (EXPENSE), NET

Interest expense, net of interest income increased by \$5.2 million and \$13.7

million in the third quarter and current year three quarters compared to the corresponding periods of the prior year due to increased levels of borrowings under the Company's revolving credit facility.

INCOME TAXES

The provision (benefit) for income taxes for the third quarter and first nine months of 2000 results in effective tax rates of 40.5% and (22.12)% of pretax income, as compared to 5.9% and 75.9% for the same calendar periods of 1999, respectively. The tax provision for the first nine months of 2000 decreased compared to the prior year due to a decrease in taxable income. The increase in the effective tax rate for the third quarter compared to the prior year is due to a reduction in net operating loss associated with Reel.com.

LIQUIDITY AND CAPITAL RESOURCES

The amount of cash generated from operations in the nine months ending September 30 exceeded the Company's current debt service requirements. The capital expenditures (including purchases of rental inventory) of the Company are primarily funded by the excess operating cash flow and through loans under a revolving line of credit.

At September 30, 2000, the Company had cash and cash equivalents of \$10.1 million and a working capital deficit of \$267 million. Rental inventories are accounted for as non-current assets under generally accepted accounting principles because they are not assets which are reasonably expected to be completely realized in cash or sold in the normal business cycle. Although the rental of this inventory generates a substantial portion of the Company's revenue, the classification of these assets as non-current excludes them from the computation of working capital. The acquisition cost of rental inventories, however, is reported as a current liability until paid and, accordingly, included in the computation of working capital. Consequently, the Company believes working capital is not as significant a measure of financial condition for companies in the video retail industry as it is for companies in other industries. Because of the accounting treatment of rental inventory as a non-current asset, the Company may, from time to time, operate with a working capital deficit.

Cash Provided by Operating Activities

Net cash provided by operating activities increased by \$45.2 million in the current year three quarters compared with the corresponding period of the prior year. This increase was primarily due to an increase in accounts payable and accrued liabilities, which are sources of operating cash. A significant part of the increase was related to Reel.com.

Cash Used in Investing Activities

Net cash used in investing activities decreased by \$2.4 million in the current year three quarters compared with the corresponding period of the prior year. The decrease is primarily due to fewer new store openings offset by a higher reinvestment of revenue in new release movies and an increase in other assets.

Net cash provided by financing activities decreased by \$44.4 million in the current year three quarters compared with the corresponding period of the prior year. The decrease was primarily due to a decrease in the issuance of long-term obligations as the Company begins to open fewer stores and fund them through operating cash flow instead of debt financing. The Company has the availability of up to \$300 million in revolving credit loans. The Company may utilize the revolving credit facility as needed for working capital, capital expenditures and general corporate purposes. The total commitment under this facility will begin its scheduled reduction of \$37.5 million per quarter beginning on December 5, 2000 which continues until the revolving credit facility's maturity in September 2002. The Company expects to meet scheduled commitment reductions through debt pay-down from internally generated cash flow from operations and a slowdown in capital expenditures and investing activities. The Company may seek additional financing as a source of incremental liquidity as necessary. As of September 30, 2000, \$300 million was outstanding under the revolving credit agreement with \$10.1 million in cash and cash equivalents on hand.

Capital Expenditures

The Company's capital expenditures include product for stores, store equipment and fixtures, remodeling a certain number of existing stores, implementing and upgrading office, warehouse and store technology and opening for new store locations. Each new store opening requires initial capital expenditures, including leasehold improvements, inventory, equipment and costs related to site location, lease negotiations and construction permits. We currently anticipate that capital expenditures based on opening 215 stores to be approximately \$110 million in 2000, of which \$95 million is anticipated to relate to new, relocated, remodeled stores and game departments. The remaining balance relates to corporate capital expenditures. We expect the total investment per new store to approximate \$420,000 this year, which includes rental and merchandise inventory, leasehold improvement, signage, furniture, fixtures and equipment. However, the cost of opening a new store can vary based on size, construction costs in a particular market, and other factors. These capital expenditures will be funded primarily by cash generated by operations, supplemented by the availability of the revolving line of credit or other forms of equipment financing and/or leasing if necessary. The Company expects to open between 50 and 80 stores in 2001, based upon the availability of capital resources coupled with the Company's desire to reduce outstanding indebtness.

PART II OTHER INFORMATION

ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K

- (a) Exhibit 10.1 Amended Revolving Credit Agreement
- (b) Exhibit 10.2 Change of Control Plan for Senior Management
- (c) Exhibit 27.1 Financial data schedule (electronic filing only)

(d)	Reports	on	Form	8-K
-----	---------	----	------	-----

None filed during the period.

HOLLYWOOD ENTERTAINMENT CORPORATION

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

HOLLYWOOD ENTERTAINMENT CORPORATION

(Registrant)

November 16, 2000 /S/David G. Martin

(Date) David G. Martin

Executive Vice President and Chief Financial Officer (Authorized Officer and Principal Financial and Accounting Officer of the Registrant)

REVOLVING CREDIT AGREEMENT as amended as of August 4, 2000

among

HOLLYWOOD ENTERTAINMENT CORPORATION, as Borrower,

and

SOCIETE GENERALE
GOLDMAN SACHS CREDIT PARTNERS L.P.

and

the other financial institutions identified on the signature pages hereto, as Lenders,

and

SOCIETE GENERALE, as Agent for the Lenders,

GOLDMAN SACHS CREDIT PARTNERS L.P., as Documentation Agent, and

CREDIT LYONNAIS LOS ANGELES BRANCH, DEUTSCHE BANK AG, NEW YORK BRANCH, U.S. BANK NATIONAL ASSOCIATION,

and

KEYBANK NATIONAL ASSOCIATION, as Co-Agents

Arranged by Societe Generale Goldman Sachs Credit Partners L.P.

TABLE OF CONTENTS

Page

1.1. Certain Defined Terms 2	
1.2. General Principles 22	
1.3. Accounting Terms 23	
ARTICLE II. THE LOANS 23	
2.1. The Facility 23	
2.2. Manner of Borrowing a Loan 24	
2.3. Reductions of Commitments 26	
2.4. Repayment of Principal 27	
2.5. Agent's Right to Fund 28	
2.6. Interest on Loans 29	
2.7. Compensation for Increased Costs 32	
2.8. Prepayments 34	
2.9. Notes 34	
2.10. Manner of Payments 34	
2.11. Fees 35	
2.12. Sharing of Payments, Etc 37	
2.13. Taxes 37	
2.14. Certain Acknowledgments by Borrower 40	
2.15. Designated Senior Indebtedness 40	
ARTICLE III. CONDITIONS OF LENDING 40	
3.1. Conditions to Effectiveness Availability	40
3.2. Conditions Precedent to all Loans 42	
ARTICLE IV. REPRESENTATIONS AND WARRANTIES	437
4.1. Corporate Existence and Power 43	
4.2. Corporate Authorization 44	
4.3. Government Approvals, Etc 44	
4.4. Binding Obligations, Etc 44	
4.4. Binding Obligations, Etc 44 4.5. Litigation 44	
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448	
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45	
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39	
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46	
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46	
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46	
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470	
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470 4.13. Patents, Licenses, Franchises 48	
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470 4.13. Patents, Licenses, Franchises 48 4.14. New Investment Company, Etc 481	
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470 4.13. Patents, Licenses, Franchises 48 4.14. New Investment Company, Etc 481 4.15. Insurance 481	
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470 4.13. Patents, Licenses, Franchises 48 4.14. New Investment Company, Etc 481 4.15. Insurance 481 4.16. Certain Representations Regarding Subsidiaries	48
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470 4.13. Patents, Licenses, Franchises 48 4.14. New Investment Company, Etc 481 4.15. Insurance 481 4.16. Certain Representations Regarding Subsidiaries 4.17. Subsidiaries 48	48
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470 4.13. Patents, Licenses, Franchises 48 4.14. New Investment Company, Etc 481 4.15. Insurance 481 4.16. Certain Representations Regarding Subsidiaries 4.17. Subsidiaries 48 4.18. Hedging Agreements 48	48
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470 4.13. Patents, Licenses, Franchises 48 4.14. New Investment Company, Etc 481 4.15. Insurance 481 4.16. Certain Representations Regarding Subsidiaries 4.17. Subsidiaries 48 4.18. Hedging Agreements 48 4.19. Lessors 48	48
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470 4.13. Patents, Licenses, Franchises 48 4.14. New Investment Company, Etc 481 4.15. Insurance 481 4.16. Certain Representations Regarding Subsidiaries 4.17. Subsidiaries 48 4.18. Hedging Agreements 48 4.19. Lessors 48 4.20. Representations as a Whole 48	48
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470 4.13. Patents, Licenses, Franchises 48 4.14. New Investment Company, Etc 481 4.15. Insurance 481 4.16. Certain Representations Regarding Subsidiaries 4.17. Subsidiaries 48 4.18. Hedging Agreements 48 4.19. Lessors 48 4.20. Representations as a Whole 48 ARTICLE V. AFFIRMATIVE COVENANTS 492	48
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470 4.13. Patents, Licenses, Franchises 48 4.14. New Investment Company, Etc 481 4.15. Insurance 481 4.16. Certain Representations Regarding Subsidiaries 4.17. Subsidiaries 48 4.18. Hedging Agreements 48 4.19. Lessors 48 4.20. Representations as a Whole 48 ARTICLE V. AFFIRMATIVE COVENANTS 492 5.1. Use of Proceeds 492	48
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470 4.13. Patents, Licenses, Franchises 48 4.14. New Investment Company, Etc 481 4.15. Insurance 481 4.16. Certain Representations Regarding Subsidiaries 4.17. Subsidiaries 48 4.18. Hedging Agreements 48 4.19. Lessors 48 4.20. Representations as a Whole 48 ARTICLE V. AFFIRMATIVE COVENANTS 492 5.1. Use of Proceeds 492 5.2. Payment 492	
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470 4.13. Patents, Licenses, Franchises 48 4.14. New Investment Company, Etc 481 4.15. Insurance 481 4.16. Certain Representations Regarding Subsidiaries 4.17. Subsidiaries 48 4.18. Hedging Agreements 48 4.19. Lessors 48 4.20. Representations as a Whole 48 ARTICLE V. AFFIRMATIVE COVENANTS 492 5.1. Use of Proceeds 492 5.2. Payment 492 5.3. Preservation of Corporate Existence, Etc	48
4.4. Binding Obligations, Etc 44 4.5. Litigation 44 4.6. Financial Condition 448 4.7. Indebtedness; Title and Liens 45 4.8. Taxes 39 4.9. Laws, Orders; Other Agreements 46 4.10. Lien Priority 46 4.11. Federal Reserve Regulations 46 4.12. ERISA 470 4.13. Patents, Licenses, Franchises 48 4.14. New Investment Company, Etc 481 4.15. Insurance 481 4.16. Certain Representations Regarding Subsidiaries 4.17. Subsidiaries 48 4.18. Hedging Agreements 48 4.19. Lessors 48 4.20. Representations as a Whole 48 ARTICLE V. AFFIRMATIVE COVENANTS 492 5.1. Use of Proceeds 492 5.2. Payment 492	

5.7. 5.8. 5.9.	Maintenance of Property, Etc 50 Compliance With Laws, Etc 503 Other Obligations 50 Insurance 51
5.11.	Financial Information 535 Notification 55 Blocked Account Agreement 56
5.13.	-
5.14. 5.15	Independent Board of Directors and Committees 568 Tangible Net Worth 56
	Leverage Ratio 49
5.17.	
5.18.	Interest and Rent Coverage Ratio 58
	Average Per Store Contribution to Profit 59
	Adjusted Leverage Ratio 591
	Dissolution of Subsidiary 602
	Officer's Certificate 602 E VI. NEGATIVE COVENANTS 60
	Liquidation, Merger, Sale of Assets, Fundamental Change 61
	Indebtedness 613
	Guaranties, Etc 63
	Liens 63
6.5.	Location of Collateral 645
	Investments 64
	Limitations on Prepayment or Amendment of Subordinated Debt 646
	ERISA Compliance 656
	Transactions with Affiliates 65
	Change in Business 65 Accounting Change 667
	Capital Expenditures 667
	Stores 67
	E VII. EVENTS OF DEFAULT 678
7.1.	Events of Default 678
7.2.	Consequences of Default 711
ARTICI	E VIII. THE AGENT 722
	Authorization and Action 722
	Duties and Obligations 734
	Dealings Between Agent and Borrower 75
	Lender Credit Decision 755
	Indemnification by Lenders 76 Successor Agent 766
	Documentation Agent; Co-Agents 77
	E IX. MISCELLANEOUS 777
	No Waiver; Remedies Cumulative 777
	GOVERNING LAW 787
9.3.	Notices 787
9.4.	Assignment and Participations 788
	Indemnification by Borrower; Expenses 83
	Setoff 853
9.7.	No Third Party Beneficiaries 853

- 9.8. Severability; Independence of Covenants 85
 - 9.9. Survival 864
 - 9.10. Release 864
- 9.11. Limitation on Liability 875
- 9.12. Entire Agreement; Amendment 875
- 9.13. WAIVER OF JURY TRIAL 885
- 9.14. Headings 88
- 9.15. Counterparts; Telecopied Signatures 886
- 9.16. FORUM SELECTION AND CONSENT TO JURISDICTION 886

EXHIBITS

Exhibit A - Form of Revolving Promissory Note
Exhibit B - Form of Assignment Agreement

SCHEDULE

Schedule 1 - Breakage Fee

Schedule 3.1.4 - Capital Leases

Schedule 4.17 - Subsidiary Assets

Schedule 6.2(a) - Funded Debt

Schedule 6.2(d) - Capital Leases

REVOLVING CREDIT AGREEMENT

THIS REVOLVING CREDIT AGREEMENT (this "Agreement") is the Existing Credit Agreement (as defined below), as amended as of August 4, 2000, by and among (i) HOLLYWOOD ENTERTAINMENT CORPORATION, (d/b/a "Hollywood Video"), an Oregon corporation (the "Borrower"), (ii) SOCIETE GENERALE, GOLDMAN SACHS CREDIT PARTNERS L.P. and the other financial institutions identified as Lenders on the signature pages hereto (each individually a "Lender" and collectively the "Lenders"), and (iii) SOCIETE GENERALE, as agent for the Lenders (in that capacity, the "Agent") and GOLDMAN SACHS CREDIT PARTNERS L.P., as documentation agent (in that capacity, the "Documentation Agent"), and CREDIT LYONNAIS LOS ANGELES BRANCH, DEUTSCHE BANK AG, NEW YORK BRANCH, U.S. BANK NATIONAL ASSOCIATION, and KEYBANK NATIONAL ASSOCIATION, as Co-Agents (in that capacity, each individually a "Co-Agent" and collectively the "Co-Agents").

WITNESSETH:

WHEREAS, pursuant to that certain Revolving Credit
Agreement dated as of September 5, 1997 (as amended, supplemented or
otherwise modified prior to the date hereof by (a) the First
Amendment Agreement, dated as of March 31, 1998, (b) the Second
Amendment Agreement, dated as of March 1, 1999, (c) the Consent and

Waiver, dated May 18, 1999, (d) the Fourth Amendment Agreement and Consent, dated as of November 22, 1999, (e) the Fifth Amendment Agreement and Consent, dated as of March 22, 2000, (f) the Waiver Letter, dated June 8, 2000, (g) the Waiver Extension Letter, dated June 20, 2000 and (h) the Waiver Extension Letter, dated July 28, 2000, the "Existing Credit Agreement"), by and among (i) the Borrower, (ii) SOCIETE GENERALE, GOLDMAN SACHS CREDIT PARTNERS L.P., and the Lenders, and (iii) the Agent and the Documentation Agent, the Lenders agreed to provide the Borrower with a revolving credit facility (the "Existing Facility");

WHEREAS, the Borrower has requested that the Agent and the Lenders amend the Existing Credit Agreement, and the Agent and the Lenders are willing to amend the Existing Credit Agreement as set forth herein;

NOW, THEREFORE, the parties hereto hereby agree as follows:

ARTICLE I.

DEFINITIONS

- 1.1. Certain Defined Terms. As used in this Agreement,(i) terms defined in the preamble or in the recitals hereto, unless otherwise defined, have the meanings given to them therein, and(ii) the following terms have the following meanings:
- "Acquisition Price" means, for any asset, the aggregate purchase price therefor (whether paid in cash or other property, and determined using the fair market value of property other than cash), including bona fide purchase money financing provided by the seller and all existing Indebtedness pertaining to such asset (a) that is assumed as part of the acquisition of such asset or (b) subject to which such asset is acquired.
- "Additional Similar Assets" means (a) any property or assets (other than Indebtedness or securities) in a line of business in which the Borrower is engaged as of the date hereof or that is not substantially different from any such line, provided that with respect to the acquisition of a Store with Net Available Cash derived from the disposition of a Store, such new Store shall not encompass a materially greater amount of square footage than the disposed of Store, or (b) the capital stock of a Person engaged in a line of business in which the Borrower is engaged as of the date hereof.
- "Adjusted Base Rate" means a per annum rate of interest equal to the Base Rate (changing as such Base Rate changes) plus the Margin (changing as such Margin changes).

"Adjusted Current Assets" means, in relation to any Person as at any date, all amounts (other than cash and Cash Equivalents) which would be set forth opposite the caption "total current assets" (or any like caption) on a balance sheet of such Person as at such date, prepared in accordance with GAAP.

"Adjusted Current Liabilities" means, in relation to any Person as at any date, all amounts which would be set forth opposite the caption "total current liabilities" (or any like caption) on a balance sheet of such Person as at such date, prepared in accordance with GAAP.

"Adjusted EBITDA" means, for any Person for any period,
(a) such Person's EBITDA for such period minus (b) all costs of
acquisition of Rental Items for Stores for such period, to the
extent that (i) such costs of acquisition are accounted for as
Capital Expenditures under GAAP and (ii) such Rental Items were
acquired for a particular Store (A) after its opening or (B) in the
case of a Store acquired during such period, after it commenced to
operate as a Store (i.e., excluding the cost of acquisition of such
Store's initial inventory, the intent of the parties being to adjust
EBITDA for the recurring costs of keeping a Store in stock, and not
for the initial cost of equipping a Store with inventory), plus
(c) the cost of product transferred from rental inventory to
previously viewed merchandise inventory and sold.

"Adjusted Leverage Ratio" means the ratio of, for any period of four consecutive fiscal quarters of the Borrower (determined on a consolidated basis), (a) (x) the Borrower's Funded Debt as of the last day of such period plus (y) an amount equal to six times the aggregate rent expense incurred by the Borrower for such period to (b) the Borrower's Adjusted Net Cash EBITDA for such period plus the aggregate rent expense incurred by the Borrower for such period.

"Adjusted Net Cash EBITDA" means Adjusted EBITDA minus the cost of acquisition of initial inventory for new Stores for the corresponding period.

"Affiliate" means, with respect to any specified Person, any other Person that, directly or indirectly, controls, is controlled by, or is under common control with, such specified Person. For purposes of the Loan Documents, "control" means the power to direct the management and policies of another Person, directly or indirectly, through the ownership of voting securities, by contract, or otherwise, and the terms "controlled by" and "under common control with" have correlative meanings.

In addition, the term "Affiliate" also means, with respect to the Borrower or any Subsidiary, (a) any beneficial owner of capital stock of the Borrower (or of options, warrants or other rights to acquire capital

stock of the Borrower, whether or not exercisable at the time of determination) representing ten percent (10%) or more, on a fully diluted basis, of the total voting power of all voting securities of the Borrower entitled to vote in the election of directors, and (b) any Person who would be an Affiliate of such beneficial owner pursuant to the first sentence of this definition.

"Agent" means Societe Generale in its capacity as agent for the Lenders, and any successor agent selected pursuant to Section 8.6.

"Agent-Related Persons" has the meaning given in Section 8.4.

"Aggregate Investment" means (a) with respect to Investments in a Person, the sum of (i) the Acquisition Price of the securities of such Person, plus (ii) all contributions of cash or other property (valued at fair market value) made, or committed to be made, to such Person, plus (iii) all other Investments in such Person, plus (iv) (without duplication of amounts included pursuant to clause (a) of this definition) aggregate Indebtedness (including, without limitation, quaranties and other contingent obligations in support of Indebtedness of another Person) incurred in support of the obligations of such Person; and (b) with respect to a line of business not conducted through another Person, the sum of (i) the aggregate Acquisition Price of all property used in the conduct of such line of business (or, if such property is not acquired for the purpose of conducting such line of business, but is later used therein, the fair market value of such property at the time such use commences), plus (ii) cumulative operating losses (before depreciation) incurred in the conduct of such line of business.

"Agreement" has the meaning provided in the preamble hereof.

"Ancillary Business" means any business (a) substantially different from those lines of business carried on by the Borrower on the date hereof, that (b) the Board of Directors of the Borrower determines in good faith is ancillary or complementary to the lines of business carried on by the Borrower on the date hereof.

"Applicable Interest Period" means, with respect to any IBOR Loan, the period commencing on the first day the Borrower elects to have such IBOR Rate apply to such Loan and (subject to Section 2.10.3) ending on a day, one, two or three months thereafter, as specified in the Interest Rate Notice given in respect of such Loan or as otherwise determined pursuant to Section 2.6.2; provided, however, that (a) no Applicable Interest Period may be selected for a Loan if it extends beyond the Maturity Date, and (b) no Applicable Interest Period may be selected so as to end later than the next succeeding date of any scheduled reduction

in the Total Commitment if the aggregate principal amount of the IBOR Loan subject thereto, when aggregated with the aggregate principal amount of all other then outstanding IBOR Loans subject to Applicable Interest Periods that end after the date of such scheduled reduction, would exceed the amount of the Total Commitment, as reduced thereon.

"Applicable Interest Rate" means, for each Loan, the Adjusted Base Rate or IBOR Rate, as designated by the Borrower in an Interest Rate Notice given with respect to such Loan (or portion thereof) or as otherwise determined pursuant to Section 2.6.2.

"Arrangers" means Societe Generale and Goldman Sachs Credit Partners L.P.

"Asset Disposition" means any sale, lease, transfer or other disposition (or series of sales, leases, transfers or other dispositions):

- (a) by the Borrower or any Subsidiary (other than to the Borrower) (including, without limitation, any disposition by way of merger, consolidation or similar transaction),
- (b) to any Person (other than in the case of a disposition by a Subsidiary to the Borrower)

(each of the foregoing, for purposes of this definition, a "disposition"), of (x) any shares of the capital stock of any Subsidiary (other than directors' qualifying shares and, to the extent required by local ownership laws in any foreign country, shares owned by foreign shareholders), (y) all or substantially all of the assets of any division, business segment or comparable line of business of the Borrower or any Subsidiary, or (z) any other assets of the Borrower or any Subsidiary outside the ordinary course of such Person's business (including, without limitation, the forgiveness of any debt owing to the Borrower); but excluding (i) any disposition that constitutes a Permitted Investment; (ii) distributions to shareholders, to the extent permitted under the Loan Documents; (iii) sales of Rental Items consistent with past practice; (iv) a disposition (other than a sale or closure of a Store) for fair market value, of assets having a fair market value of less than \$500,000, if the fair market value of the assets subject to such disposition, when aggregated with the fair market value of the assets subject to all prior dispositions of assets during the same fiscal year that have been excluded from the definition of "Asset Disposition" pursuant to this clause (iv), does not exceed \$2,000,000, (v) in connection with a termination referred to in the proviso in Section 6.13, (vi) the transfer of any or all Trademarks to Hollywood Management on or prior to December 31, 2000, provided that contemporaneously with such transaction, the Borrower shall execute and deliver to the Agent any and all documentation

deemed necessary by the Agent such that the Lenders' security interest in such assets shall continue in full force, effect and priority, (vii) the maintenance of the Borrower's cash management system in existence on the Closing Date in order to pay expenses in the ordinary course of business and (viii) without limiting the proviso in Section 6.13 and subject to the limitations contained in clause (iv) above, the closure (other than a sale or transfer) of not more than 20 Stores in any fiscal year of the Borrower reasonably acceptable to the Agent.

"Assignment Agreement" has the meaning provided in Section 8.2.1.

"Bankruptcy Code" means the Bankruptcy Code of the United States, 11 U.S.C. 101 et seq., as amended from time to time, and any successor statute.

"Base Rate" means, for any day, the higher of: (a) 0.50% per annum above the latest Federal Funds Rate; and (b) the rate of interest in effect for such day as publicly announced from time to time by Societe Generale, New York Branch, as its "reference rate." (The "reference rate" is a rate set by Societe Generale, New York Branch, based upon various factors including its costs and desired return and general economic conditions, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate.)

"Base Rate Loan" means a Loan bearing interest at the Adjusted Base Rate.

"Blocked Account Agreement" means that certain Blocked Account Agreement dated as of August 2, 2000, among Bank of America, N.A., the Borrower, Hollywood Management and the Agent.

"Borrower" has the meaning provided in the preamble hereof.

"Borrower Disclosure Letter" means that certain letter, dated as of the date hereof, signed on behalf of Borrower and delivered to the Agent on the Closing Date.

"Borrower Security Agreement" means that certain Security Agreement dated as of September 5, 1997 made by the Borrower in favor of the Agent.

"Business Day" means any day other than Saturday, Sunday or another day on which banks are authorized or obligated by law to close in New York City or Portland, Oregon, except in the context of the selection of a Loan accruing interest at the IBOR Rate or the calculation of the IBOR Rate for any Applicable Interest Period, in which event "Business Day" means any day other than Saturday or

Sunday on which dealings in foreign currencies and exchange between banks may be carried on in Grand Cayman, British West Indies, and New York City.

"Capital Assets" means, with respect to any Person, all equipment, fixed assets and real property or improvements of such Person, or replacements or substitutions therefor or additions thereto, that, in accordance with GAAP, have been capitalized on the books of such Person. Capital Assets shall also (a) include Rental Items excluded from the calculation of Adjusted EBITDA and (b) exclude Rental Items included in the calculation of Adjusted EBITDA.

"Capital Expenditures" means, with respect to any Person for any period, all expenditures made directly or indirectly by such Person during such period for Capital Assets (including all expenditures for maintenance and repairs which are required, in accordance with GAAP, to be capitalized on the books of such Person).

"Capital Leases" means for any Person, all obligations of such Person under leases which are, or, in accordance with GAAP, are required to be, recorded as capital leases.

"Cash Equivalents" means any or all of the following: (a) obligations of, or guaranteed as to principal and interest by, the United States government or any agency thereof maturing within one year after the date on which such obligations are purchased; open market commercial paper of any corporation (other than the Borrower) incorporated under the laws of the United States or any State thereof or the District of Columbia rated P-1 or its equivalent or higher by Moody's or A-1 or its equivalent or higher by S&P; (c) time deposits, certificates of deposit or bankers' acceptances maturing within one year after the acquisition thereof issued by commercial banks organized under the laws of any country which is a member of the OECD and having a combined capital and surplus in excess of \$ 100,000,000 or which is a Lender; (d) repurchase agreements with respect to Securities described in clause (a) above entered into with an office of a bank or trust company meeting the criteria specified in clause c); (e) money market funds investing only in Investments described in clauses (a) through (d); and (f) direct obligations of, or obligations the principal and interest of which are unconditionally guaranteed by, any State of the United States or any foreign state having, at the date of its acquisition by the Borrower, a rating of at least AA from S&P or Aa from Moody's, in each case maturing within one year from the date of the acquisition.

[&]quot;Closing Date" means August 4, 2000.

[&]quot;Code" means the Internal Revenue Code of 1986, as amended

from time to time.

"Collateral" means, except to the extent expressly excluded therefrom pursuant to the terms of any Security Agreement, all personal property of the Borrower and of each Subsidiary (whether or not the creation of a Lien thereon is subject to the Uniform Commercial Code, as in effect in any relevant jurisdiction), including, without limitation, whether now owned or hereafter acquired, all right, title and interest of the Borrower and each Subsidiary in any and all (a) goods, (b) accounts, (c) chattel paper, (d) documents, (e) instruments, (f) investment property, (g) rights to proceeds of written letters of credit, (h) money, (i) general intangibles, (j) intellectual property and (k) subject to the provisions of Section 5.13, real property and leasehold interests.

"Commitment" has the meaning given in Section 2.1.1.

"Controlled Group" means all members of a controlled group of corporations and all trades or businesses (whether or not incorporated) under common control which, together with the Borrower, are treated as a single employer under Section 414(b) or 414(c) of the Code.

"Dallas Office" means Societe Generale, Southwest Agency.

"Default" means any event which, but for the passage of time, with or without the giving of notice, would be an Event of Default.

"Default Rate" has the meaning given in Section 2.6.1(a).

"Defaulting Lender" means any Lender that fails or refuses to perform its obligations under this Agreement within the time period specified for performance of such obligation or, if no time frame is specified, for a period of five (5) Business Days after notice from Agent.

"EBITDA" means, for any Person for any period, such Person's net income (or net loss), excluding any extraordinary gains or losses and taxes associated therewith, plus the sum of (a) interest expense, (b) income tax expense, (c) depreciation expense, (d) amortization expense, and (e) all other noncash items deducted for purposes of determining net income (other than items that will require cash payments and for which an accrual or reserve is, or is required under GAAP, to be made), in each case determined on a consolidated basis in accordance with GAAP for such Person for such period, provided, that (x) for any period that includes results from any or all of the fiscal quarters ended September 30, 1999, December 31, 1999, March 31, 2000 and June 30, 2000, there shall be excluded any EBITDA attributable to the results of operations of

Reel.com in an amount not to exceed (9,200,000), (22,200,000), (15,400,000) and (15,000,000), respectively and (y) for any period that includes the three months ended June 30, 2000, there shall be excluded the amount of the Borrower's nonrecurring restructuring charge (which may be categorized both in a restructuring charge line item and in cost of goods sold) related to Reel.com after March 31, 2000 in an amount not to exceed \$70,000,000, as reflected in the Borrower's financial statements for the three months ended June 30, 2000; provided further, that in determining net income (or net loss) for the Borrower x) for any period that includes the three months ended September 30, 1999, December 31, 1999, March 31, 2000 or June 30, 2000, there shall be excluded the nonrecurring \$2.3 million pre-tax expense related to the settlement in such period of the lawsuit Twentieth Century Fox Home Entertainment, Inc. v. Hollywood Entertainment Corporation and Does 1 through 100 and (y) for any period that includes the three months ended December 31, 1999, March 31, 2000, June 30, 2000 or September 30, 2000, there shall be excluded the nonrecurring \$23.1 million pre-tax expense related to the settlement in such period of the lawsuit Rentrak Corporation v. Hollywood Entertainment, et al.

"Eligible Assignee" means (a) (i) a commercial bank, savings and loan association or savings bank, in each case, organized under the laws of the United States or any state thereof; (ii) a commercial bank organized under the laws of any other country or a political subdivision thereof, provided that (A) such bank is acting through a branch or agency located in the United States, or (B) such bank is organized under the laws of a country that is a member of the Organization for Economic Cooperation and Development or a political subdivision of such a country; and (iii) any other Person that (A) is an "accredited investor" (as defined in Regulation D promulgated under the Securities Act of 1993, as amended), and (B) extends credit or buys loans as one of its businesses (including, without limitation, insurance companies, mutual funds and lease financing companies); and (b) any Lender or Affiliate of any Lender; provided, however, that (x) neither the Borrower nor any Affiliate of the Borrower shall be an Eligible Assignee, and (y) no Person shall be an Eligible Assignee unless, at the time it first acquires an interest in the Loans, the Borrower would not be required to deduct or withhold any Taxes or other amounts from or in respect of any sum payable hereunder or under any other Loan Document to such Person.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time.

"Event of Default" has the meaning given in Section 7.1.

"Excess Cash Flow" means, for any fiscal quarter of the Borrower (commencing with the fiscal quarter ended March 31, 2001 and determined on or prior to the 45th day after the end of each such fiscal quarter), the amount, if any, by which the sum of:

- (i) Adjusted Net Cash EBITDA of the Borrower and its Subsidiaries for such period, plus
- (ii) any increase in Working Capital of the Borrower and its Subsidiaries for such period,

exceeds an amount equal to 110% of the sum of:

- (i) the aggregate amount of all prepayments of Loans during such period, solely to the extent accompanied by a permanent reduction of the Total Commitment, plus
- (ii) Fixed Charges of the Borrower and its Subsidiaries for such period, plus
- (iii) any decrease in Working Capital of the Borrower and its Subsidiaries for such period, plus
- (iv) \$10,000,000.
- "Existing Credit Agreement" has the meaning provided in the first recital hereof.
- "Facility" means the revolving line of credit (including the Swing Line subfacility) described in Section 2.1.
- "Federal Funds Rate" means, for any period, a fluctuating interest rate per annum equal for each day during such period to the weighted average of the rates on overnight federal funds transactions with members of the Federal Reserve System arranged by federal funds brokers, as published for such day (or, if such day is not a Business Day, for the next preceding Business Day) by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for such day on such transactions received by Agent from three federal funds brokers of recognized standing selected by it.
- "Financing Statements" means those Uniform Commercial Code financing statements, satisfactory in form and substance to the Agent and naming the Agent as secured party, executed by the Borrower or any Subsidiary, as the case may be, as debtor, in form acceptable for filing in each of the fifty states and the District of Columbia and identifying by item or type the Collateral described in the applicable Security Agreement.
- "Fixed Charges" means, in relation to any Person for any period, the sum (without duplication) of (a) the aggregate amount of interest paid in cash with respect to Indebtedness for such period, plus (b) the aggregate amount of income taxes paid (net of income

tax refunds received) in cash by such Person during such period, plus (c) Capital Expenditures (other than the cost of acquisition of initial inventory for new Stores) of such Person paid in cash for such period, plus (d) all regularly scheduled principal payments during such period with respect to Indebtedness of such Person, all determined in accordance with GAAP.

"Funded Debt" means, for any Person, without duplication:

(a) all indebtedness or liability of such Person for borrowed money and all indebtedness or liability for borrowed money secured by a Lien on the assets of such Person, whether or not such indebtedness or liability has been assumed by such Person, (b) all indebtedness and liability of such Person for Capital Leases and (c) all indebtedness or liability for borrowed money or for Capital Leases for which such Person is indirectly or contingently liable (as obligor, guarantor, or otherwise), but, in the case of Funded Debt described in this clause (c), not in excess of any contractual limit on such Person's liability therefor.

"Funding Date" means, with respect to any Loan, the date on which such Loan is funded pursuant to Article II.

"GAAP" has the meaning given in Section 1.3.

"Government Approval" means an approval, permit, license, authorization, certificate, or consent of any Governmental Authority.

"Governmental Authority" means the government of the United States or any State or any foreign country, or of any political subdivision of any thereof, or any branch, department, agency, instrumentality, court, tribunal or regulatory authority which constitutes a part or exercises any sovereign power of any of the foregoing.

"Guaranty" means any guaranty granted by any Subsidiary in favor of the Agent, substantially in the form of the Hollywood Management Guaranty.

"Hedging Agreement" means any interest rate swap agreement, interest rate cap agreement or other financial agreement or arrangement designed to protect the Borrower or any Subsidiary against fluctuations in interest rates or in the prices of any goods or services.

"High Yield Notes" means (a) the "Initial Securities", as defined in, and issued pursuant to, the Indenture, and (b) the "Exchange Securities" (as defined in the Indenture), if any, issued in exchange therefor as contemplated by the High Yield Offering Memorandum and the Indenture, provided that such Exchange Securities have terms (including subordination provisions) substantially

identical in all material respects to such Initial Securities, except that such Exchange Securities need not contain the same restrictions on transfer or registration rights as do such Initial Securities.

"High Yield Offering" means (a) at any time prior to the offering of the "Exchange Securities" described in the Indenture, the Borrower's offering of the High Yield Notes outstanding as of the September 5, 1997, and (b) upon commencement of the Borrower's offering of such Exchange Securities, both the offering described in the preceding clause (a) and the offering of such Exchange Securities pursuant to the "Exchange Offer Registration Statement" described in the High Yield offering Memorandum.

"High Yield Offering Memorandum" means (a) at any time prior to the filing of the "Exchange Offer Registration Statement" referred to above, the Offering Memorandum dated July 18, 1997, relating to the issuance of the Borrower's senior subordinated notes due 2004, and (b) following the filing of such Exchange Offer Registration Statement, collectively, the Offering Memorandum described in the preceding clause (a), such Exchange Offer Registration Statement and any offering memorandum or prospectus, in each case, as at any time amended.

"Hollywood Management" means Hollywood Management Company, an Oregon corporation.

"Hollywood Management Guaranty" means that certain Guaranty, dated as of June 23, 2000, between Hollywood Management and the Agent.

"Hollywood Management Security Agreement" means that certain Subsidiary Security Agreement, dated as of June 23, 2000, between Hollywood Management and the Agent.

"IBOR Loan" means a Loan bearing interest at an IBOR Rate.

"IBOR Rate" means, with respect to any Loan for any Applicable Interest Period, an interest rate per annum equal to the sum of (a) the Margin (changing effective as of the date of each change in the Margin) and (b) the product of (i) the Offshore Rate; and (ii) the Eurodollar Reserves (as defined below) in effect on the first day of such Applicable Interest Period. As used herein, the "Offshore Rate" means the rate of interest per annum determined by the Agent as the rate at which dollar deposits in the approximate amount of such Loan for such Applicable Interest Period would be offered by Societe Generale's Grand Cayman Branch, Grand Cayman B.W.I. (or such other office as may be designated for such purpose by Societe Generale) to major banks in the offshore dollar interbank market at their request at approximately 11:00 a.m. (New York City time) two (2) Business Days prior to the commencement of such

Applicable Interest Period. As used herein, "Eurodollar Reserves" means a fraction (expressed as a decimal), the numerator of which is the number one and the denominator of which is the number one minus the aggregate of the maximum reserve percentages (including, without limitation, any special, supplemental, marginal or emergency reserves), expressed as a decimal, established by the Board of Governors of the Federal Reserve System or any other banking authority to which the Lenders are subject for Eurocurrency Liability (as defined in Regulation D of such Board of Governors). It is agreed that for purposes hereof, each Loan accruing interest at the IBOR Rate shall be deemed to constitute a Eurocurrency Liability and to be subject to the reserve requirements of Regulation D, without benefit of credit or proration, exemptions or offsets which might otherwise be available to the Lenders from time to time under such Regulation D. Eurodollar Reserves shall be adjusted automatically on and as of the effective date of any change in any reserve percentage and shall apply to Applicable Interest Periods commencing after the effective date of such change.

"Indebtedness" means, for any Person, without duplication:
(a) all items of indebtedness or liability (other than capital, surplus, deferred credits and reserves) which, in accordance with GAAP, would be included in determining total liabilities as shown on the liability side of a balance sheet as of the date as of which indebtedness is determined, (b) indebtedness secured by any Lien, whether or not such indebtedness shall have been assumed, (c) any other indebtedness or liability for borrowed money or for the deferred purchase price of property or services for which such Person is directly or contingently liable as obligor, guarantor, or otherwise, or in respect of which such Person otherwise assures a creditor against loss, (d) any other obligations of such Person under Capital Leases, and (e) any obligations of such Person under any Hedging Agreement.

"Indenture" means the Indenture dated as of August 13, 1997, between the Borrower, as Issuer, and U.S. Trust Company of California, N.A., as Trustee, with respect to the Borrower's 10-5/8% Senior Subordinated Notes Due 2004 as at any time amended or supplemented in accordance with the provisions hereof.

"Interest and Rent Coverage Ratio" means, for any period of four consecutive fiscal quarters of the Borrower, the ratio of (a) Adjusted Net Cash EBITDA for such period, minus cash taxes paid by the Borrower and its consolidated Subsidiaries during such period (net of tax refunds received), plus the aggregate rent expense incurred by the Borrower for such period to (b) total interest expense of the Borrower for such period, plus the aggregate rent expense incurred by the Borrower for such period determined on a consolidated basis in accordance with GAAP.

"Interest Coverage Ratio" means, for any period of four

consecutive fiscal quarters of the Borrower, the ratio of
(a) Adjusted Net Cash EBITDA for such period, less cash taxes paid
by the Borrower and its consolidated Subsidiaries during such
period, to (b) total interest expense of the Borrower for such
period, determined on a consolidated basis in accordance with GAAP.

"Interest Rate Notice" has the meaning given in Section 2.6.2.

"Investment" means, as applied to any Person, (a) any direct or indirect purchase or other acquisition by that Person of (i) any equity or debt securities (including, without limitation, capital stock, shares, voting trust certificates, bonds, debentures, notes or instruments or other securities or evidences of indebtedness or other obligations), or (ii) a beneficial interest in any equity or debt securities of any other Person; and (b) any direct or indirect loan, advance (other than deposits with financial institutions available for withdrawal on demand, prepaid expenses, and similar items made or incurred in the ordinary course of business), or capital contribution by such Person to any other Person (including, without limitation, by means of any transfer of cash or other property or payment for property or services for the use of such other Person), including all Indebtedness and accounts owed by that other Person (x) that arise from any Asset Disposition, (y) that are not current assets or (z) that did not arise from sales of goods or services to that Person in the ordinary course of business and on arm's-length terms.

"Lenders" means Societe Generale, Goldman Sachs Credit Partners L.P. and each of the other Persons identified as Lenders on the signature pages hereto, and any Successors to or permitted assigns under Section 9.4 of any of the foregoing.

"Leverage Ratio" means, for any period of four consecutive fiscal quarters of the Borrower (determined on a consolidated basis), the ratio of (a) the Borrower's Funded Debt as of the last day of such period, to (b) the Borrower's Adjusted Net Cash EBITDA for such period; provided that solely for the purpose of calculating the applicable Margin, clause (b) shall be "the Borrower's Adjusted EBITDA for such period."

"Lien" means, for any Person, any security interest, pledge, mortgage, charge, assignment, hypothecation, encumbrance, attachment, garnishment, execution or other voluntary or involuntary security interest or lien in, on or affecting the revenues of such Person or any real or personal property in which such Person has or hereafter acquires any interest, except (i) (without limiting the provisions of Section 7.1.12(b)) liens for Taxes which are not delinquent or which remain payable without penalty or the validity or amount of which is being contested in good faith by appropriate proceedings upon stay of execution of the enforcement thereof with

appropriate reserves having been established therefor, (ii) liens imposed by law (such as mechanics' liens) incurred in good faith in the ordinary course of business which are not delinquent or which remain payable without penalty or the validity or amount of which is being contested in good faith by appropriate proceedings upon stay of execution of the enforcement thereof with, in the case of liens on property of the Borrower, provision having been made to the satisfaction of the Agent for the payment thereof in the event the contest is determined adversely to the Borrower, (iii) deposits or pledges under worker's compensation, unemployment insurance, social security or other similar laws or made to secure the performance of bids, tenders, contracts (except for repayment of borrowed money), or leases (other than Capital Leases), or to secure statutory obligations or surety or appeal bonds or to secure indemnity, performance or other similar bonds given in the ordinary course of business and (iv) liens arising solely by virtue of any statutory or common law provision relating to banker's liens, rights of setoff or similar rights and remedies as to deposit accounts or other funds maintained with a creditor depository institution (except as may be modified by or under the Blocked Account Agreement), provided that (A) such deposit account or other funds is/are available to such Person on demand, and (B) such deposit account or other funds are neither established nor maintained with such depository institution as security for, or a condition to or other requirement imposed in connection with the incurring or maintenance of, any Indebtedness to such depository institution.

"Loan" means a loan made by one or more Lenders pursuant hereto, provided that (a) if any Revolving Loan or Revolving Loans (or portions thereof) are combined or subdivided pursuant to Section 2.6.2, the term "Loan" means such combination or such subdivided portion thereof, as the case may be, (b) where the context so requires, the term "Loan" means, with respect to a particular Lender, the advance made (or required to be made) by such Lender in the amount of such Lender's Percentage Interest of a borrowing hereunder (or, if less, the remaining unused portion of such Lender's Commitment) and (c) except where the context otherwise requires, the term "Loan" also means each advance under the Swing Line.

"Loan Documents" means this Agreement, the Notes, the Security Agreements, each Guaranty, the Blocked Account Agreement, the Wattles Assignment Agreement and all other certificates, instruments and other documents executed by the parties hereto (or their predecessors or successors in interest) or any of their respective officers or directors in connection with this Agreement or the transactions contemplated hereby and any amendments thereto made in accordance with the terms of this Agreement.

"Majority Lenders" means at any time Lenders having an aggregate Percentage Interest of greater than fifty percent (50%).

"Mandatory Reduction" means a reduction in the Total Commitment effected (a) pursuant to the definition of "Total Commitment" or (b) pursuant to Section 2.3.2.

"Margin" means a per annum rate determined in accordance with the following chart, adjusted as of the first day of the first full calendar month beginning after the date on which the Borrower delivers the quarterly compliance certificates required under Section 5.10.3 with respect to any fiscal quarter, based on the Leverage Ratio for the period of four consecutive fiscal quarters ended as of the last day of such fiscal quarter, provided that if the Borrower fails to deliver such quarterly compliance certificate on or before the date it is due, then, from the first day of the first full calendar month beginning after the date such delivery was due until the first Business Day after such quarterly compliance certificate is delivered, the Margin shall be the lesser of (a) 25 basis points over the Margin theretofore in effect, or (b) the maximum Margin (for Base Rate Loans or IBOR Loans, as the case may be) forth set in the following chart:

Margin for	use Margin	for use	
in determining in determining			
Adjuste	d Base Rate		
0 1.50	0.875%		0.000%
t less			
75	1.125%		0.125%
t less			
00	1.375%		0.375%
t less			
50	1.625%		0.625%
t less			
00	1.875%		0.875%
	2.125%	1.125	0
	ng in Adjusted to 1.50 to less 75 to less 00 to less 50 to less	in determining Adjusted Base Rate 0 1.50	Adjusted Base Rate 0 1.50

"Material Adverse Effect" means a material adverse effect on the business, operations or financial condition of the Borrower and its Subsidiaries taken as a whole; or a material adverse effect on the ability of the Borrower and its Subsidiaries to perform their obligations under any of the Loan Documents.

"Net Available Cash" means, with respect to any Asset Disposition, the cash payments received by the Borrower or any Subsidiary therefrom (including, without limitation (but without limiting the provisions of Section 6.6), as and when received, any cash payments received by way of deferred payment of principal

[&]quot;Maturity Date" means September 5, 2002.

pursuant to any note or installment receivable or otherwise and cash received upon the conversion into cash of marketable securities or other property received as consideration for such Asset Disposition), in each case net of all (a) (i) legal, title, recording and transfer tax expenses relating to such Asset Disposition, (ii) commissions and other fees and expenses incurred to any Person other than an Affiliate of the Borrower in connection with such Asset Disposition, and (iii) Federal, state, provincial, foreign and local taxes required to be paid, as a consequence of such Asset Disposition, within twelve months after the date thereof; (b) payments made on any Indebtedness (i) secured by any assets subject to such Asset Disposition, in accordance with the terms of any Lien of any kind thereon, as a condition precedent to the consummation of such Asset Disposition, (ii) required to be made, at or before the consummation of such Asset Disposition, to obtain a necessary consent thereto, or (iii) required to be paid from the proceeds of such Asset Disposition under applicable law; (c) in the case of an Asset Disposition by a Subsidiary, distributions and other payments required to be made to any holders of minority interests in such Subsidiary as a consequence of such Asset Disposition; and (d) appropriate reserves established by the seller, in accordance with GAAP or as otherwise agreed by the Majority Lenders, against any liabilities (i) associated with the assets subject to such Asset Disposition (including, without limitation, liabilities under any indemnity obligations incurred as part of such Asset Disposition), (ii) that are retained by the Borrower or the Subsidiary party to such Asset Disposition.

"Net Cash Proceeds" means, with respect to any issuance or sale of debt or capital stock by the Borrower, the cash proceeds of such issuance or sale, net of (a) (i) attorneys' fees and disbursements, (ii) accountants' fees, (iii) underwriters' or placement agents' fees, discounts or commissions, and (iv) brokerage, consultants' and other fees, in each case, actually incurred in connection with such issuance or sale and (b) taxes paid or payable as a result thereof.

"Net Income" means, for any accounting period, the consolidated net income of the Borrower and its Subsidiaries for such period, determined in accordance with GAAP.

"Net Worth" means, for any Person, such Person's owners' or shareholders', as the case may be, equity, determined in accordance with GAAP.

"New York Office" means Societe Generale, New York Branch.

"Notes" has the meaning given in Section 2.9.

"Notice of Borrowing" means a written request for a Loan, executed by the Borrower and delivered to the Agent and containing

the information set forth in Section 2.2.

"Obligations" means, from time to time, all indebtedness or other obligations of the Borrower and its Subsidiaries owing to the Agent, any Lender or any Indemnified Person (as defined in Section 9.5.1), or any of their respective Successors, transferees or permitted assigns, of every type and description, whether or not evidenced by any note, quaranty or other instrument, arising under or in connection with this Agreement or any other Loan Document, whether or not for the payment of money, whether direct or indirect (including those acquired by assignment), absolute or contingent, due or to become due, now existing or hereafter arising and however acquired, and including, without limitation, all present and future obligations for the payment of principal, premium, interest (including interest accruing after the filing of any petition under the Bankruptcy Code, at the Default Rate), penalties, fees, indemnifications, reimbursements, damages and other liabilities at any time payable under the Loan Documents.

"Officer's Certificate" means a certificate signed in the name of the Borrower by its Chairman, its President, its Chief Financial Officer, its Senior Vice President of Finance, its Controller or its Treasurer.

"Passive Investment" means the acquisition by the Borrower of voting securities of a Person, engaged (or being formed to engage) in an Ancillary Business, provided that (a) such Person is a corporation, limited partnership or other limited liability entity and the securities therein acquired by the Borrower do not expose the Borrower generally to liability for the obligations of such Person; (b) Borrower does not own a majority of the outstanding voting securities of such Person or otherwise have the right to control the management of such Person; (c) the financial results of such Person are not required, under GAAP, to be reported on a consolidated basis with those of the Borrower; and (d) the Borrower does not become indirectly or contingently liable (as obligor, guarantor, or otherwise) for any obligation of such Person, or otherwise assure any creditor of such Person against loss.

"PBGC" means the Pension Benefit Guaranty Corporation or any entity succeeding to any or all of its functions under ERISA.

"Pension Plan" means an "employee pension benefit plan" (as such term is defined in ERISA) from time to time maintained by the Borrower or a member of the Controlled Group.

"Percentage Interest" means, for any Lender at any time, the percentage that such Lender's Commitment is of the Total Commitment.

"Permitted Acquisition" means an acquisition by the

Borrower of (a) all or substantially all of the assets (or of all or substantially all of the assets of any division, business segment or comparable line of business) of another Person, or (b) all of the outstanding capital stock of another Person (including acquisitions by merger), in either case, that does not otherwise cause the occurrence of a Default or Event of Default, provided that no later than twenty (20) Business Days prior to the consummation of such acquisition, the Borrower delivers to the Agent the Pro Forma Calculations with respect to such acquisition.

"Permitted Investment" means any Investment permitted under Section 6.6.

"Person" means an individual, a corporation, a partnership (whether general, limited, or limited liability), a limited liability company, a trust, an unincorporated association, a joint venture, a joint-stock company, a government (including any political subdivision of a government), a Governmental Authority or any other entity.

"Plan" means, at any time, an employee pension benefit plan which is covered by Title IV of ERISA or subject to the minimum funding standards under Section 412 of the Code and is either (a) maintained by the Borrower or any member of the Controlled Group for employees of the Borrower or any member of such Controlled Group or (b) maintained pursuant to a collective bargaining agreement or any other arrangement under which more than one employer makes contributions and to which the Borrower or any member of the Controlled Group is then making or accruing an obligation to make contributions or has within the preceding five (5) plan years made contributions.

"Pro Forma Calculations" means, with respect to any transaction, pro forma financial statements of the Borrower (a) prepared as of the last day of the then most recently completed fiscal quarter of the Borrower, (b) giving pro forma effect to such transaction as of such date, (c) establishing that, on a pro forma basis, the Borrower would have been in compliance with each of the financial covenants set forth in Sections 5.15 through 5.20 as of the end of such most recently completed fiscal quarter notwithstanding the consummation of such transaction (provided that financial covenants otherwise required to be tested with respect to any period of four consecutive fiscal quarters shall be calculated on the basis of the twelve-month period ended as of the last day of such then most recently completed fiscal quarter and as if such transaction had been consummated on the first day of such twelvemonth period), and (d) certified by the Borrower's chief financial officer as having been prepared in accordance with the provisions of this Agreement as being accurate in all material respects.

"Reel.com" means Reel.com, Inc., a Delaware corporation,

which was merged as of June 9, 2000 into the Borrower.

"Rental Items" means videotapes, video discs (regardless of format), videogames, audiotapes and related equipment to the extent that such items were acquired by Borrower (or the relevant Subsidiary) for sale or rental to its customers or are held by Borrower (or the relevant Subsidiary) for rental to its customers.

"Rental Revenues" means revenues received by the Borrower (on a consolidated basis) for the rental of Rental Items.

"Revolving Loan" means each Loan made by the Lenders under the facility described in Section 2.1.1, as well as each Swing Line Borrowing deemed converted into a Revolving Loan pursuant to Section 2.2.1, provided that (a) if any Revolving Loan or Revolving Loans (or portions thereof) are combined or subdivided pursuant to Section 2.6.2, the term "Revolving Loan" means such combination or such subdivided portion thereof, as the case may be; (b) where the context so requires, the term "Revolving Loan" means with respect to a particular Lender, the advance made (or required to be made) by such Lender in the amount of such Lender's Percentage Interest of a borrowing under the facility described in Section 2.1.1 (or, if less, the remaining unused portion of such Lender's Commitment).

"Security Agreements" means the Borrower Security Agreement and each Subsidiary Security Agreement.

"Store" means each separate location (notwithstanding the corporate or other organization of any multitude of locations) at which the Borrower engages in the rental of Rental Items to the public.

"Subordinated Debt" means the obligations of the Borrower in respect of (a) the High Yield Notes and (b) any other subordinated debt issued in accordance with the provisions of Section 6.2(f).

"Subsidiary" means any corporation, association, limited liability company or other business entity of which the Borrower owns directly or indirectly more than fifty percent (50%) of the voting securities, membership interests or other equity interest thereof or in which the Borrower otherwise owns a controlling interest.

"Subsidiary Security Agreement" means any security agreement executed by any Subsidiary substantially in the form of the Hollywood Management Security Agreement.

"Successor" means, for any corporation or banking association, any successor by (a) merger or consolidation or (b) acquisition of substantially all of the assets of the

predecessor.

"Swing Line" means the subfacility for short-term borrowings described in Section 2.1.2.

"Swing Line Borrowing" means a borrowing under the Swing Line.

"Swing Line Lender" means Societe Generale, acting in its capacity as the lender under the Swing Line, and any Successor to Societe Generale in that capacity.

"Tangible Net Worth" means (a) Net Worth, less (b) all components thereof attributable to intangible assets (determined in accordance with GAAP).

"Tax" has the meaning provided in Section 2.13(a).

"Total Commitment" means (subject to reduction or termination pursuant to Section 2.3 or Section 7.2):

Period Amount

September 5, 1997 through and including December 4, 2000 \$300,000,000

December 5, 2000 through and including March 4, 2001 \$262,500,000

March 5, 2001 through and including June 4, 2001 \$225,000,000

June 5, 2001 through and including September 4, 2001 \$187,500,000

September 5, 2001 through and including December 4, 2001 \$150,000,000

December 5, 2001 through and including March 1, 2002 \$112,500,000

March 2, 2002 through and including June 4, 2002 \$75,000,000

June 5, 2002 through and including September 4, 2002 \$37,500,000

Thereafter \$0

"Total Revenues" means, for any period, the Borrower's aggregate revenues for such period, determined in accordance with GAAP.

"Trademarks" has the meaning given to such term in the Trademark Mortgage and Security Agreement dated as of September 11, 1997 by the Borrower in favor of Societe Generale, as Agent, as amended as of August 2, 2000.

"Unfunded Vested Liabilities" means, with respect to any Plan, at any time, the amount (if any) by which (a) the present value of all vested nonforfeitable benefits under such Plan exceeds (b) the fair market value of all Plan assets allocable to such benefits, all determined as of the then most recent valuation date for such Plan, but only to the extent that such excess represents a potential liability of the Borrower or any member of the Controlled Group to the PBGC or the Plan under Title IV of ERISA.

"Voluntary Reduction" has the meaning given in Section 2.3.1.

"Wattles Assignment Agreement" means that certain Security Assignment Agreement dated as of August 2, 2000 between the Borrower and the Agent.

"Wattles Loan" means the loan by the Borrower of an amount not to exceed \$15,000,000 to Mark J. Wattles and the related pledge of stock by Mr. Wattles to the Borrower to collateralize such loan under the Wattles Pledge Agreement.

"Wattles Note" means the Term Promissory Note, in the aggregate principal amount of \$15,000,000 executed by Mark J. Wattles in favor of the Borrower, dated as of May 8, 2000.

"Wattles Pledge Agreement" means that certain Pledge Agreement (Securities) dated as of June 30, 2000 made by Mark J. Wattles in favor of the Borrower.

"Working Capital" means, in relation to the any Person as at any date, the excess of the Adjusted Current Assets of such Person as at such date, over the Adjusted Current Liabilities of such Person as at such date.

- 1.2. General Principles. Applicable to Definitions,
 Certain Usages of Terms. Definitions given in Section 1.1 shall be
 equally applicable to both singular and plural forms of the terms
 therein defined, and references herein to "he" or "it" shall be
 applicable to Persons whether masculine, feminine or neuter.
 References herein to any document, including, without limitation,
 this Agreement, shall be deemed references to such document as it
 now exists, and as, from time to time hereafter, the same may be
 amended, supplemented or restated. Unless otherwise specified, any
 reference in a Loan Document to an article, section, paragraph,
 subparagraph, schedule or exhibit shall be deemed a reference to an
 article, section, paragraph, subparagraph, schedule or exhibit of or
 to such Loan Document. The word "or" shall not be limiting, but
 shall have the meaning commonly ascribed to "and/or."
- 1.3. Accounting Terms. Except as otherwise expressly provided herein, accounting terms not specifically defined shall be

construed, and all accounting procedures shall be performed: (a) in accordance with generally accepted United States accounting principles consistently applied ("GAAP") and as in effect on the date of application, and (b) in the case of a Person whose financial results would be reported, in accordance with GAAP, on a consolidated basis with the results of Persons under such Person's control, on a consolidated basis.

ARTICLE II.

THE LOANS

- 2.1. The Facility.
- 2.1.1. Revolving Line of Credit. Each Lender severally agrees, on the terms and conditions of this Agreement, to make loans ("Revolving Loans") to the Borrower from time to time on Business Days during the period beginning on the September 5, 1997 and ending on the day before the Maturity Date, in an aggregate principal amount not exceeding at any one time (a) for any Lender, the principal amount set forth below such Lender's name on the signature pages hereto, subject to reduction or termination in accordance with Section 2.3 and Section 7.2 (such amount, at any time, such Lender's "Commitment"), and (b) for all Lenders, (i) the Total Commitment less (ii) the aggregate principal amount of all Swing Line Borrowings that would remain outstanding immediately following the making of any such Revolving Loan and the application of the proceeds thereof to the payment of any outstanding Swing Line Borrowings. The line of credit extended hereunder is a revolving line of credit, and, subject to the terms and conditions hereof, the Borrower may borrow, repay and reborrow up to the maximum principal amount provided for in this Section 2.1.1, at any time before the Maturity Date.
- 2.1.2. Swing Line. The Swing Line Lender additionally agrees, on the terms and conditions of this Agreement, to make advances to the Borrower from time to time on Business Days during the period beginning on the September 5, 1997 and ending on the day before the Maturity Date, in an aggregate principal amount not exceeding at any one time the lesser of (a) (i) the Total Commitment, less (ii) the outstanding principal of all Loans other than Swing Line Borrowings, and (b) \$10,000,000. The Swing Line Lender's obligation to fund Swing Line Borrowings shall be unaffected by its making of any Revolving Loans, notwithstanding that the sum of the Swing Line Borrowings plus the Swing Line Lender's Percentage Interest of the aggregate principal amount of the outstanding Revolving Loans, may exceed the Swing Line Lender's Commitment.
- 2.2. Manner of Borrowing a Loan.

- (a) In connection with each Revolving Loan, the Borrower shall deliver to the Agent at both its Dallas Office and its New York Office, no later than 12:30 p.m. (New York City time), (i) at least one Business Day prior to the requested Funding Date in the case of a Base Rate Loan, or (ii) at least three Business Days prior to the requested Funding Date in the case of an IBOR Loan, a Notice of Borrowing specifying the Funding Date of the requested Revolving Loan and the amount thereof, which shall be in an amount equal to \$2,000,000 or an integral multiple of \$1,000,000 in excess thereof.
- If, as of 12:30 p.m. (New York City time) on the (b) fourteenth day after the Swing Line Lender has funded any Swing Line Borrowing, the Borrower has not repaid such Swing Line Borrowing in full (or if, prior thereto, an Event of Default shall have occurred or the Loans shall be accelerated for any reason whatsoever): (i) the Agent shall promptly (or, if an Event of Default has occurred but the Loans have not been accelerated, may) notify each Lender by telephone (confirmed promptly by telex, facsimile transmission or cable), telex, facsimile transmission, or cable of the amount of such Swing Line Borrowing; and (ii) each Lender shall (subject to Section 2.1.1), before 2:30 p.m. (New York City time), on such date, pay to the Agent at its Dallas Office (or at such other place as the Agent may from time to time specify for such purpose), in immediately available funds, such Lender's Percentage Interest of the principal amount of such Swing Line Borrowing. such funding, each Lender shall be deemed to have acquired from the Swing Line Lender (and the Swing Line Lender shall be deemed to have assigned to each such Lender) a percentage interest in such Swing Line Borrowing equal to such Lender's Percentage Interest, and, for purposes of determining the availability of Swing Line Borrowings or Revolving Loans, such Swing Line Borrowing shall be deemed a Revolving Loan (and no longer a Swing Line Borrowing); provided that the obligations of the Lenders under this Section 2.2.1(b) shall not be subject to the notice or amount requirements, or to satisfaction of conditions precedent, otherwise applicable to the making of Revolving Loans. Each Lender's obligation to fund, and to purchase from the Swing Line Lender, its Percentage Interest of a Swing Line Borrowing pursuant to this Section 2.1.1(b) shall be absolute and unconditional under any and all circumstances (including, without limitation, any intervening bankruptcy of the Borrower or termination of the Total Commitment). It is not the parties' intent that the obligations of the Lenders under this Section 2.1.1(b) constitute guaranties or obligations of suretyship. If and to the extent, however, that the obligations of any Lender under this Section 2.1.1(b) are determined to be those of a quarantor or surety, such Lender, with full knowledge of the consequences thereof, hereby expressly waives the benefit of each and every right or defense of a guarantor or surety the effect of which would

relieve such Lender of all or any portion of its obligations under this Section 2.1.1(b). In the event that any Lender fails to pay to the Agent when due any amount it is required to fund under this Section 2.2.1(b), such Lender and the Borrower severally agree to pay to the Agent, on demand, the amount such Lender has failed to so pay, together with interest thereon for each day from the date on which such payment was due until the date such amount is repaid to the Agent, at (x) in the case of the Borrower, the Applicable Interest Rate, or (y) in the case of such Lender, the Federal Funds Rate. Any such repayment by the Borrower shall be without prejudice to any rights it may have against the Lender that has failed pay when due any such amount.

2.2.2. Swing Line Borrowings. In connection with each Swing Line Borrowing, the Borrower shall deliver to the Agent, no later than 12:30 p.m. (New York City time) at its New York Office and its Dallas Office on the requested Funding Date, a Notice of Borrowing specifying the Funding Date of the requested Swing Line Borrowing and the amount thereof, which shall be in an amount equal to \$500,000 or an integral multiple of \$10,000 in excess thereof.

2.2.3. General Provisions.

- (a) Any Notice of Borrowing received after 12:30 p.m. (New York City time) by either the Agent's New York Office or its Dallas Office on any day will be deemed to have been received on the next succeeding Business Day. Each Notice of Borrowing shall be irrevocable and shall constitute a representation and warranty by the Borrower that, as of the date of the Notice of Borrowing, (i) no Default or Event of Default has occurred and is continuing and (ii) each of the representations and warranties of the Borrower or any Subsidiary set forth in any Loan Document are true and correct in all material respects on and as of such date (except to the extent such representations or warranties are conditioned by reference to a specific date, in which case they shall be true and correct in all material respects as of such date), with the same effect as though such representations and warranties had been made on and as of such date.
- (b) (i) On receipt of any Notice of Borrowing with respect to a Revolving Loan, the Agent shall promptly notify each Lender by telephone (confirmed promptly by telex, facsimile transmission or cable), telex, facsimile transmission, or cable of the requested Funding Date. Each Lender shall, before 2:30 p.m. (New York City time), on the Funding Date of the requested Revolving Loan, pay to the Agent at its Dallas Office (or at such other place as the Agent may from time to time specify for such purpose), in immediately available funds, the lesser of (A) such Lender's Percentage Interest of the aggregate principal amount of the Revolving Loan requested in the Notice of Borrowing or (C) the maximum amount such Lender is committed to advance pursuant to the

- (ii) Upon (A) fulfillment to the Agent's satisfaction of the applicable conditions precedent set forth in Article III, and (B) in the case of a Revolving Loan, after receipt by the Agent of immediately available funds from the Lenders pursuant to the preceding subparagraph (i), the Agent shall promptly make funds available to the Borrower in the amount of the requested Revolving Loan or Swing Line Borrowing, as the case may be, by wire transfer in accordance with such written instructions as may be delivered by the Borrower.
- 2.3. Reductions of Commitments.
- 2.3.1. Voluntary Reductions. Upon not less than five (5) Business Days' written notice to the Agent, the Borrower may terminate the Total Commitment, or reduce it in part, provided that: (a) each partial reduction of the Total Commitment pursuant to this Section 2.3.1 (each such partial reduction, a "Voluntary Reduction") shall be in an amount equal to at least \$5,000,000; (b) in no event may the Total Commitment be reduced to an amount less than the then outstanding aggregate principal balance of the Loans (after giving effect to any prepayment of the Loans made at the time of any Voluntary Reduction).
- 2.3.2. Mandatory Reductions. The Total Commitment shall be reduced (a) (i) to the extent that, by the date that is 180 days after the Borrower's (or any Subsidiary's) receipt of any Net Available Cash in respect of an Asset Disposition, the Borrower has not reinvested such Net Available Cash in Additional Similar Assets (determined on a first-in, first-out basis), or (ii) to the extent that such Net Available Cash, when aggregated with all prior acquisitions of Additional Similar Assets effected with Net Available Cash during the fiscal year in which such Net Available Cash is received, exceeds \$50,000,000, on the date such excess Net Available Cash is received (the amount of such unreinvested Net Available Cash, or of such excess Net Available Cash, as the case may be, the "Excess Disposition Proceeds"), by an amount equal to such Excess Disposition Proceeds; (b) on or prior to the date which is 30 days after the Borrower's receipt thereof, by an amount equal to 50% of the Net Cash Proceeds of any issuance or sale by the Borrower of its capital stock; (c) on or prior to the date which is 30 days after the Borrower's receipt thereof by an amount equal to 50% of the Net Cash Proceeds of any issuance or sale of Subordinated Debt by the Borrower; and (d) on the date of any determination thereof, by an amount equal 100% of any Excess Cash Flow; provided that after payment by the Borrower to the Agent of any amounts owing under clause (b) or (c), (i) an amount not to exceed 10% of Net Cash Proceeds received by the Borrower (in addition to amounts not used under clause (ii) below) shall be used by the Borrower for general working capital purposes (other than Store improvements or growth)

and (ii) an amount not to exceed 40% of Net Cash Proceeds received by the Borrower, shall (without limitation of the parenthetical in clause (i) above) be used solely in accordance with clause (z) (III) of Section 6.12.

2.3.3. Effect of Reduction.

- (a) Each Voluntary Reduction or Mandatory Reduction shall proportionately reduce each Lender's Commitment such that, after such reduction takes effect, each Lender's Commitment will be an amount equal to its Percentage Interest (calculated immediately before such Voluntary Reduction or Mandatory Reduction) of the Total Commitment, as reduced.
- (b) No Voluntary Reduction or unscheduled Mandatory Reduction shall affect the amount of any subsequent scheduled Mandatory Reduction in the amount of the Total Commitment pursuant to the definition of "Total Commitment", and, effective as of any Voluntary Reduction or unscheduled Mandatory Reduction, the amount of the Total Commitment set forth in the definition of "Total Commitment" for any time after such Voluntary Reduction or unscheduled Mandatory Reduction (if applicable, as previously reduced in respect of any prior Voluntary Reduction or unscheduled Mandatory Reduction) shall be deemed to have been amended so as to reduce such amount by the amount of such Voluntary Reduction or unscheduled Mandatory Reduction.

2.4. Repayment of Principal.

- 2.4.1. Maturity Date. Subject to Section 2.4.2, the Borrower shall repay to the Agent, for the account of the Lenders, the principal amount of each Loan on or before the Maturity Date.
- 2.4.2. Mandatory Payments and Prepayments.
- (a) If at any time the outstanding principal amount of the Loans exceeds the Total Commitment by reason of a scheduled reduction in the Total Commitment or otherwise, the Borrower shall pay to the Agent on such date, for the account of the Lenders and for application to the repayment of the Loans, the amount of such excess.
- (b) Concurrently with each Mandatory Reduction of the Total Commitment pursuant to Section 2.3.2, the Borrower shall prepay the principal of the Loans in an amount equal to the lesser of (a) the amount of such Mandatory Reduction or (b) the outstanding principal amount of the Loans.
- (c) (i) If, at the time of any payment or prepayment of principal, there are outstanding any Swing Line Borrowings, such payment or prepayment of principal shall be applied

first to the payment or prepayment, as applicable, of outstanding Swing Line Borrowings, and then in accordance with the following subparagraph (ii).

subparagraph (i): unless otherwise specified by the Borrower (solely with respect to voluntary prepayments) in writing at the time of any payment or prepayment of principal, each payment or prepayment of principal shall be applied first to the payment or prepayment, as applicable, of outstanding Base Rate Loans, and second to the payment or prepayment, as applicable, of outstanding IBOR Rate Loans, in the order in which their respective Applicable Interest Periods terminate (and, in the case of IBOR Loans having Applicable Interest Periods ending on the same date, pro rata in accordance with their outstanding principal amounts or, if no Event of Default has occurred and is continuing and the Borrower so directs at the time of such payment, in such manner as the Borrower may direct to minimize the Borrower's liability under Section 2.8).

Any payment or prepayment of the principal of the Loans shall be accompanied by the payment of (x) such fees, expenses or other amounts as may then be due and payable under the Loan Documents and (y) accrued but unpaid interest on the principal so paid or prepaid.

- 2.5. Agent's Right to Fund. Unless the Agent shall have received notice from a Lender at both its New York Office and its Dallas Office prior to 2:30 p.m. (New York City time) on the Funding Date of any Revolving Loan that such Lender will not make available to the Agent such Lender's Percentage Interest of the requested Revolving Loan ("such Lender's Loan"), (a) the Agent may assume that such Lender has made such funds available to the Agent on such Funding Date in accordance with Section 2.2 and (b) the Agent may, in reliance upon such assumption, make available to the Borrower on such Funding Date the amount of such Lender's Loan. If and to the extent that (x) such Lender shall not have so made the amount of such Lender's Loan available to the Agent and (y) the Agent shall have advanced the amount of such Lender's Loan to the Borrower, then such Lender and the Borrower severally agree to pay to the Agent, on demand, the amount of such Lender's Loan, together with interest thereon for each day from the date such amount is made available to the Borrower until the date such amount is repaid to the Agent, at (i) in the case of the Borrower, the Applicable Interest Rate, or (ii) in the case of such Lender, the Federal Funds Rate. repayment by the Borrower shall be without prejudice to any rights it may have against the Lender that has failed to make available such Lender's Loan on any Funding Date.
- 2.6. Interest on Loans.
- 2.6.1. General Provisions.

- (a) The Borrower agrees to pay to the Agent, for the account of each Lender, interest on the unpaid principal amount of each Loan from the date of such Loan until such Loan becomes due and payable at a per annum rate equal to the Applicable Interest Rate, and, if default shall occur in the payment when due of all or any portion of the principal of such Loan, from the date such principal becomes due until it is paid in full at a per annum rate equal to two percentage points (2.00%) above the Base Rate (changing as the Base Rate changes, the "Default Rate").
- (b) Accrued but unpaid interest on each IBOR Loan shall be paid on the last day of the Applicable Interest Period, on the date of any payment of the principal of such IBOR Loan (to the extent accrued on the principal amount paid), at the Maturity Date.
- (c) Accrued but unpaid interest on each Base Rate
 Loan shall be paid on the last Business Day of each calendar month
 (the first such payment due August 31, 2000), on the date of any
 payment of the principal of the Base Rate Loans (to the extent
 accrued on the principal amount paid) and at the Maturity Date.
- (d) Unpaid interest accruing on amounts in default shall be payable on demand.
- (e) Notwithstanding anything herein or in any other Loan Document to the contrary, under no circumstances whatsoever shall interest charged in respect of the Loans for the benefit of any Lender, however such interest may be characterized or computed, exceed the highest rate permissible under any law that a court of competent jurisdiction shall, in a final determination, deem applicable to the Loans of such Lender. If such a court determines that any Lender has received interest hereunder in excess of the highest rate applicable to its Loans, such Lender shall, at the Lender's election, either (i) promptly refund such excess interest to the Borrower or (ii) credit such excess to the principal balance of such Lender's Loans. This provision shall control over every other provision of all agreements between the Borrower and any Lender.

2.6.2. Selection of Alternative Rate.

(a) (i) Each Swing Line Borrowing shall be a Base Rate Loan. The Borrower may, subject to the requirements of this Section 2.6.2, on at least three (3) Business Days' prior written notice, elect to have interest accrue on any Revolving Loan or any portion thereof at the IBOR Rate for an Applicable Interest Period. Such notice (an "Interest Rate Notice") shall be deemed delivered on receipt by Agent, except that an Interest Rate Notice received by the Agent at either its New York Office or its Dallas Office after 12:30 p.m. (New York City time) on any Business Day shall be deemed to have been received on the immediately succeeding Business Day.

The Interest Rate Notice may be given with, and contained in, any Notice of Borrowing with respect to a Revolving Loan.

- Each Interest Rate Notice shall identify, subject to the conditions of this Section 2.6.2, the Revolving Loan or, portions thereof and the Applicable Interest Period which the Borrower selects. Any Interest Rate Notice that specifies an IBOR Rate but fails to identify an Applicable Interest Period shall be deemed to be a request for the designated IBOR Rate for an Applicable Interest Period of one (1) month. The Borrower's right to select an IBOR Rate to apply to a Revolving Loan or any portion thereof shall be subject to the following conditions: (A) the aggregate amount of all Revolving Loans or portions thereof to accrue interest at a particular IBOR Rate for the same Applicable Interest Period shall be \$2,000,000 or an integral multiple of \$1,000,000 in excess thereof; (B) the Borrower shall not have selected more than twelve (12) different IBOR Rates or Applicable Interest Periods to be applicable to portions of the Revolving Loans at any one time; (C) an IBOR Rate may not be selected for any Swing Line Borrowing; (D) an IBOR Rate may not be selected for any Revolving Loan or portion thereof that is already accruing interest at an IBOR Rate unless such selection is only to become effective at the maturity of the Applicable Interest Period then in effect; (E) neither the Agent nor any Lender shall have given notice pursuant to Section 2.6.4 that the IBOR Rate selected by Borrower is not available; and (F) no Default or Event of Default shall have occurred and be continuing.
- On receipt of any Interest Rate Notice, the Agent shall promptly notify each Lender by telephone (confirmed promptly by telex or facsimile transmission) of the information set forth in the Interest Rate Notice. Each Interest Rate Notice shall be irrevocable, and shall constitute a representation and warranty by the Borrower that, as of the date of such Interest Rate Notice, no Event of Default or Default has occurred and is continuing. the Borrower delivers an Interest Rate Notice with any Notice of Borrowing for a Loan and the Borrower thereafter declines to take such Loan, or a condition precedent to the making of such Loan is not satisfied or waived, Borrower shall indemnify the Agent, and each Lender, for all losses and any costs which the Agent or any Lender may sustain as a consequence thereof, including, without limitation, the costs of redeployment of funds at rates lower than the cost to the Lenders of such funds. A certificate of the Agent or any Lender setting forth the amount due to it pursuant to this subparagraph (b) and the basis for, and the calculation of, such amount shall, absent manifest error, be conclusive evidence of the amount due pursuant to this subparagraph (b). Payment of the amount owed shall be due within fifteen (15) days after the Borrower's receipt of such certificate.
- (c) Each IBOR Loan shall be allocated among the

Lenders in proportion to their respective Percentage Interests. In the absence of an effective request for the application of an IBOR Rate (including, without limitation, as of the last day of the Applicable Interest Period of any IBOR Loan), the Loans shall accrue interest at the Adjusted Base Rate.

2.6.3. Applicable Days for Computation of Interest.

Computations of interest accruing at an IBOR Rate shall be made on the basis of a year of three hundred sixty (360) days, for the actual number of days (including the first day but excluding the last day) occurring in, the period for which such interest is payable. Computations of interest accruing at the Adjusted Base Rate or at Default Rate shall be made on the basis of a year of three hundred sixty-five or three hundred sixty-six (365 or 366) days, as the case may be, in each case, for the actual number of days (including the first day but excluding the last day) occurring in the period for which such interest is payable.

2.6.4. Unavailable IBOR Rate. If the Agent determines that for any reason fair and adequate means do not exist for establishing the IBOR Rate, or if any Lender determines (and gives notice to the Agent) that accruing interest on any Loan at an IBOR Rate by such Lender has become unlawful, or that the IBOR Rate (exclusive of the Margin) does not accurately reflect the cost to such Lender of making or maintaining IBOR Loans, the Agent shall give notice of that fact to each Lender and to the Borrower, and such determination shall be conclusive and binding absent manifest error. After such notice has been given and until the Agent notifies the Borrower, or such Lender notifies the Borrower and the Agent, that the circumstances giving rise to such notice no longer exist, the IBOR Rate shall no longer be available in respect of Loans, and any request by the Borrower to have interest accrue on a Loan at the IBOR Rate shall be deemed to be a request for interest to accrue at the Adjusted Base Rate; provided, however, that if it is unlawful for the affected Lender to maintain any then outstanding IBOR Loans as IBOR Loans until the end of the relevant Applicable Interest Period, such IBOR Loans shall automatically convert into Base Rate Loans with effect from the date of such Lender's notice to the Agent, and, if the interest rate applicable to Base Rate Loans is then lower than the relevant IBOR Rate(s), the Borrower shall pay to each Lender, within thirty (30) days following demand accompanied by a certificate of such Lender describing the basis therefor and the calculation thereof, such amount or amounts as may be necessary to compensate such Lender for any loss in connection therewith. the circumstances giving rise to the notice described herein cease to exist, the Agent or the relevant Lender, as the case may be, shall, promptly upon becoming aware of that fact, notify the Borrower and (if applicable) the Agent in writing of that fact, and the Borrower shall then once again become entitled to request that the IBOR Rate apply to Loans in accordance with Section 2.6.2.

- 2.7. Compensation for Increased Costs.
- 2.7.1. Borrower's Obligation to Compensate. In the event that after the date hereof any change occurs in any applicable law, regulation, guideline, treaty or directive or interpretation thereof by any authority charged with the administration or interpretation thereof, or any condition is imposed by any such authority after the date hereof or any change occurs in any condition imposed by any such authority on or prior to the date hereof which:
- (a) subjects any Lender to any Tax, or changes the basis of taxation of any payments to any Lender on account of principal of or interest on any IBOR Loan, such Lender's Note (to the extent such Note evidences IBOR Loans) or other amounts payable with respect to IBOR Loans (other than a change in the rate of tax based solely on the overall net or gross income of such Lender); or
- (b) imposes, modifies or determines applicable any reserve, deposit or similar requirements against any assets held by, deposits with or for the account of, or loans or commitments by, any office of any Lender in connection with its IBOR Loans (but only to the extent that any such amount is in excess of, or was not applicable at the time of computation of, the amounts provided for in the definition of the IBOR Rate); or
- (c) imposes upon any Lender any other condition with respect to its IBOR Loans or its obligation to make IBOR Loans; which, as a result thereof, (i) increases the cost to any Lender of making or maintaining its IBOR Loans or its Commitment hereunder, or (ii) reduces the net amount of any payment received by any Lender in respect of its IBOR Loans (whether of principal, interest, commitment fees or otherwise), or (iii) requires any Lender to make any payment on or calculated by reference to the gross amount of any sum received by it in respect of its IBOR Loans; or
- (d) affects the amount of capital required or expected to be maintained by banks generally or corporations controlling banks, and any Lender reasonably determines that the capital such Lender or any corporation controlling such Lender is required or expected to maintain, or the amount by which such Lender or any corporation controlling such Lender is required or expected to increase its capital, is increased by (or based upon), the existence of this Agreement or of such Lender's Loans (whether or not IBOR Loans) or Commitment hereunder by an amount which the affected Lender in its sole but reasonable judgment deems material, then and in any such case the Borrower shall pay to the Agent, for the account of such Lender, on demand, such amount or amounts as will compensate such Lender (on an after-tax basis) for any increased cost, deduction or payment actually incurred or made by such Lender in respect of such circumstances. The demand for

payment by any Lender shall be delivered to both the Agent and the Borrower and shall state the subjection or change which occurred, or the reserve or deposit requirements or other conditions which have been imposed upon such Lender, or the request, direction or requirement with which it has complied, together with the date thereof, the amount of such cost, reduction or payment and the manner in which such amount has been calculated. The statement of any Lender as to the additional amounts payable pursuant to this Section 2.7 shall, absent manifest error, be conclusive evidence of the amounts payable hereunder.

- 2.7.2. Effect of Invalidity or Inapplicability. protection of this Section 2.7 shall be available to each Lender regardless of any possible contention of invalidity or inapplicability of the relevant law, regulation, guideline, treaty, directive, condition or interpretation thereof. In the event that the Borrower pays to any Lender all or any portion of the amount necessary to compensate such Lender for any charge, deduction or payment incurred or made by such Lender as provided in this Section 2.7, and such charge, deduction, payment, or any part thereof, is subsequently returned to such Lender as a result of the final determination of the invalidity or inapplicability of the relevant law, regulation, guideline, treaty, directive or condition, then such Lender shall remit to the Borrower the amount paid by the Borrower which has actually been returned to such Lender (together with any interest actually paid to Lender on such returned amount), less such Lender's costs and expenses incurred in connection with such governmental regulation or any challenge made by such Lender with respect to its validity or applicability.
- 2.7.3. Failure or Delay. No failure of any Lender to demand compensation for any increased cost or reduction in amounts received or receivable or for any reduction in return on capital shall constitute a waiver of such Lender's right to demand such compensation; provided, however, that no Lender shall be entitled under this Section 2.7 to any such compensation attributable to any period preceding the date that is ninety (90) days prior to the date of such Lender's demand.
- 2.8. Prepayments. Base Rate Loans may be repaid at any time without penalty or premium. If any IBOR Loan (or portion thereof) is paid prior to the end of the relevant Applicable Interest Period, a fee computed in the manner set out in Schedule 1 shall be assessed and paid at the time of such prepayment. Such fee shall apply in all circumstances where such a Loan (or portion thereof) is paid prior to the end of the Applicable Interest Period, regardless of whether such payment is voluntary, mandatory, or the result of acceleration or of the Agent's or Lenders' collection efforts.
- 2.9. Notes. The Loans made by each Lender shall, if

requested by such Lender, be evidenced by a promissory note of the Borrower, substantially in the form of Exhibit A hereto, with appropriate insertions, dated as of the date hereof and payable to the order of such Lender, in the face amount of such Lender's Commitment (all promissory notes so requested and issued, the "Notes"). Each Lender is hereby authorized to record the date and the amount of each Loan it makes, and the date and amount of each payment of principal and interest thereon, (i) on a schedule annexed to its Note, if any, or maintained in connection therewith or (ii) otherwise in such Lender's records. Any such recordation by any Lender shall constitute prima facie evidence of the accuracy of the information so recorded; provided, however, that neither the failure to make any such recordation, nor any error in any such recordation, shall affect the Obligations. The election by any Lender not to request the execution and delivery of a Note, or its surrender of a Note for cancellation without requesting the issuance of a replacement Note, shall in no way impair or otherwise adversely affect the Borrowers' Obligations to such Lender in respect of the Loans, which Obligations shall be evidenced (or continue to be evidenced) by the Loan Documents. References in the Loan documents to amounts payable under any Lender's Note shall mean, with respect to a Lender to whom no Note is then outstanding, amounts payable to such Lender under the Loan Documents. In connection with any permitted assignment of an interest in the Loans by a Lender to whom no Note is then issued, then, notwithstanding the provisions of Section 9.4.5(b) requiring the surrender by the assigning Lender of its Note, the assignee shall be entitled, upon request, to the issuance to it of a Note reflecting the amount of its Commitment.

2.10. Manner of Payments.

- 2.10.1. Time, Place and Manner. All payments and prepayments of principal and interest on any Loan, and all other amounts payable hereunder or under any other Loan Document by the Borrower to the Agent or any Lender, shall be made by paying the same in United States dollars and in immediately available funds to the Agent at its Dallas Office (or at such other place as the Agent may from time to time specify for such purpose) not later than 11:30 a.m. (Dallas, Texas time) on the date on which such payment or prepayment is due.
- 2.10.2. Statements. At least two (2) Business Days prior to each date on which any regularly scheduled installment of interest is due in respect of the Loans, the Agent shall give written notice (which may be by telegram, facsimile transmission, cable or telex) of the amounts that will be due on such date. The calculation of the amounts due will be made on the assumption that no new extensions of credit or payments will be made between the date of such notice and the date on which such installment of interest is due, and that there will be no changes in the relevant interest rate(s). Notwithstanding the foregoing to the contrary,

amounts due on any date on which interest is payable hereunder shall be calculated in accordance with this Agreement, and the failure of the Agent to give notice as provided in this Section 2.10.2 shall not affect the Obligations.

- 2.10.3. Payments on Days Other than Business Days.

 Whenever any payment hereunder or under any other Loan Document shall be stated to be due, and whenever the last day of any Applicable Interest Period would otherwise occur, on a day other than a Business Day, such payment shall be made, or the last day of such Applicable Interest Period shall occur, as the case may be, on the next succeeding Business Day, and such extension of time shall in such case be included in the computation and payment of interest or facility or commitment fees, as the case may be, unless such extension would cause such payment to be made, or the last day of such Applicable Interest Period to occur, in the next following calendar month, in which case such payment shall be due, or the last day of such Applicable Interest Period shall occur, as the case may be, on the next preceding Business Day.
- 2.10.4. Application of Payments. Any payment made by the Borrower hereunder shall be applied first, against fees, expenses and indemnities then due and payable hereunder; second, against accrued and unpaid interest; and thereafter, against the principal of the Loans, in accordance with Section 2.4.2(c).

2.11. Fees.

- 2.11.1. Unused Commitment Fee. The Borrower agrees to pay to the Agent, for the account of the Lenders in proportion to their Percentage Interests, an unused commitment fee, computed daily at the rate of 0.50% per annum for the period commencing on September 5, 1997 and ending on the Maturity Date, on the difference between (a) the Total Commitment and (b) the aggregate outstanding principal amount of all Loans (including Swing Line Borrowings), payable in arrears on the last Business Day of each calendar quarter, on the Maturity Date, and on demand after the occurrence of any Default or Event of Default but prior to termination of the Commitments. Computations of the unused commitment fee shall be made on the basis of a year of three hundred sixty (360) days for the actual number of days (including the first day but excluding the last day) occurring in the period for which such fee is payable.
- 2.11.2. Loan Fees. On or before the Closing Date, the Borrower agrees to pay to the Agent, for its own account, the fees provided in accordance with that certain letter agreement dated as of June 20, 2000, between the Borrower and the Agent.
- 2.11.3. Agency Fees. The Borrower shall pay to the Agent, for its own account, on September 5, 1997 and each anniversary thereof during the term of the Loans, an agency fee in

an amount equal to the product of (a) \$2,500 and (b) the number of Lenders party to the Loan Agreement as of September 5, 1997 or such anniversary, as the case may be.

- 2.11.4. Utilization Fee. The Borrower shall pay to the Agent, for the account of the Lenders in proportion to their Percentage Interests, a utilization fee, computed daily at the rate of 0.375% per annum on the aggregate unpaid principal amount of the Loans, payable in arrears on the last Business Day of each calendar quarter, on the Maturity Date, and on demand after the occurrence of any Default or Event of Default but prior to termination of the Commitments. Computations of the utilization fee shall be made on the basis of a year of three hundred sixty (360) days for the actual number of days (including the first day but excluding the last day) occurring in the period for which such fee is payable.
- 2.11.5. Facility Restatement Fee. The Borrower shall pay to the Agent on the last day of each three-month period commencing on the Closing Date for account of the Lenders in proportion to their Percentage Interests, a facility restatement fee computed daily at the rate of 1.50% per annum on the average daily Total Commitment as in effect from time to time; provided that the rate at which the facility restatement fee is calculated shall be reduced by an amount equal to 1/4 of 1% per annum for each permanent reduction of the Total Commitment of \$25,000,000 (other than in connection with scheduled Mandatory Reductions).
- 2.11.6. Step-up Fee. The Borrower shall pay to the Agent for the account of the Lenders in proportion to their Percentage Interests, on the last day of each three-month period commencing on December 5, 2001, a step-up fee computed daily in the amount of 3% per annum on the average daily Total Commitment, as in effect from time to time.
- 2.11.7. Nonrefundable. Each of the fees provided for under this Section 2.11 shall be fully earned as of the day it is due hereunder, and shall be nonrefundable.
- 2.12. Sharing of Payments, Etc. Each borrowing of Loans from the Lenders under Section 2.1 shall be made pro rata in accordance with each Lender's Percentage Interest. Each payment and prepayment of the Loans, and each payment of interest on the Loans, shall be made pro rata to each Lender in accordance with its Percentage Interest. If any Lender shall obtain any payment in respect of the Borrower's Obligations, under this Agreement or the Notes, or any other Loan Document (whether voluntary or involuntary, through the exercise of any right of setoff or otherwise), in excess of the share which it would have been entitled to receive had such payment been made to the Agent, such Lender shall forthwith notify the Agent and shall purchase from the other Lenders such participations in the Loans made by them as shall be necessary to

cause such purchasing Lender to share the excess payment ratably with each of them; provided that if all or any portion of such excess payment is afterward recovered from such purchasing Lender, the purchase shall be rescinded and the purchase price restored, without interest, to the extent of such recovery. The Borrower authorizes the purchase of such participations and agrees that any Lender so purchasing a participation from another Lender may exercise all of its rights to payment (including the right of set off) with respect to such participation as fully as if such Lender were the direct creditor of the Borrower (in addition to in respect of its own Loans) in respect of such participation.

2.13. Taxes.

Subject to Section 2.13(b), all payments made by (a) the Borrower under this Agreement and the Notes shall be made free and clear of, and without deduction or withholding for or on account of, any present or future income, stamp or other taxes, levies, imposts, duties, charges, fees, deductions or withholdings, now or hereafter imposed, levied, collected, withheld or assessed by any Governmental Authority, excluding, in the case of the Agent and each Lender, income taxes and franchise taxes (or other similar taxes imposed as a result of being incorporated, having an office or doing business in a particular jurisdiction) imposed on the Agent or any Lender, as the case may be, as a result of a present or former connection between the jurisdiction of the government or taxing authority imposing such tax and the Agent or such Lender (excluding a connection arising solely from the Agent or such Lender having executed, delivered or performed its obligations or received a payment under, or enforced, this Agreement or the Notes) or any political subdivision or taxing authority thereof or therein (all such non-excluded taxes, levies, imposts, duties, charges, fees, deductions and withholdings being hereinafter called "Taxes"). If any Taxes are required to be withheld from any amounts payable to the Agent or any Lender hereunder or under the Notes, and subject to the Agent's or such Lender's compliance with this Section 2.13, the amounts so payable to the Agent or such Lender shall be increased to the extent necessary to yield to the Agent or such Lender (after payment of all Taxes) interest or any such other amounts payable hereunder at the rates or in the amounts specified in this Agreement and the Notes. Whenever any Taxes are payable by the Borrower, as promptly as possible thereafter the Borrower shall send to the Agent for its own account or for the account of such Lender, as the case may be, a certified copy of an original official receipt received by the Borrower showing payment thereof. If any Lender shall become entitled to any tax credit or receive a refund from any Governmental Authority, in either case, directly attributable to the payment by the Borrower of any Taxes, such Lender shall return to the Agent for the benefit of the Borrower an amount equal to the lesser of (x) the amount of such credit or refund, as the case may be, and (y) the amount of the Taxes so paid by the Borrower that gave rise to such

credit or refund. Any payment to be made by a Lender to the Borrower pursuant to the preceding sentence shall be made at the time, and solely to the extent, that (i) in the case of a refund, such refund is actually received by the Lender from the applicable Governmental Authority, and (ii) in the case of a credit, a payment of Taxes actually made by the Lender to a Governmental Authority is reduced as a direct result of such credit. If the Borrower fails to pay any Taxes when due to the appropriate taxing authority or fails to remit to the Agent the required receipts or other required documentary evidence, the Borrower shall indemnify the Agent and the Lenders for any incremental taxes, interest or penalties that may become payable by the Agent or any Lender as a result of any such failure. The agreements in this Section 2.13 shall survive the payment of the Notes and all other Obligations.

Each Lender that is not a United States person (as such term is defined in Section 7701(a)(30) of the Code) agrees to deliver to the Borrower and the Agent on or prior to the Closing Date, or in the case of a Lender that is an assignee or transferee of an interest under this Agreement pursuant to Section 9.4 (unless the respective Lender was already a Lender hereunder immediately prior to such assignment or transfer), on the date of such assignment or transfer to such Lender, (i) (x) two accurate and complete original signed copies of Internal Revenue Service Form W-8BEN or W-8EC1 (or successor forms) certifying to such Lender's entitlement to a complete exemption from United States withholding tax with respect to payments to be made under this Agreement and under any Note and (y) for periods prior to January 1, 2001, a duly completed copy of Internal Revenue Service Form W-8 or W-9 (or successor form), or (ii) if the Lender is not a "bank" within the meaning of Section 881(c)(3)(A) of the Code and cannot deliver either Internal Revenue Service Form W-8BEN or W-8EC1 pursuant to clause (i) above, (x) a certificate in form and substance satisfactory to the Agent (any such certificate, a "Section 2.13(b)(ii) Certificate") and (y) two accurate and complete original signed copies of Internal Revenue Service Form W-8 (or successor form) certifying to such Lender's entitlement to a complete exemption from United States withholding tax with respect to payments of interest to be made under this Agreement and under any Note. In addition, each Lender agrees that from time to time after the Closing Date, when a lapse in time or change in circumstances renders the previous certification obsolete or inaccurate in any material respect, it will deliver to the Borrower and the Agent two new accurate and complete original signed copies of Internal Revenue Service Form W-8BEN or W-8EC1 and Form W-8 and W-9, or Form W-8 and a Section 2.13(b)(ii) Certificate, as the case may be, and such other forms as may be required in order to confirm or establish the entitlement of such Lender to a continued exemption from or reduction in United States withholding tax with respect to payments under this Agreement and any Note, or it shall immediately notify the Borrower and the Agent of its inability to deliver any

such form or certificate. Notwithstanding anything to the contrary contained in Section 2.13(a), but subject to Section 9.4.3 and the immediately succeeding sentence, (x) the Borrower shall be entitled, to the extent it is required to do so by law, to deduct or withhold income or similar taxes imposed by the United States (or any political subdivision or taxing authority thereof or therein) from interest, fees or other amounts payable hereunder for the account of any Lender which is not a United States person (as such term is defined in Section 7701(a)(30) of the Code) for U.S. Federal income tax purposes to the extent that such Lender has not provided to the Borrower U.S. Internal Revenue Service Forms that establish a complete exemption from such deduction or withholding and (y) the Borrower shall not be obligated pursuant to Section 2.13(a) hereof to gross-up payments to be made to a Lender in respect of income or similar taxes imposed by the United States if (I) such Lender has not provided to the Borrower the Internal Revenue Service Forms required to be provided to the Borrower pursuant to this Section 2.13(b) or (II) in the case of a payment, other than interest, to a Lender described in clause (ii) above, to the extent that such forms do not establish a complete exemption from withholding of such taxes. Notwithstanding anything to the contrary contained in the preceding sentence or elsewhere in this Section 2.13(b), and except as set forth in Section 9.4.3, the Borrower agrees to pay additional amounts and to indemnify each Lender in the manner set forth in Section 2.13(a) (without regard to the identity of the jurisdiction requiring the deduction or withholding) in respect of any amounts deducted or withheld by it as described in the immediately preceding sentence as a result of any changes after September 5, 1997 in any applicable law, treaty, governmental rule, regulation, guideline or order, or in the interpretation thereof, relating to the deducting or withholding of Taxes.

Certain Acknowledgments by Borrower. The Borrower recognizes and acknowledges the credit facility made available hereunder is being established and will be maintained in the manner provided herein and in the other Loan Documents at the express request of the Borrower, to accommodate the administrative and operational requirements of the Borrower, its existing Subsidiaries and such Persons as may, at any time during the continuance hereof, become Subsidiaries. Specifically, the credit facility might have been established to provide for direct borrowings by Borrower and by each Subsidiary, subject to individual borrowing limits consistent with the Lenders' prudent lending practices, based on each such Person's borrowing capacity, with additional credit needs of such Person in excess of such borrowing limit being accommodated by loans made to such Person by the Borrower or other Subsidiaries having excess borrowing capacity. For administrative and operational reasons imposed by the Borrower as aforesaid, however, the credit facility provided hereby has been established and will be maintained as described above, but with the intention (but without limiting in

any manner the obligations of Borrower to repay any and all Loans in accordance with the terms hereof and of the Notes) that the Borrower and each of the Subsidiaries ultimately share, among themselves, repayment obligations under the Loan Documents to the same extent as if such borrowings had been made under the alternative, individualized arrangement described above. In addition, it is further recognized and acknowledged that Borrower, and each Subsidiary, will directly and indirectly benefit from the Borrower's and its Subsidiaries' collective business activities, as facilitated by the Loans.

2.15. Designated Senior Indebtedness. The Borrower acknowledges and agrees that the Obligations are intended to and shall constitute "Designated Senior Indebtedness" for all purposes of the High Yield Notes and the Indenture.

ARTICLE III.

CONDITIONS OF LENDING

- 3.1. Conditions to Effectiveness Availability. The effectiveness of this Agreement is subject to the satisfaction of the following conditions precedent on or before August 4, 2000:
- 3.1.1. Execution and Delivery of Loan Documents. The Agent shall have received each of the following, in each case duly executed and delivered by the Borrower:
- (a) this Agreement;
- (b) the Borrower Security Agreement;
- (c) the Hollywood Management Guaranty;
- (d) the Hollywood Management Security Agreement;
- (e) the Amendment to Trademark Mortgage and Security Agreement, dated as of August 2, 2000, between the Borrower and the Agent;
- (f) the Notes; and

each of the Agent and the Lenders shall have received, in form and substance satisfactory to it, the Borrower Disclosure Letter.

- 3.1.2. Corporate Authority. The Agent shall have received in form and substance reasonably satisfactory to it:
- (a) a certified copy of resolutions adopted by the Board of Directors of the Borrower authorizing the execution, delivery and performance of the Loan Documents; and

- (b) evidence of the authority and specimen signatures of the individuals who have signed the Loan Documents on behalf of the Borrower, together with such other evidence of corporate existence, authority or good standing in any jurisdiction as the Agent or any Lender may reasonably require.
- 3.1.3. Legal Opinion. The Agent shall have received a written legal opinion of Stoel Rives LLP, in form and substance reasonably satisfactory to the Agent, addressed to the Agent and the Lenders and addressing such matters as the Agent or any Lender may reasonably require.
- 3.1.4. Perfected Security Interests. The Agent shall have received evidence, reasonably satisfactory to it, that the security interests created by each of the Security Agreements have been duly perfected by the filing of all Financing Statements, and the taking of all such other or additional acts as the Agent may reasonably deem necessary or advisable to create a valid and perfected Lien in the Collateral enforceable in all jurisdictions to secure the Obligations. The Agent shall have also received such evidence as it may reasonably require (a) that its Liens in the Collateral have priority over any and all other Liens therein except for (i) purchase money Liens covering only assets purchased by the Borrower in the ordinary course of business and permitted to exist pursuant to Section 6.4, (ii) Liens of lessors under Capital Leases permitted to exist pursuant to Section 6.4, including, without limitation, Capital Leases on Schedule 3.1.4. and (iii) other Liens, in existence as of the date hereof, permitted to exist pursuant to Section 6.4 and (b) that the Collateral is free and clear of all Liens, except as permitted by this Agreement.
- 3.1.5. Stock Certificates; Stock Powers. The Agent shall have received original stock certificates representing ownership of each of the Borrower's Subsidiaries and the related undated stock powers duly endorsed in blank.
- 3.1.6. Payment of Fees. The Borrower shall have paid all accrued and unpaid fees, costs and expenses due hereunder and under that certain letter agreement referred to in Section 2.11.2.
- 3.1.7. Wattles Collateral. The Agent shall have received the Wattles Assignment Agreement, duly executed by the Borrower, by which the Borrower transfers as Collateral pursuant to the Borrower Security Agreement, all of its right, title and interest in and to the securities pledged by Mark J. Wattles to the Borrower under the Wattles Pledge Agreement.
- 3.1.8. Blocked Account Agreement. The Agent shall have received duly executed counterparts from Bank of America, N.A.,

Hollywood Management and the Borrower of the Blocked Account Agreement which shall be in full force and effect.

- 3.1.9. Perfection Certificate. The Agent shall have received a completed perfection certificate, certified by an Authorized Officer of the Borrower, containing information requested by, and satisfactory to, the Agent, in its sole discretion.
- 3.1.10. Amendment Fee. The Agent shall have received from the Borrower an amendment fee for the account of each Lender which timely executes this Agreement, in an amount equal to 1/8 of 1% of each such Lender's Commitment.
- 3.1.11. Other Information. The Agent, and each Lender, shall have received such other statements, opinions, certificates, documents and information with respect to the matters contemplated by this Agreement as it may reasonably request.
- 3.2. Conditions Precedent to all Loans. The obligation of each Lender to deliver its Loan proceeds to the Agent, and the obligation of the Agent to disburse such proceeds to the Borrower on the date of any Loan hereunder, are subject to the satisfaction, as reasonably determined by the Agent, of the following conditions precedent:
- 3.2.1. Notice of Borrowing. The Borrower shall have timely delivered to the Agent a Notice of Borrowing with respect to the requested Loan.
- 3.2.2. Absence of Defaults, Accuracy of Representations. At the date of each Loan:
- (a) no Default or Event of Default shall have occurred and be continuing or will occur as a result of the making of the Loan; and
- (b) each of the representations and warranties of the Borrower or any Subsidiary set forth in any Loan Document shall be true and correct in all material respects on the date of such Loan (except to the extent such representations or warranties are conditioned by reference to a specific date, in which case they shall be true and correct in all material respects as of such date), with the same effect as though such representations and warranties had been made on and as of such date.

The Borrower's delivery of any Notice of Borrowing shall be deemed the Borrower's representation and warranty that, as of the date of such Notice of Borrowing and as of the date of the requested Loan, each of the foregoing conditions will be satisfied.

ARTICLE IV.

Borrower represents and warrants to Agent and the Lenders as follows, which representations and warranties shall survive the execution of this Agreement except as set forth in the Borrower Disclosure Letter (which indicates which sections of this Agreement are qualified by the disclosures set forth therein, provided that inadvertent failure to indicate all sections of this Agreement that a particular disclosure is intended to qualify shall not constitute a breach hereunder):

- 4.1. Corporate Existence and Power. The Borrower is a corporation duly incorporated, validly existing and active under the laws of the State of Oregon. The Borrower is duly qualified to do business in each other jurisdiction where the nature of its activities or the ownership of its properties requires such qualification, except where the failure to so qualify has not had and is not reasonably likely to have a Material Adverse Effect on the Borrower. The Borrower has full corporate power, authority and legal right to carry on its business as presently conducted, to own and operate its properties and assets, and to execute, deliver and perform each of the Loan Documents.
- 4.2. Corporate Authorization. The execution, delivery and performance by the Borrower of this Agreement and each of the other Loan Documents, and any borrowing hereunder or thereunder, have been duly authorized by all necessary corporate action of the Borrower, do not require any shareholder approval or the approval or consent of any trustee or the holders of any Indebtedness of the Borrower, except such as have been obtained (certified copies thereof having been delivered to the Agent), do not contravene any law, regulation, rule or order binding on it or its Articles of Incorporation or Bylaws and do not contravene the provisions of or constitute a default under any indenture, mortgage, contract or other agreement or instrument to which the Borrower is a party or by which the Borrower or any of its properties may be bound or affected.
- 4.3. Government Approvals, Etc. No Government Approval or filing or registration with any Governmental Authority is required for the execution, delivery and performance by the Borrower of the Loan Documents or in connection with any of the transactions contemplated thereby (including the Wattles Loan), except such as have been heretofore obtained and are in full force and effect (certified copies thereof having been delivered to the Agent).
- 4.4. Binding Obligations, Etc. This Agreement has been duly executed and delivered by the Borrower and constitutes, and each of the other Loan Documents when duly executed and delivered will constitute, the legal, valid and binding obligations of the

Borrower and its Subsidiaries, as the case may be, enforceable against the Borrower and its Subsidiaries, as the case may be, in accordance with their respective terms.

- 4.5. Litigation. Except as reflected in the financial statements referred to in Section 4.6 or in the Borrower Disclosure Letter, there are no actions, proceedings, investigations, or claims against or affecting the Borrower or any Subsidiary now pending before any arbitrator or court or other Governmental Authority (nor, to the knowledge of the Borrower has any thereof been threatened) which, if determined adversely to the Borrower or such Subsidiary, would be reasonably likely to have a Material Adverse Effect.
- Financial Condition. The consolidated balance sheet of the Borrower and its Subsidiaries as at March 31, 2000, and the consolidated statements of operations and cash flows for the period then ended, copies of which have been furnished to the Agent, fairly present the financial condition of the Borrower and its Subsidiaries as at such date, all determined in accordance with GAAP. Neither the Borrower nor any of its Subsidiaries have on such date any contingent liabilities for Taxes, unusual forward or long-term commitments or unrealized or anticipated losses from any unfavorable commitments, except as referred to or reflected or provided for in the balance sheets or the notes thereto. Since the date of such financial statements there has been no change in the financial condition, operations, or business of the Borrower and its Subsidiaries, taken as a whole, which has resulted, or could reasonably be expected to result, in a Material Adverse Effect (other than in connection with the merger of Reel.com into the Borrower).
- 4.7. Indebtedness; Title and Liens.
- 4.7.1. Neither the Borrower nor any Subsidiary is liable in respect of any Indebtedness other than in respect of (i) the High Yield Notes; (ii) to the extent that they constitute Indebtedness, obligations described in clauses (i) through (iv) of the definition of "Lien"; (iii) accounts payable and other liabilities that, in each case, do not constitute Funded Debt and have been incurred in the ordinary course of business; (iv) the Obligations; and (v) other Indebtedness permitted under Section 6.2. The Obligations are prior in right of payment to any and all obligations of the Borrower in respect of the High Yield Notes, as and to the extent provided in the Indenture.
- 4.7.2. The Borrower, and each of its Subsidiaries, as the case may be, has good and marketable title to each of the properties and assets reflected in the balance sheets referred to in Section 4.6 (except for such assets as (i) have been since sold or otherwise disposed of in the ordinary course of business or (ii) that, in the aggregate, are insignificant to the proper conduct

of the Borrower's or such Subsidiary's business). No assets or revenues of the Borrower or any Subsidiary are subject to any Lien except as permitted by this Agreement. Except to the extent that failure to be so in compliance has not had, and cannot reasonably be expected to have, a Material Adverse Effect: (x) all properties of the Borrower and each Subsidiary, and such Person's use thereof, comply in all material respects with applicable zoning and use restrictions and with applicable laws and regulations relating to health, safety and the environment; and (y) without limiting the generality of the foregoing clause (x), the Borrower, and each Subsidiary, is in compliance in all material respects with all laws and regulations relating to pollution, hazardous substances and environmental control in all jurisdictions in which the Borrower or such Subsidiary, as the case may be, is doing business. Borrower, in the ordinary course of business, conducts a review of the effect of existing laws and regulations pertaining to the environment and existing claims advanced under or in respect of environmental laws and regulations on its business, operations and properties and those of its Subsidiaries, and, as a result thereof, the Borrower has reasonably concluded that such environmental laws and claims have not had, and cannot reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

- 4.8. Taxes. Each of the Borrower and its Subsidiaries
 (a) has filed all United States federal and Oregon state income tax returns and reports required of it; (b) except to the extent that the failure to do so has not had, and is not reasonably likely to have, a Material Adverse Effect, (i) has filed all other tax returns and reports required of it and (ii) has paid all Taxes which are due and payable, and (b) has provided adequate reserves for payment of any Tax the payment of which is being contested. The charges, accruals and reserves on the books of the Borrower or any Subsidiary in respect of taxes for all fiscal periods to date are accurate in all material respects. There are no material questions or disputes between the Borrower or any Subsidiary, and any Governmental Authority with respect to any Taxes except as disclosed in the balance sheet referred to in Section 4.6.
- 4.9. Laws, Orders; Other Agreements. Except to the extent that such occurrence has not had, and is not reasonably likely to have, a Material Adverse Effect, neither the Borrower nor any of its Subsidiaries (a) is in violation of, or subject to any contingent liability on account of, any laws, statutes, rules, regulations and orders of any Governmental Authority, or (b) is in material breach of or default under any agreement to which it is a party or which is binding on it or any of its assets. Each of the Borrower and its Subsidiaries is in material compliance with the Fair Labor Standards Act.
- 4.10. Lien Priority. On the date of any Loan:
- (a) Financing Statements will have been duly filed in all places

where filing is necessary, and all other or additional acts will have been taken as are necessary, to perfect the Agent's Liens on the Collateral; (b) each of the Security Agreements will constitute a valid and perfected Lien on the Collateral to which it relates, enforceable in all jurisdictions, to secure (i) in the case of the Borrower Security Agreement, all Obligations or (ii) in the case of a Subsidiary Security Agreement, the obligations under the Loan Documents of the Subsidiary party thereto; (c) the Agent's Liens on the Collateral have priority over any and all other security interests or other Liens therein except (i) for purchase money Liens, to the extent permitted under Section 6.4(c), on assets purchased by the Borrower or the relevant Subsidiary in the ordinary course of business, (ii) the Liens of lessors under Capital Leases, to the extent permitted under Section 6.4(d), and (iii) other Liens, in existence as of the date hereof, permitted to exist under Section 6.4 and (d) the Collateral is free and clear of all Liens except for Liens permitted under Section 6.4.

4.11. Federal Reserve Regulations. The Borrower is not engaged principally or as one of its important activities in the business of extending credit for the purpose of purchasing or carrying any margin stock (within the meaning of Federal Reserve Regulation U), and no part of the proceeds of any Loan will be used to purchase or carry any such margin stock or to extend credit to others for the purpose of purchasing or carrying any such margin stock, or for any other purpose that violates the applicable provisions of any Federal Reserve Regulation. The Borrower will furnish to the Agent or any Lender, on request, a statement conforming with the requirements of Regulation U.

4.12. ERISA.

- 4.12.1. The present value of all benefits vested under all Pension Plans did not, as of the most recent valuation date of such Pension Plans, exceed the value of the assets of the Pension Plans allocable to such vested benefits by an amount that would (a) represent a potential material liability of the Borrower or (b) affect materially the ability of the Borrower to perform its Obligations.
- 4.12.2. No Plan or any trust created thereunder, or any trustee or administrator thereof, has engaged in a "prohibited transaction" (as such term is defined in Section 406 of ERISA or Section 4975 of the Code) that could subject such Plan or any other Plan, any trust created thereunder, or any trustee or administrator thereof, or any party dealing with any Plan or any such trust, to the tax or penalty on prohibited transactions imposed by Section 502 of ERISA or Section 4975 of the Code.
- 4.12.3. No Pension Plan, or any trust created thereunder, has been terminated, except in accordance with the Code,

ERISA, and the regulations of the Internal Revenue Service and the PBGC as applicable to solvent plans in which benefits of participants are fully protected. No "reportable event", as defined in Section 4043 of ERISA, has occurred for which notice has not been waived or for which alternative notice procedures are permitted.

- 4.12.4. No Pension Plan, or any trust created thereunder, has incurred any "accumulated funding deficiency" (as such term is defined in Section 302 of ERISA), whether or not waived, since the effective date of ERISA.
- 4.12.5. The required allocations and contributions to Pension Plans will not violate Section 415 of the Code.
- 4.12.6. The Borrower has no withdrawal liability to any trust created pursuant to a multiemployer pension or benefit plan, nor would it be subject to any such withdrawal liability in excess of \$1,000,000 if it withdrew from any such plan or if its participation therein were otherwise terminated.
- 4.13. Patents, Licenses, Franchises. The Borrower and its Subsidiaries own or possess all the patents, trademarks, service marks, trade names, copyrights, licenses, franchises, permits and rights with respect to the foregoing necessary to own and operate their properties and to carry on their business, as presently conducted and presently planned to be conducted, without conflict with the rights of others, except to the extent that the failure to do so has not had, and is not reasonably likely to have, a Material Adverse Effect.
- 4.14. New Investment Company, Etc. The Borrower is not now, and after the application by the Borrower of the proceeds of any Loan will not be, subject to regulation under the Investment Company Act of 1940, the Public Utility Holding Company Act of 1935, the Federal Power Act, the Interstate Commerce Act, any state public utilities code or any other federal or state statute or regulation limiting its ability to incur Indebtedness.
- 4.15. Insurance. The properties of the Borrower and of each Subsidiary are insured with financially sound and reputable insurance companies that are not Affiliates of the Borrower, in such amounts, with such deductibles and covering such risks as are customarily carried by companies engaged in similar businesses and owning similar properties in localities where the Borrower or any of its Subsidiaries operates.
- 4.16. Certain Representations Regarding Subsidiaries. As of the date of this Agreement, the only Subsidiaries of the Borrower in existence are listed in the Borrower Disclosure Letter, each of which is wholly-owned by the Borrower. All of the Borrower's shares of each of its Subsidiaries have been validly issued and are fully

paid and nonassessable, and all certificates evidencing such shares have been delivered to the Agent prior to the Closing Date.

- 4.17. Subsidiaries. None of the Borrower's Subsidiaries owns or possesses any assets other than as described on Schedule 4.17.
- 4.18. Hedging Agreements. On the Closing Date there are no Hedging Agreements contemplated or in effect as permitted under Section 6.2(g) to which the Borrower or any of its Subsidiaries is a party.
- 4.19. Lessors. All obligations of the Borrower with respect to any arrangements, by lease or otherwise, with Heller Financial Leasing, Inc. and Transamerica Equipment Financial Services Corporation are listed on Schedule 6.2.
- 4.20. Representations as a Whole. This Agreement, the other Loan Documents, the financial statements referred to in Section 4.6, and all other instruments, documents, certificates and statements furnished to the Agent or any Lender by the Borrower, taken as a whole, do not contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements contained herein or therein not misleading. The Borrower has disclosed to the Lenders in writing any and all facts known to the Borrower's general counsel or to any of the officers of the Borrower listed in the definition of "Officer's Certificate," which could reasonably be expected to have a Material Adverse Effect.

ARTICLE V.

AFFIRMATIVE COVENANTS

So long as any Lender shall have any Commitment hereunder, and until payment in full of each Loan and the Notes and performance of all other Obligations, the Borrower agrees that, unless the Agent (with the consent of the Majority Lenders) shall otherwise give its prior written consent:

- 5.1. Use of Proceeds. The Borrower shall use the proceeds of the Loans for general corporate purposes.
- 5.2. Payment. The Borrower will pay the principal of and interest on the Loans in accordance with the terms of this Agreement and the Notes, and shall pay when due all other amounts payable by the Borrower hereunder or under any other Loan Document.
- 5.3. Preservation of Corporate Existence, Etc. The Borrower (a) shall preserve and maintain, and (except to the extent otherwise expressly permitted hereunder or contemplated by the

Borrower Disclosure Letter) shall cause each Subsidiary to preserve and to maintain, its corporate (or other entity, as the case may be) existence, rights, franchises and privileges in the jurisdiction of its formation, and (b) except where the failure to do so could not reasonably be expected to have a Material Adverse Effect, (i) shall qualify and remain qualified (and shall cause each Subsidiary to qualify and remain qualified) as a foreign corporation (or other entity, as the case may be) in each jurisdiction where such qualification is necessary or desirable in view of its business and operations or the ownership of its properties; and (ii) shall own or possess (and shall cause each Subsidiary to own or possess) all the patents, trademarks, service marks, trade names, copyrights, licenses, franchises, permits and rights with respect to the foregoing necessary to own and operate its properties and to carry on its business, as at any time conducted, without conflict with the rights of others.

- 5.4. Visitation Rights. At any time, and from time to time upon reasonable notice, the Borrower shall permit, and shall cause each of its Subsidiaries to permit, the Agent or any Lender, or the representatives of the Agent or any Lender, to examine and make copies of and abstracts from its records and books of account, to visit its properties and to discuss the affairs, finances and accounts of the Borrower or such Subsidiary with any of its officers, directors or employees, and (provided that the Borrower shall make the arrangements therefor) with the Borrower's independent public accountants.
- 5.5. Keeping of Books and Records. The Borrower shall keep, and shall cause each of its Subsidiaries to keep, adequate records and books of account in which complete entries will be made, in accordance with GAAP, reflecting all of its financial transactions.
- 5.6. Maintenance of Property, Etc. The Borrower shall maintain and preserve, and shall cause each Subsidiary to maintain and preserve, all of its properties (except with respect to properties that, in the aggregate, are insignificant to the proper conduct of the Borrower's business) in good working order and condition, ordinary wear and tear excepted, and will from time to time make (and cause each Subsidiary to make) all needed repairs, renewals, or replacements so that the efficiency of such properties shall be fully maintained and preserved.
- 5.7. Compliance With Laws, Etc. Except to the extent that failure to do so would not result in a Material Adverse Effect:
- (i) the Borrower shall comply, and shall cause each of its Subsidiaries to comply, in all material respects with all laws, regulations, rules, and orders of Governmental Authorities, except any thereof whose validity is being contested in good faith by appropriate proceedings upon stay of execution of the enforcement

thereof and with provision having been made to the satisfaction of the Agent for the payment of any fines, charges, penalties or other costs in respect thereof in the event the contest is determined adversely to the Borrower or the affected Subsidiary, and (ii) without limiting the generality of the foregoing clause (i), the Borrower shall conduct its operations, and keep and maintain its property, in compliance with all environmental laws, regulations, rules and orders (and shall cause each of its Subsidiaries to do so) except, in each case, to the extent that noncompliance would not result in a Material Adverse Effect.

5.8. Other Obligations. The Borrower shall pay and discharge, and shall cause each Subsidiary to pay and discharge, all Indebtedness, Taxes, and other obligations for which the Borrower or such Subsidiary is liable or to which the Borrower's or such Subsidiary's income or property is subject and all claims for labor and materials or supplies which, if unpaid, might become by law a Lien upon assets of the Borrower or any Subsidiary, except (a) any such obligation the validity or amount of which is being contested in good faith (and, if necessary, by appropriate proceedings upon stay of execution of the enforcement thereof) and with provision having been made to the satisfaction of the Agent for the payment thereof in the event the contest is determined adversely to the Borrower and (b) any trade payables, arising from the purchase of inventory, that are paid in accordance with industry practice and prior to the time any collection proceeding therefor is commenced by any vendor.

5.9. Insurance.

- 5.9.1. The Borrower shall keep in force, and shall cause each of its Subsidiaries to keep in force, upon all of their respective properties and operations policies of insurance carried with financially sound and reputable companies in such amounts and covering all such risks as from time to time may be customary in the industry and in the locations in which the Borrower or the relevant Subsidiary is engaged in business and as shall be reasonably satisfactory to the Agent, which insurance shall in any event include:
- (a) insurance against loss or damage to any of the Collateral by fire or any of the risks covered by insurance known as "all risk coverage," in an amount not less than the full replacement cost thereof, each such policy to contain a replacement cost endorsement reasonably satisfactory to the Agent;
- (b) business interruption or loss of income
 insurance;
- (c) commercial general liability insurance against claims for personal injury, including, but not limited to, bodily

injury, death or property damage occurring on, in or about the premises of the Borrower and its Subsidiaries, including any real property and adjoining ways; and

(d) products liability and contractual liability insurance coverage or endorsements.

From time to time, on request, the Borrower will furnish to the Agent certificates of insurance or duplicate policies evidencing such coverages. If the Borrower shall fail to maintain or cause to be maintained any of the foregoing insurance as and when required hereby, or if any such insurance shall be materially reduced, canceled, surrendered or not renewed (and not replaced so as to avoid any lapse in the required coverages), the Agent may (but is not required to do so), in its own name only, for the benefit of itself and the Lenders or for the benefit of the Borrower, or both, obtain all or a portion of such insurance from any carrier selected by the Agent in its sole discretion, and the Borrower shall reimburse the Agent, within fifteen (15) days after the Agent's demand, for any premiums or other costs associated with obtaining such insurance. Sums paid by the Agent hereunder shall bear interest at the Default Rate.

- All insurance required by this Section 5.9 shall be provided by policies written on terms, in amounts, and by companies, reasonably satisfactory to the Agent; provided that (a) the Borrower and its Subsidiaries may maintain a system of selfinsurance in an aggregate amount, and with respect to such risks, as are both (i) customary for prudently managed corporations engaged in the same or similar businesses as the Borrower and its Subsidiaries, owning or operating similar properties as the Borrower and its Subsidiaries, and having a credit standing similar to that of the Borrower and its Subsidiaries, and (ii) reasonably acceptable to the Agent; (b) the Agent, and each Lender, shall be named as additional insured on all liability policies, as its interests may appear; and (c) (i) losses under all policies insuring the Collateral, other than policies insuring property subject to a Lien in favor of another creditor (with priority over the Lien of the Agent), shall be payable to the Agent, as its interests may appear, and (ii) payments by any insurer in respect of business interruption or loss of income coverage at any time that an Event of Default has occurred and is continuing shall be payable to the Agent, in each case, pursuant to a loss payee endorsement reasonably satisfactory to the Agent.
- 5.9.3. At least twenty (20) days prior to the expiration of each such policy, the Borrower shall furnish the Agent with evidence reasonably satisfactory to the Agent of the continuation or replacement of such policy, maintaining in force such coverages as are required by this Agreement. All such policies shall contain provisions to the effect that (a) except in the case

of liability insurance, they shall provide coverage to the Agent and the Lenders notwithstanding any breach by the Borrower or any Subsidiary of the Borrower (or by any predecessor of the Borrower or of any Subsidiary of the Borrower) of any representation or warranty, (b) there shall be no recourse to the Agent or to any Lender for payment of premiums or other amounts with respect thereto, (c) they shall not be subject to co-insurance, (d) they may not be canceled (except for non-payment of premiums or fraud), or amended to reduce any limitation of liability, to increase the amount of any deductible or co-insurance, or to add any exclusions to the coverage provided thereunder, without at least thirty (30) days' prior written notice to the Agent, and (e) they may not be canceled for non-payment of premiums or for fraud without at least ten (10) days' prior written notice to the Agent. Upon the Agent's (or any Lender's) reasonable request, the Borrower shall cause copies of all bills, statements or other documents relating to the foregoing insurance to be sent or mailed directly to the Agent, and shall in any event give notice (with a copy to the Agent) to each insurer providing any such insurance policy of the Agent's security interest therein.

- 5.10. Financial Information. The Borrower will deliver to the Agent and the Lenders:
- 5.10.1. Annual Audited Financial Statements. As soon as available and in any event within ninety (90) days after the end of each fiscal year of the Borrower, the Borrower's complete Report on Form 10-K (together with the consolidated balance sheet of the Borrower as at the end of such fiscal year and the consolidated statements of operations, statements of changes in shareholders' equity and statements of cash flows for such year, accompanied by the audit report thereon by independent certified public accountants of nationally recognized standing selected by the Borrower and approved by the Agent (which approval shall not be unreasonably withheld or delayed, and the Agent's consent to PricewaterhouseCoopers LLP being hereby given) which report shall be prepared in accordance with generally accepted auditing standards and shall be unqualified, together with any related annual report to the Borrower's shareholders.
- 5.10.2. Quarterly Reports.
- (a) As soon as available and in any event within sixty (60) days after the end of each of the first three fiscal quarters of the Borrower, (i) the Borrower's Report on Form 10-Q (including, without limitation, the consolidated balance sheet and statements of operations and statements of cash flows of the Borrower and its Subsidiaries (in each case, unaudited) as of the end of such fiscal quarter (including the fiscal year to the end of such fiscal quarter)) prepared in accordance with GAAP and (ii) an Officer's Certificate to the effect that the unaudited balance sheet

and statements included therein have been prepared in accordance with GAAP and present fairly the financial position and results of operations of the Borrower and its Subsidiaries as at the end of and for such fiscal quarter (subject to normal year-end audit adjustments) and whether, since the then most recent fiscal year-end report delivered pursuant to Section 5.10.1, there has been any change in the financial condition or operations of the Borrower and its Subsidiaries the result of which could reasonably be likely to result in a Material Adverse Effect as shown on the balance sheet as of said date and, if any such change has occurred, the nature thereof.

- (b) As soon as available and in any event within sixty (60) days after the end of each of the Borrower's fiscal quarters, (i) the unaudited statement of financial performance for the Borrower's stores for such fiscal quarter, prepared in a manner consistent with those heretofore delivered to the Lenders and accompanied by an Officer's Certificate containing any changes in the location of assets held or owned by each of the Subsidiaries of the Borrower as of the date of such certificate, all in form and substance reasonably acceptable to the Agent.
- 5.10.3. Quarterly Compliance Certificates. As soon as available and in any event within sixty (60) days after the close of each of the first three fiscal quarters of the Borrower and within one hundred twenty (120) days after the close of each of the Borrower's fiscal years, a compliance certificate of Borrower in form reasonably acceptable to the Agent, (i) certifying that, as of the close of such fiscal quarter no Default or Event of Default had occurred and was continuing, and, (ii) setting forth calculations demonstrating compliance as of the end of such fiscal quarter with the financial and other covenants set forth in Sections 5.15 through 5.20 and Sections 6.2(d), 6.2(e), 6.2(f), 6.2(g), 6.4(c), 6.4(e), 6.6.2, 6.6.3, 6.6.5, 6.6.6, 6.7, 6.11, 6.12 and 6.13.
- 5.10.4. Monthly Store Sales Comparisons. As soon as available and in any event within thirty (30) days after the end of each calendar month, same store sales comparisons of the Borrower for such month and monthly year to date and in comparative form to the previous year, all in a form to be provided by the Agent not later than August 20, 2000 and accompanied by an Officer's Certificate.
- 5.10.5. Budget; Financial Projections.
- (a) As soon as available and in any event not later than November 30 of each fiscal year, the Borrower's business plan, including, without limitation, the Borrower's annual operating budget and financial projections (presented in a quarterly format) for the period beginning on the first day of the next succeeding fiscal year and ending December 31, 2003, in form and substance

satisfactory to the Agent (including, without limitation, any and all major assumptions contemplated therein).

- (b) As soon as available and in any event not later than January 31 of each year, the Borrower's presentation of its business plan described in clause (b) above, to the Lenders.
- 5.10.6. Shareholder, SEC and Government Reports. As soon as available, all reports sent by the Borrower, or any Subsidiary, to its shareholders and all quarterly and annual reports filed by the Borrower, or any Subsidiary, with the Securities and Exchange Commission and each other Governmental Authority having jurisdiction over the Borrower or such Subsidiary.
- 5.10.7. Subordinated Debt. Concurrently with the delivery thereof to any holder of (or trustee for the holders of) any Subordinated Debt, copies of all notices or other communications delivered by the Borrower or any Subsidiary with respect to such Subordinated Debt.
- 5.10.8. Other Information.
- (a) All monthly reports reasonably requested by the Agent, which shall be reviewed by FTI Consulting/Policano & Manzo until such time (after February 28, 2001) that the Agent shall determine, in its sole discretion, that such review is no longer required.
- (b) On a bi-monthly basis, all information pertaining to the financial performance of the Borrower for the two fiscal month period most recently ended (the "Two-Month Period"), past projections related to such Two-Month Period, projections for the two fiscal month period commencing at the end of such Two-Month Period and other information as requested by the Agent, commencing on September 15, 2000 and on the 15th day of each second month thereafter, to be reviewed with the Agent, until such time that the Agent shall determine in its sole discretion that such reviews are no longer required.
- (c) All other statements, reports and other information as the Agent, any Lender or FTI Consulting/Policano & Manzo may reasonably request concerning the financial condition and business affairs of the Borrower.
- 5.11. Notification. Promptly after learning thereof, the Borrower shall notify the Agent and Lenders of (a) the details of any action, proceeding, investigation or claim against or affecting the Borrower or any Subsidiary, instituted before any arbitrator or court or other Governmental Authority or, to the Borrower's knowledge threatened to be instituted, which, if determined adversely would be likely to have a Material Adverse Effect; (b) any

substantial dispute between the Borrower or any Subsidiary and any Governmental Authority; (c) any labor controversy which has resulted in or, to the Borrower's knowledge, threatens to result in a strike which would materially affect the business operations of the Borrower or any Subsidiary; (d) if the Borrower or any member of the Controlled Group gives or is required to give notice to the PBGC of any "reportable event" (as defined in subsections (b)(1), (2), (5) or (6) of ' 4043 of ERISA) with respect to any Plan (or the Internal Revenue Service gives notice to the PBGC of any "reportable event," as defined in subsection (c)(2) of ' 4043 of ERISA, and the Borrower obtains knowledge thereof) which might constitute grounds for a termination of such Plan under Title IV of ERISA, or knows that the plan administrator of any Plan has given or is required to give notice of any such reportable event, the notice of such reportable event given or required to be given to the PBGC; and (e) the occurrence of any Default or Event of Default.

- 5.12. Blocked Account Agreement. The Borrower shall take all actions necessary to maintain the Blocked Account Agreement in full force and effect.
- 5.13. Additional Payments; Additional Acts. From time to time and upon demand by the Agent, the Borrower shall (a) pay or reimburse the Agent and the Lenders for all Taxes imposed on this Agreement and any other Loan Document; (b) obtain and promptly furnish to the Agent evidence of all such approvals of any Governmental Authority as may be required to enable the Borrower to comply with its obligations under the Loan Documents; and (c) execute and deliver (and cause each Subsidiary to execute and deliver) all such other instruments, and perform all such other acts, as the Agent or any Lender may reasonably request to carry out the transactions contemplated by this Agreement and the other Loan Documents (including, without limitation, in connection with the delivery to the Agent of leasehold mortgages and related documents, approvals and filings).
- 5.14. Independent Board of Directors and Committees. The Borrower shall take all actions necessary to timely comply with the NASDAQ Stock Market regulations approved on December 14, 1999 by the Securities and Exchange Commission in connection with independent director and audit committee standards, provided, that if the Borrower shall qualify for the one non-independent director exception, it shall not avail itself of such option. In addition, the Borrower shall take all necessary corporate action such that at the time of compliance with such NASDAQ Stock Market regulations, the compensation committee of the Borrower's Board of Directors is comprised solely of independent directors.
- 5.15. Tangible Net Worth. The Borrower shall maintain at all times a Tangible Net Worth not less than the sum of:

- (b) 75% of the Borrower's cumulative Net Income for those fiscal quarters of the Borrower ended after June 30, 2000, in which the Borrower's Net Income was greater than zero, plus
- (c) 100% of the amount, if any, by which the Tangible Net Worth of the Borrower increases after June 30, 2000, as a result of the issuance of common stock or the conversion of debt securities into common stock in connection with the acquisition of another Person (or of some or all of the assets of another Person), plus
- (d) 90% of the amount, if any, by which the shareholders' equity of the Borrower increases after June 30, 2000, as a result of all other issuances of common stock or conversions of debt securities into common stock; minus
- (e) the amount by which Borrower writes down the value of its inventory after March 31, 2000, in an amount not to exceed \$24,000,000, as reflected in Borrower's financial statements for the year ended December 31, 2000; minus
- (f) the amount of the Borrower's nonrecurring restructuring charge (which may be categorized both in a restructuring charge line item and in cost of goods sold) related to Reel.com after March 31, 2000, in an amount not to exceed \$70,000,000, as reflected in Borrower's financial statements for the three months ended June 30, 2000.
- 5.16. Leverage Ratio. The Borrower shall maintain its Leverage Ratio for the four fiscal quarters ended on the following respective dates such that it shall not exceed the following respective ratios:

Date Leverage Ratio June 30, 2000 5.70 to 1.00 September 30, 2000 5.70 to 1.00 December 31, 2000 4.00 to 1.00 March 31, 2001 3.10 to 1.00 June 30, 2001 2.15 to 1.00 September 30, 2001 1.80 to 1.00 December 31, 2001 1.50 to 1.00 March 31, 2002 1.30 to 1.00 June 30, 2002 1.15 to 1.00 September 30, 2002 1.00 to 1.00;

provided, that if the Borrower incurs Subordinated Debt under Section 6.2(f), the Borrower shall maintain its Leverage Ratio for the four fiscal quarters ended on the following respective dates such that it shall not exceed the following respective ratios:

Date Leverage Ratio June 30, 2000 6.70 to 1.00 September 30, 2000 6.70 to 1.00 December 31, 2000 4.40 to 1.00 March 31, 2001 3.40 to 1.00 June 30, 2001 2.35 to 1.00 September 30, 2001 1.95 to 1.00 December 31, 2001 1.70 to 1.00 March 31, 2002 1.50 to 1.00 June 30, 2002 1.35 to 1.00 September 30, 2002 1.30 to 1.00.

5.17. Interest Coverage Ratio. The Borrower shall maintain its Interest Coverage Ratio for the four fiscal quarters ended on the following respective dates at a level which is not to be less than the following respective ratios:

Date Interest Coverage Ratio
June 30, 2000 1.80 to 1.00
September 30, 2000 1.80 to 1.00
December 31, 2000 2.20 to 1.00
March 31, 2001 2.65 to 1.00
June 30, 2001 and thereafter 3.50 to 1.00

5.18. Interest and Rent Coverage Ratio. The Borrower shall maintain its Interest and Rent Coverage Ratio for the four fiscal quarters ended on the following respective dates at a level which is not to be less than the following respective ratios:

Date Interest Coverage Ratio
June 30, 2000 1.00 to 1.00
September 30, 2000 1.00 to 1.00
December 31, 2000 1.10 to 1.00
March 31, 2001 1.15 to 1.00
June 30, 2001 1.40 to 1.00
September 30, 2001 and thereafter 1.50 to 1.00

5.19. Average Per Store Contribution to Profit. The Borrower shall maintain for each period of four consecutive fiscal quarters of the Borrower an Average Per Store Contribution to Profit of \$90,000. As used herein, "Average Per Store Contribution to Profit" means, for any period of four fiscal quarters of the Borrower, (a) the sum of Borrower's (i) operating income, (ii) general and administrative expense, and (iii) amortization of intangibles for such period (in each case, determined in the same manner in which such amount is determined for purposes of preparing the Borrower's financial statements in accordance with GAAP and in a manner consistent with the financial statements referred to in

Section 4.6, but limited to amounts properly allocable to Stores); divided by the time-weighted average number of such Stores of the Borrower open during such period of four consecutive fiscal quarters (determined in accordance with the Borrower's current practices for determining such time-weighted average, as delivered to the Lenders in connection with this Agreement).

5.20. Adjusted Leverage Ratio. The Borrower shall maintain its Adjusted Leverage Ratio for the four fiscal quarters ended on the following respective dates such that it shall not exceed the following respective ratios:

Adjusted Leverage Ratio June 30, 2000 6.50 to 1.00 September 30, 2000 6.50 to 1.00 December 31, 2000 5.70 to 1.00 March 31, 2001 5.25 to 1.00 June 30, 2001 4.45 to 1.00 September 30, 2001 4.15 to 1.00 December 31, 2001 4.00 to 1.00 March 31, 2002 3.75 to 1.00 June 30, 2002 3.65 to 1.00 September 30, 2002 3.55 to 1.00;

provided, that if the Borrower incurs Subordinated Debt under Section 6.2(f), the Borrower shall maintain its Adjusted Leverage Ratio for the four fiscal quarters ended on the following respective dates such that it shall not exceed the following respective ratios:

Date Adjusted Leverage Ratio June 30, 2000 6.50 to 1.00 September 30, 2000 6.80 to 1.00 December 31, 2000 5.90 to 1.00 March 31, 2001 5.35 to 1.00 June 30, 2001 4.50 to 1.00 4.20 to 1.00 September 30, 2001 December 31, 2001 4.00 to 1.00 March 31, 2002 3.85 to 1.00 June 30, 2002 3.75 to 1.00 3.75 to 1.00. September 30, 2002

- 5.21. Dissolution of Subsidiary. Title Wave Stores, Inc., a Minnesota corporation, shall be legally dissolved no later than September 6, 2000.
- 5.22. Officer's Certificate. On the day which is 15 days prior the date on which an interest payment with respect to any Subordinated Debt is due, the Borrower shall deliver to the Agent an Officer's Certificate which shall certify to the Agent's satisfaction (with all relevant calculations to support such certification attached thereto) that such interest payment when

taken together with (a) the amount necessary to effect the next succeeding Mandatory Reduction and (b) other relevant factors then existing, will not result in a breach of any of the covenants in Sections 5.15 through 5.20.

ARTICLE VI.

NEGATIVE COVENANTS

So long as any Lender shall have any Commitment hereunder, and until payment in full of each Loan and the Notes and performance of all other Obligations, the Borrower agrees that, unless the Agent (with the consent of the Majority Lenders) shall otherwise give its prior written consent:

- 6.1. Liquidation, Merger, Sale of Assets, Fundamental Change. The Borrower shall not (and shall not permit any Subsidiary to):
- (a) merge with or into any other Person, except
 (i) in a transaction in which the Borrower is the surviving entity
 and if, immediately after giving effect to such transaction, the
 surviving entity will have a Net Worth equal to at least the Net
 Worth of the Borrower immediately prior to giving effect to such
 transaction;
- (b) liquidate or dissolve (except (i) as contemplated by the Borrower Disclosure Letter and (ii) that any Subsidiary may liquidate or dissolve into the Borrower);
- (c) enter into any consolidation, joint venture, partnership or other combination with any other Person, except to the extent permitted under Section 6.6.3;
- (d) contract to effect, or consummate, any Asset Disposition except with the prior written consent of the Majority Lenders; or
- (e) create or acquire or otherwise own or possess any Subsidiary not in existence on the Closing Date, without (x) 30 days' prior written notice to the Agent and (y) causing the Agent to receive immediately upon the formation or acquisition thereof, (i) a duly executed Subsidiary Security Agreement which shall grant the Agent a first priority security interest in all of such Subsidiary's assets, (ii) duly executed Financing Statements for filing in jurisdictions as requested by the Agent in its sole discretion and (iii) a Guaranty with respect to such new Subsidiary.

Upon any sale or other transfer, to a Person other than the Borrower or a Subsidiary, of the Borrower's or any Subsidiary's entire interest in any assets effected in accordance with the provisions of

the Loan Documents, then the Agent shall (i) release its Lien on the assets subject to such sale or transfer; and (ii) in the case of the Borrower's sale or transfer of its entire interest in a Subsidiary, release (A) the Agent's pledge of and Lien on the outstanding capital stock of such Subsidiary and its Lien on the assets of such Subsidiary, and (B) the applicable Guaranty.

- 6.2. Indebtedness. The Borrower shall not create, incur, become liable for or suffer to exist (or permit any Subsidiary to create, incur or become liable for or suffer to exist) any Indebtedness other than the following (determined on a consolidated basis):
- (a) the Loans (including any Guaranty);
- (b) Funded Debt, other than Capital Leases, in existence as of the date hereof and identified in Schedule 6.2(a)
- (c) Funded Debt incurred in the ordinary course of the Borrower's business to finance the acquisition of Rental Items, secured solely by a purchase money Lien on the Rental Items so acquired (to the extent such Lien is permitted under Section 6.4(c));
- (d) Funded Debt in the form of Capital Leases (including Capital Leases in existence as of the date hereof and identified in Schedule 6.2(d)) the aggregate principal amount of which (determined in accordance with GAAP and measured in connection with each entry into a Capital Lease after the date hereof, at the time such obligation is incurred), does not exceed an amount equal to 5% of the Borrower's Total Revenues for the period of four fiscal quarters then most recently ended;
- (e) Indebtedness under the High Yield Notes, subject to the limitation that the aggregate original principal amount of the High Yield Notes, whenever issued, shall not exceed \$250,000,000;
- Indebtedness under the High Yield Notes) (i) issued in an aggregate principal amount over the term of the Loans that does not exceed the sum of (A) \$350,000,000 minus (B) the aggregate principal amount of all High Yield Notes, whenever issued; and (ii) that (A) contains subordination provisions substantially identical to those contained in the Indenture (or otherwise acceptable to the Majority Lenders in their sole discretion), (B) matures after the Maturity Date and provides for no amortization of principal until after the Maturity Date, and (C) is unsecured; provided that, no later than twenty (20) Business Days prior to the incurrence thereof, the Borrower delivers to the Agent the Pro Forma Calculations with respect to the incurrence of such Indebtedness;

- (g) obligations of such Person under any Hedging
 Agreement, provided that no such agreement increases the
 Indebtedness of the Borrower at any time except as a result of
 fluctuations in interest rates or by reason of fees, indemnities and
 compensation payable thereunder;
- (h) guaranties permitted to exist under Section 6.3;
- (i) to the extent that they constitute Indebtedness, obligations described in clauses (i) through (iv) of the definition of "Lien";
- (j) accounts payable and other liabilities, in each case, that do not constitute Funded Debt and that have been, or are, incurred in the ordinary course of such Person's business; and
- (k) Indebtedness in connection with Investments under Section 6.6.5 and Section 6.6.6.
- 6.3. Guaranties, Etc. The Borrower shall not (and shall not permit any Subsidiary to) assume, quaranty, endorse or otherwise become or remain directly or contingently liable for, or obligated to purchase, pay or provide funds for the payment of or otherwise assure a creditor against loss in respect of, any obligation or Indebtedness of any other Person, except that (a) the Borrower and its Subsidiaries may endorse negotiable instruments for deposit or collection, or incur similar liabilities in the ordinary course of business and (b) any Subsidiary required, under the terms of the Indenture (as in effect on August 13, 1997, or as amended with the prior written consent of the Majority Lenders), to guaranty the Borrower's obligations with respect to the High Yield Notes may so guaranty such obligations, provided that such Subsidiary's obligations under such guaranty are contractually subordinated to the payment of the Obligations, to the same extent that the Borrower's obligations in respect of the High Yield Notes are subordinated to the Obligations.
- 6.4. Liens. The Borrower shall not create, assume or suffer to exist (or permit any Subsidiary to create, assume or suffer to exist) any Lien on any of its assets, except for (determined, in each case, on a consolidated basis): (a) existing Liens securing liabilities reflected in the balance sheets referred to in Section 4.6 or disclosed in the Borrower Disclosure Letter; (b) Liens in favor of the Agent under the Security Agreements or any other Loan Document; (c) purchase money Liens covering videotapes, video discs or videogames purchased by the Borrower or any Subsidiary in the ordinary course of business; provided that such purchase money Liens do not secure at any time an amount which exceeds 5% of the Borrower's Total Revenues for the period of four fiscal quarters then most recently ended; (d) Liens securing

Indebtedness under Capital Leases incurred in compliance with Section 6.2; (e) additional Liens (including, without limitation, Liens existing on assets at the time they are, or the Person owning them is, acquired by the Borrower or any Subsidiary) which do not at any one time, in the aggregate, secure Indebtedness exceeding an amount equal to five percent (5%) of the Borrower's Net Worth; (f) judgment Liens, to the extent that the existence thereof does not constitute an Event of Default; and (g) Liens securing the extension, renewal or refinancing of the Indebtedness secured by Liens previously existing in compliance with this Section 6.4, provided that (i) each such Lien is limited to the property theretofore encumbered by such Lien (and proceeds thereof) and (ii) the principal amount of the Indebtedness so extended, renewed or refinanced is not increased.

- 6.5. Location of Collateral. The Borrower shall not move, and shall not permit any Subsidiary to move, the location of any Collateral except (a) in the ordinary course of its business and (b) to a jurisdiction within the United States in which the Agent has previously filed UCC-1 financing statements naming the Borrower or Subsidiary, as the case may be, as debtor and perfecting in favor of the Agent a security interest in such Collateral with the priority required by the Agreement.
- 6.6. Investments. The Borrower shall not make or permit to remain outstanding (and shall not permit any Subsidiary to make or permit to remain outstanding) any Investment, except that the Borrower may:
- 6.6.1. Own, purchase or acquire Cash Equivalents;
- 6.6.2. Without limiting the provisions of Section 6.12 or Section 6.13, make any Permitted Acquisition on or after April 1, 2001;
- 6.6.3. Make or permit to remain outstanding a Passive Investment;
- 6.6.4. Transfer Trademarks to Hollywood Management and maintain its cash management system as contemplated under clauses (vi) and (vii), respectively, in the definition of "Asset Disposition."
- 6.6.5. Permit to remain outstanding loans to Mark J. Wattles in the aggregate principal amount not to exceed \$14,900,000 subject to scheduled principal amortization thereof set forth in the Wattles Note; and
- 6.6.6. Make or permit to remain outstanding (a) employee loans (other than to Mr. Wattles), in an aggregate principal amount outstanding at any time not to exceed \$2,675,000

and (b) advances to employees (other than Affiliates) in an aggregate principal amount outstanding at any time not to exceed \$1,000,000;

provided that (x) the cost of any individual Investment or series of related Investments described in Section 6.6.2 and 6.6.3 shall not exceed \$2,000,000 and (y) the aggregate cost of all Investments shall not exceed \$5,000,000 in the aggregate at any time.

- Limitations on Prepayment or Amendment of Subordinated Debt. The Borrower shall not, and shall not permit any Subsidiary to: (a) make any payments in respect of the principal of any Subordinated Debt, (b) make any payment or transfer of assets in respect of the defeasance of any Subordinated Debt, (c) repurchase, redeem or otherwise acquire any Subordinated Debt, (d) make any payment in respect of Subordinated Debt more than one (1) Business Day prior to the date on which it becomes due and payable, (e) at any time that a Default or Event of Default has occurred and is continuing, make any payment of interest or any other amount due in respect of Subordinated Debt, or (f) amend, or consent to any amendment of, the documentation governing any Subordinated Debt (other than the Second Supplemental Indenture, dated as of January 20, 2000, among Borrower and U.S. Trust Company, National Association, as trustee under the Indenture, in the form filed with the Securities and Exchange Commission as Exhibit 10.3 to Borrower's Annual Report on Form 10-K for the fiscal year ended December 31, 1999).
- 6.8. ERISA Compliance. Neither Borrower nor any member of the Controlled Group nor any Plan of any of them shall (a) engage in any "prohibited transaction" (as such term is defined in ' 406 of ERISA or 4975 of the Code); (b) incur any "accumulated funding deficiency" (as such term is defined in ' 302 of ERISA) whether or not waived; (c) terminate any Pension Plan in a manner which could result in the imposition of a Lien on any property of Borrower or any member of the Controlled Group pursuant to ' 4068 of ERISA; or (d) violate state or federal securities laws applicable to any Plan.
- 6.9. Transactions with Affiliates. The Borrower shall not (and shall not permit any Subsidiary to) enter into any transaction with any Affiliate of the Borrower except (a) upon fair and reasonable terms no less favorable to the Borrower or the affected Subsidiary than would obtain in a comparable arm's-length transaction with a Person not an Affiliate of the Borrower; (b) (i) if any such transaction (or series of such transactions) involves aggregate payments in excess of \$1,000,000 in any fiscal year, such transaction(s) is/are: (A) set forth in writing, (B) approved by a majority of the disinterested members of the Borrower's Board of Directors; (c) if any transaction or series of transactions (other than royalty arrangements between the Borrower and Hollywood Management in connection with the Tradename transfer

permitted under Section 6.6.4) involves aggregate payments in excess of \$5,000,000 in any fiscal year, such transaction(s) are additionally determined to be fair, from a financial standpoint, to the Borrower or the relevant Subsidiary by a nationally recognized consulting, accounting, appraisal or investment banking firm and (d) without limiting the foregoing, as permitted under Section 6.6.5.

- 6.10. Change in Business. The Borrower shall not (and shall not permit any Subsidiary to) engage in any material line of business substantially different from those lines of business carried on by the Borrower on the date hereof; except that the Borrower may, either directly or through one or more Subsidiaries, engage in one or more Ancillary Businesses, provided that the sum of Borrower's Aggregate Investments in such Ancillary Businesses, plus Borrower's Aggregate Investments in Passive Investments permitted under Section 6.6.3, does not at any time exceed 5% of the Borrower's Net Worth.
- 6.11. Accounting Change. The Borrower, and each of its Subsidiaries, shall maintain a fiscal year ending on December 31 and shall not make any significant change in accounting policies or reporting practices other than changes required by law.
- 6.12. Capital Expenditures. The Borrower shall not (and shall not permit any of its Subsidiaries to) incur any Capital Expenditures other than those incurred in connection with (x) inventory acquired in the ordinary course of business, (y) repair and maintenance costs, in the ordinary course of business and (z) building or acquiring new Stores, provided, that
- (I) for the period commencing on June 30, 2000 and extending through and including December 31, 2000, the aggregate number of additional Stores permitted under Section 6.13 shall not exceed 65 Stores;
- (II) for the period commencing on January 1, 2001 and extending through and including June 30, 2001, the aggregate number of additional Stores permitted under Section 6.13 shall not exceed an amount equal to the sum of (x) 139 minus (y) additional Stores permitted (for which expenses have been incurred) under clause (I); and
- (III) subject to the limitation set forth in the proviso in Section 2.3.2, the aggregate number of additional Stores permitted under clauses (I) and (II) may be augmented if the Borrower shall have received any Net Cash Proceeds, by (a) for newly built (by the Borrower) Stores, an amount equal to

- (i) 40% of such Net Cash Proceeds divided by (ii) \$500,000 and (b) for acquired Stores, any number of additional Stores for which the maximum cost of each individual acquisition is not greater than \$700,000, provided that the aggregate acquisition cost of all such Stores shall not exceed the sum of (i) 40% of Net Cash Proceeds minus (ii) amounts used under clause (III) (a).
- 6.13. Stores. Other than as provided in Section 6.12(z)(III), the Borrower shall not enter into any new store leases, or make any material expenditure in connection with any new store lease or store opening unless there is already in existence as of the Closing Date a fully-executed store lease; provided that the Borrower may enter into a substitute lease (contemporaneously or within a reasonable time) after terminating an existing lease so long as such substitute lease (x) does not increase the Borrower's total store count and (y) does not contemplate a space which encompasses a materially greater amount of square footage than the original lease.

ARTICLE VII.

EVENTS OF DEFAULT

- 7.1. Events of Default. The occurrence of any of the following events (for whatever reason, and whether voluntarily or involuntarily, by operation of law or otherwise) shall constitute an "Event of Default" hereunder:
- 7.1.1. Loan Payment Default. The Borrower shall fail to pay when due any amount of principal on any Loan; or
- 7.1.2. Other Payment Default. (a) The Borrower shall fail to pay any interest on any Loan and such failure shall remain unremedied for two (2) Business Days, or (b) the Borrower, or any Subsidiary, shall fail to pay any other amount payable by it hereunder or under any Loan Document and such failure shall remain unremedied for five (5) days; or
- 7.1.3. Breach of Warranty. Any representation or warranty made or deemed made by the Borrower or any Subsidiary under or in connection with this Agreement or any other Loan Document, or set forth in any other certificate or document delivered in connection herewith or therewith, shall prove to have been incorrect in any material respect when made or deemed made; or
- 7.1.4. Breach of Certain Covenants; Failure of Security.
- (a) The Borrower shall fail to perform or observe

any covenant set forth in Sections 5.11(e), 5.12, 5.14, 5.15 through 5.22, Article VI or in Section 5 or 8 of the Borrower Security Agreement, or any Subsidiary shall fail to perform any covenant set forth in Section 5 or 8 of the Subsidiary Security Agreement to which it is a party; or

- (b) (i) the Borrower or any Subsidiary shall disavow, or purport to revoke, its obligations under this Agreement or any other Loan Document or (ii) this Agreement, any Security Agreement, any Guaranty or any other Loan Document shall fail to be valid or enforceable in any material respect.
- 7.1.5. Breach of Other Covenants. The Borrower or any Subsidiary shall fail to perform or observe any other covenant, obligation or term of this Agreement or any other Loan Document and such failure shall remain unremedied for thirty (30) days after written notice thereof shall have been given to the Borrower by the Agent; or
- 7.1.6. Material Adverse Changes; Extraordinary
 Situation. There shall occur (a) any event or occurrence having, or
 that could reasonably be expected to have, a Material Adverse
 Effect, or (b) any extraordinary situation, in either case, which
 gives the Lenders reasonable grounds to believe that the Borrower
 may not, or will be unable to, perform or observe in the normal
 course, in any material respect, its obligations under the Loan
 Documents; or

7.1.7. Cross-Default.

- (i) The Borrower, or any Subsidiary, shall fail (a) to pay when due (whether by scheduled maturity, required prepayment, acceleration, demand or otherwise) any principal of any Indebtedness the outstanding principal amount of which, at the time of such failure, exceeds \$2,000,000 but does not exceed \$10,000,000, or any interest or premium thereon, and such failure shall continue after the expiration of the applicable grace period, if any, specified in the agreement or instrument relating to such Indebtedness; or (ii) the Borrower, or any Subsidiary, shall fail to perform any term or covenant on its part to be performed under any agreement or instrument relating to any such Indebtedness and required to be performed, and such failure shall continue after the expiration of the applicable grace period, if any, specified in such agreement or instrument, if the effect of such failure to perform is to accelerate or to legally and in accordance with the applicable documents permit the acceleration of the maturity of such Indebtedness; or
- (b) (i) the Borrower, or any Subsidiary, shall fail to pay when due any amount due in respect of any Indebtedness the outstanding principal amount of which, at the time of such failure,

exceeds \$10,000,000, or (ii) any event shall occur that, with or without the giving of notice or the lapse of time, or both, (A) causes, or would cause, the acceleration of any Indebtedness of the Borrower or any Subsidiary having at the time an outstanding principal amount in excess of \$10,000,000, or (B) entitles, or would entitle, the holders of any such Indebtedness to accelerate, or to require the repurchase, redemption, defeasance, repayment or retirement of, such Indebtedness (including, without limitation, the occurrence of any "Change of Control," as defined in the Indenture), and, in either case (but only if the agreement or instrument governing such Indebtedness specifically provides a grace period with respect to such failure or occurrence), such event continues beyond the earlier of (1) the date that is five (5) Business Days after the occurrence thereof or (2) the expiration of the applicable grace period specified in the agreement or instrument governing such Indebtedness;

provided, however, in any such case, that mere allegations of a default by the Borrower under any agreement or instrument governing any such Indebtedness shall not, without more, be an Event of Default hereunder (it being understood that the delivery by any affected Person, in writing, of a notice of default, or of a notice asserting that a default exists, constitutes more than a mere allegation); or

- 7.1.8. Voluntary Bankruptcy, Etc. The Borrower, or any Subsidiary, shall: (a) file a petition seeking relief for itself under Title 11 of the United States Code, as now constituted or hereafter amended, or file an answer consenting to, admitting the material allegations of or otherwise not controverting, or fail timely to controvert a petition filed against it seeking relief under Title 11 of the United States Code, as now constituted or hereafter amended; or (b) file such petition or answer with respect to relief under the provisions of any other now existing or future applicable bankruptcy, insolvency, or other similar law of the United States of America or any state thereof or of any other country or jurisdiction providing for the reorganization, winding up or liquidation of corporations or an arrangement, composition, extension or adjustment with creditors; or
- 7.1.9. Involuntary Bankruptcy, Etc. (a) An order for relief shall be entered against the Borrower or any Subsidiary under Title 11 of the United States Code, as now constituted or hereafter amended, which order is not stayed within sixty (60) days thereafter; or (b) there shall be entered an order, judgment or decree by operation of law or by a court having jurisdiction in the premises, which is not stayed within sixty (60) days after its entry, (i) (A) adjudging the Borrower or any Subsidiary a bankrupt or insolvent under, or (B) ordering relief against it under, or (C) approving as properly filed a petition seeking relief against it under, the provisions of any other bankruptcy, insolvency or other

similar law of the United States of America or any state thereof or of any other country or jurisdiction (whether now existing or hereafter becoming applicable) providing for the reorganization, winding-up or liquidation of corporations or any arrangement, composition, extension or adjustment with creditors, or (ii) appointing a receiver, liquidator, assignee, sequestrator, trustee or custodian of the Borrower or any Subsidiary or of any substantial part of the property of Borrower or any Subsidiary, or ordering the reorganization, winding-up or liquidation of its affairs; or (c) a period of sixty (60) days shall expire after the filing of any involuntary petition against the Borrower or any Subsidiary seeking any of the relief specified in Section 7.1.8 or this Section 7.1.9 without the petition being dismissed prior to that time; or

- 7.1.10. Insolvency, Etc. The Borrower, or any Subsidiary, shall (a) make a general assignment for the benefit of its creditors, or (b) consent to the appointment of, or taking of possession by, a receiver, liquidator, assignee, trustee, or custodian of all or a substantial part of the property of the Borrower or any Subsidiary, or (c) admit its insolvency or inability to pay its debts generally as they become due, or (d) fail generally to pay its debts as they become due, or (e) take any action (or suffer any action to be taken by its directors or shareholders) looking to the dissolution or liquidation of the Borrower or such Subsidiary (other than a voluntary liquidation, by the Borrower, of a Subsidiary); or
- 7.1.11. ERISA. The Borrower, or any member of the Controlled Group, shall fail to pay when due an amount or amounts aggregating in excess of \$1,000,000 that it shall have become liable to pay to the PBGC or to a Plan under Section 515 of ERISA or Title IV of ERISA; or notice of intent to terminate a Plan or Plans (other than a multiemployer plan, as defined in Section 4001(3) of ERISA), having aggregate Unfunded Vested Liabilities in excess of \$1,000,000 shall be filed under Title IV of ERISA by the Borrower, any member of the Controlled Group, or any plan administrator of any such Plan, or any combination of the foregoing; or the PBGC shall institute proceedings under Title IV of ERISA to terminate any such Plan or Plans; or
- 7.1.12. Judgments; Recorded Tax Liens. (a) One or more final judgments or orders for the payment of money exceeding, in the aggregate, \$2,000,000, or its equivalent in another currency, shall be rendered against the Borrower or any Subsidiary and such judgment or order shall continue unsatisfied and in effect for a period of thirty (30) consecutive days without having being vacated, discharged or satisfied, or stayed or bonded pending appeal; or (b) the filing or recording, by any Governmental Authority with respect to any assets of the Borrower or any Subsidiary, of any one or more notices of lien in respect of unsatisfied, undischarged or

unreleased claims of or Taxes (other than ad valorem real or personal property taxes that are not yet delinquent) aggregating in excess of \$2,000,000, if such condition remains in effect for thirty (30) consecutive days.

7.1.13. Government Approvals, Etc. (a) Any approval of any Governmental Authority or registration or filing with any Governmental Authority now or hereafter required in connection with the performance by the Borrower or any Subsidiary of its obligations under this Agreement or any other Loan Document is revoked, withdrawn or withheld or shall fail to remain in full force and effect; or (b) any Governmental Authority shall take any act that, in the reasonable opinion of the Agent, deprives the Borrower or any Subsidiary of any substantial right, privilege or franchise or substantially restricts the exercise thereof, where the effect of such act, if not revoked or rescinded, could reasonably be expected to have a Material Adverse Effect on the Borrower or any Subsidiary and such act shall not be revoked or rescinded within thirty (30) days after it shall have become effective.

7.2. Consequences of Default.

- 7.2.1. If any of the Events of Default described in Section 7.1.8 or Section 7.1.9 shall occur: (a) the Total Commitment and each Lender's respective Commitment shall immediately terminate, and the principal of and the interest on the Loans and all other sums payable by Borrower hereunder, under the Notes and under the other Loan Documents shall become immediately due and payable, all without protest, presentment, notice or demand, all of which the Borrower expressly waives.
- 7.2.2. If any other Event of Default shall occur and be continuing, then in any such case and at any time thereafter so long as any such Event of Default shall be continuing: (a) the Agent (i) shall, at the request of the Majority Lenders, or (ii) may, with the consent of the Majority Lenders, immediately terminate (by written notice to the Borrower) the Total Commitment and each of the Lenders' respective Commitments, and (b) if Loans shall have been made, the Agent (i) shall, at the request of the Majority Lenders, or (ii) may, with the consent of the Majority Lenders, declare (by written notice to the Borrower) the principal of and the interest on the Loans and the Notes and all other sums payable by the Borrower hereunder or under the Notes or any other Loan Document to be immediately due and payable, whereupon the same shall become immediately due and payable, all without protest, presentment, notice, or demand, all of which the Borrower expressly waives. In addition, regardless of whether the Borrower's obligations to repay the Loans shall have been accelerated pursuant to the preceding sentences, the Agent may, at its option, realize on any or all of the Collateral by exercising any rights or remedies provided in each Security Agreement or in any other Loan Document or

otherwise available under applicable law.

ARTICLE VIII.

THE AGENT

- Authorization and Action. Each Lender hereby (subject to Section 8.6) irrevocably appoints and authorizes the Agent to take such action as agent on its behalf and to exercise such powers under this Agreement and the other Loan Documents as are delegated to the Agent by the terms hereof and thereof, together with such powers as are reasonably incidental thereto. The Agent shall have no duties or responsibilities except those expressly set forth in this Agreement. The duties of the Agent shall be mechanical and administrative in nature; the Agent shall not have by reason of this Agreement a fiduciary relationship in respect of any Lender, and nothing in this Agreement or the other Loan Documents, expressed or implied, is intended to or shall be so construed as to impose upon the Agent any obligations in respect of this Agreement or the other Loan Documents except as expressly set forth herein. As to any matters not expressly provided for by this Agreement, including enforcement or collection of the Loans, the Agent shall not be required to exercise any discretion or take any action, but shall be required to act or to refrain from acting (and shall be fully protected in so acting or refraining) upon the instructions of the Majority Lenders, and such instructions shall be binding upon all Lenders and any holders of any Note; provided, however, that:
- (a) the Agent shall not be required to take any action which exposes the Agent to personal liability or which is contrary to the Loan Documents or applicable law;
- (b) if the Agent so requests, the Agent shall first be indemnified to its satisfaction by the Lenders against any and all liability and expense that may be incurred by it by reason of taking or continuing to take any action upon the instructions of the Majority Lenders;
- (c) without the consent of all Lenders, the Agent shall not:
- (i) (except in connection with a sale or transfer of assets permitted under the terms of the Loan Documents or as otherwise expressly permitted hereunder or under any other Loan Document) release any Collateral or Guaranty;
- (ii) change the Total Commitment (other than changes made pursuant to Section 2.3);
 - (iii) change any Lender's Commitment (other

than changes made pursuant to Section 2.3);

- (iv) change the definition of "Majority
 Lenders";
- (v) change the conditions precedent set forth in Article III; or
- (vi) change the time of payment or rates of interest, the time of payment or amounts of fees, or the timing of payments or amounts of principal due in respect of Loans; and
- the terms of Section 2.5, Section 2.11.3 and (d) this Article VIII shall not be amended without the prior written consent of the Agent (acting for its own account). In the absence of instructions from the Majority Lenders, the Agent shall have the authority (but no obligation), in its sole discretion, to take or not to take any action (unless this Agreement specifically requires the consent of the Lenders or the consent of the Majority Lenders therefor), and any such action or failure to act shall be binding on all the Lenders and on all holders of the Notes; provided, however, that, upon any Lender's request that the Agent deliver a notice to the Borrower pursuant to Section 7.1.5, the Agent shall, if it concurs in such Lender's determination that the Borrower has failed to perform or observe any covenant, obligation or term of any Loan Document described therein, promptly deliver such notice. Each Lender, and each holder of any Note, shall execute and deliver such additional instruments, including powers of attorney in favor of the Agent, as may be necessary or desirable to enable the Agent to exercise its powers hereunder. With respect to matters requiring the consent or approval of all Lenders at any given time, all then existing Defaulting Lenders will be disregarded and excluded, and, for voting purposes only, "all Lenders" shall be deemed to mean "all Lenders other than Defaulting Lenders." Notwithstanding the foregoing, if a Defaulting Lender shall dispute the Agent's determination of the defaulted status of such Lender, such Lender may give written notice to the Agent and each other Lender setting forth the basis upon which such Lender disputes the Agent's determination; under such circumstances, the Lenders shall attempt in good faith to resolve such dispute within the ninety (90)-day period following Agent's original determination, and during such period the suspension of voting rights for such Defaulting Lender referred to in the preceding sentence shall not apply, provided that if such dispute is not resolved within such 90-day period, then such suspension of voting rights shall thereafter apply. The provisions of the immediately preceding sentence shall not alter a condition or any other provision of this Agreement with respect to a failure on the part of any Lender to fund its Percentage Interest of any Loan or otherwise to perform its obligations under the Loan Documents.

- 8.2. Duties and Obligations.
- 8.2.1. Limitations on Obligations. Neither the Agent nor any of its directors, officers, agents or employees shall be liable for any action taken or omitted to be taken by it or any of them under or in connection with this Agreement or any of the other Loan Documents except for its or their own gross negligence or willful misconduct. Without limiting the generality of the foregoing, the Agent:
- (a) may treat each Lender that is a party hereto as the party entitled to receive payments hereunder until the Agent receives written notice of the assignment of such Lender's interest herein, signed by such Lender and made in accordance with the terms hereof and an Assignment Agreement, duly executed by such Lender and its assignee, in substantially the form of Exhibit B (each such agreement, an "Assignment Agreement");
- (b) may consult with legal counsel (including counsel for the Borrower), independent public accountants and other experts selected by it, and shall not be liable for any action taken or omitted to be taken in good faith by it in accordance with the advice of such experts;
- (c) (i) makes no warranty or representation to any Lender, and (ii) shall not be responsible to any Lender for any statements, warranties or representations made in or in connection with this Agreement, the other Loan Documents or in any instrument or document furnished pursuant hereto or thereto;
- (d) shall not have any duty to ascertain or to inquire as to the performance of any of the terms, covenants, or conditions of the Loan Documents on the part of the Borrower or any Subsidiary or as to the use of the proceeds of any Loan or as to the existence or possible existence of any Default or Event of Default;
- (e) shall not be responsible to any Lender for the due execution, legality, validity, enforceability, genuineness, effectiveness, or value of this Agreement or any other Loan Document or of any instrument or document furnished pursuant hereto or thereto; and
- (f) shall incur no liability under or in respect of this Agreement (i) by acting upon any oral or written notice, consent, certificate or other instrument or writing (which may be by telegram, facsimile transmission, cable or telex) believed by it to be (A) genuine and (B) signed or sent by the proper party or parties or (ii) by acting upon any representation or warranty of the Borrower or any Subsidiary made or deemed to be made hereunder.
- 8.2.2. Use of Agents, Employees, Etc. The Agent may

execute any of its duties under this Agreement or any other Loan Document by or through agents, employees or attorneys-in-fact, and shall be entitled to advice of counsel concerning all matters pertaining to such duties. The Agent shall not be responsible for the negligence or misconduct of any agent or attorney-in-fact that it selects with reasonable care.

- 8.2.3. Payments, Deliveries to Lenders. The Agent shall account to each Lender for such Lender's Percentage Interest of payments of principal of, interest on and fees in respect of the Loans (other than fees payable to the Agent for its own account) received by the Agent from the Borrower or any Subsidiary, and shall promptly remit to the Lenders entitled thereto all such payments. The Agent shall transmit to each Lender copies of all documents received from the Borrower or any Subsidiary pursuant to the requirements of this Agreement, other than documents which by the terms of this Agreement the Borrower is obligated to deliver directly to the Lenders.
- 8.3. Dealings Between Agent and Borrower. With respect to its Commitment and the Loans made by it, the Agent shall have the same rights and powers under this Agreement and the other Loan Documents as any other Lender and may exercise the same as though it were not the Agent, and the term "Lender" shall, unless otherwise expressly indicated, include the Agent in its individual capacity. The Agent may accept deposits from, lend money to, act and generally engage in any kind of business with the Borrower and its Subsidiaries, and any Person that may do business with the Borrower or any of its Subsidiaries, all as if the Agent were not the Agent hereunder and without any duty to account therefor to the Lenders.
- Lender Credit Decision. Each Lender acknowledges that it has, independently and without reliance upon the Agent or any other Lender and based upon such documents, information and investigation into the business, prospects, operations, property, financial and other condition and creditworthiness of the Borrower as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement and the other Loan Documents, as applicable. Each Lender also acknowledges that it will, independently and without reliance upon the Agent or any other Lender and based upon such documents, information and investigation as it shall deem appropriate at the time, continue to make its own credit analysis and decisions with respect to the taking or not taking of any action under this Agreement or the other Loan Documents. Except for notices, reports and other documents expressly required by this Agreement or any other Loan Document to be furnished to the Lenders by the Agent, the Agent shall have no duty or responsibility to provide any Lender with any credit or other information concerning the business, prospects, operation, financial or other condition or creditworthiness of the Borrower or any of its Subsidiaries that may come into the possession of the

Agent or any of its Affiliates or any officer, director, employee, agent or attorney-in-fact of the Agent or any of its Affiliates (collectively (with the Agent), the "Agent-Related Persons").

- 8.5. Indemnification by Lenders. The Lenders agree to indemnify the Agent and each other Agent-Related Person (to the extent not reimbursed by the Borrower and without limiting the obligation of the Borrower to do so), ratably according to their respective Percentage Interests, from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever that may be imposed on, incurred by or asserted against such Agent-Related Person in any way relating to or arising out of (a) this Agreement or any other Loan Document or (b) any action taken or omitted by or on behalf of the Agent under this Agreement or any other Loan Document or otherwise in connection with the making, funding, administration or enforcement of the Loans or the preservation of any right in respect thereof, except as a result of such Agent-Related Person's gross negligence or willful misconduct. Without limiting the generality of the foregoing, each Lender agrees to reimburse the Agent promptly on demand, in proportion to its Percentage Interest, for any out-of-pocket expenses, including legal fees, incurred by the Agent in connection with the administration or enforcement of or the preservation of any rights under this Agreement or any other Loan Document (to the extent that the Agent is not reimbursed for such expenses by the Borrower and without limiting the obligation of the Borrower to do so).
- Successor Agent. The Agent may give written notice 8.6. of its resignation at any time to the Lenders and the Borrower, and may be removed at any time, with cause, by the Majority Lenders. Upon any such resignation or removal, the Majority Lenders shall have the right (with the approval (not to be unreasonably withheld) of the Borrower, unless (i) such successor Agent is the Documentation Agent or a Co-Agent, or (ii) an Event of Default has occurred and is continuing) to appoint a successor Agent, who shall concurrently assume the rights and obligations of the Swing Line Lender hereunder (including the Swing Line Lender's commitment to fund Swing Line Borrowings and its interest in outstanding advances under the Swing Line). If no successor Agent shall have been so appointed by the Majority Lenders and shall have accepted such appointment within thirty (30) days after the Agent's giving of notice of resignation or the Majority Lenders' removal of the Agent, then the Agent may, on behalf of the Lenders, appoint a successor Agent, which shall be a bank organized under the laws of the United States or of any state thereof, or any Affiliate of such bank, having a combined capital and surplus of at least \$500,000,000. Upon the acceptance of any appointment as Agent hereunder by a successor Agent, such successor Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of

the retiring Agent and Swing Line Lender, and the retiring Agent and Swing Line Lender shall be discharged from its duties and obligations under this Agreement. Until such acceptance by a successor Agent and Swing Line Lender, the retiring Agent shall continue as "Agent" and "Swing Line Lender" hereunder. Notwithstanding any retiring Agent's resignation or removal hereunder as Agent, the provisions of this Article VIII shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Agent under this Agreement. Any company into which the Agent may be merged or converted, or with which it may be consolidated, and any company resulting from any merger, conversion or consolidation to which the Agent shall be a party, and any company to which the Agent may sell or transfer all or substantially all of its agency relationships, shall be the successor to the Agent without the execution or filing of any paper or further act, anything herein to the contrary notwithstanding.

8.7. Documentation Agent; Co-Agents. No Lender identified in this Agreement or any other Loan Document as "Documentation Agent" or "Co-Agent" shall have any right, power, obligation, liability, responsibility or duty under this Agreement or any other Loan Document other than those that are applicable to all Lenders as such; and no other Person identified in this Agreement or any other Loan Document as "Documentation Agent" or "Co-Agent" shall have any right, power, obligation, liability, responsibility or duty under this Agreement or any other Loan Document other than those, if any, that are expressly to such Person. Without limiting the generality of the foregoing, no Lender or other Person so identified shall, or be deemed to, have any fiduciary relationship with any Lender. Each Lender acknowledges that it has not relied, and will not rely, on any of the Lenders or other Persons so identified in deciding to enter into this Agreement or any of the other Loan Documents or in taking or not taking action hereunder or thereunder.

ARTICLE IX.

MISCELLANEOUS

9.1. No Waiver; Remedies Cumulative. No failure by the Agent or any Lender to exercise, and no delay in exercising, any right, power or remedy under this Agreement or any other Loan Document shall operate as a waiver thereof, nor shall any single or partial exercise of any right, power or remedy under this Agreement or any other Loan Document preclude any other or further exercise thereof or the exercise of any other right, power, or remedy. The exercise of any right, power, or remedy shall in no event (a) constitute a cure or waiver of any Default Event of Default under this Agreement or any other Loan Document or (b) prejudice the rights of the Agent or any Lender in the exercise of any right hereunder or thereunder. The rights and remedies provided herein

and therein are cumulative and not exclusive of any right or remedy provided by law.

- 9.2. GOVERNING LAW. THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS (EXCEPT, IN THE CASE OF ANY OTHER LOAN DOCUMENT, TO THE EXTENT OTHERWISE PROVIDED THEREIN) SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO NEW YORK CONFLICTS OF LAWS PRINCIPLES.
- Notices. All notices and other communications provided for in this Agreement, or that any party shall desire to deliver to any other party with respect to the transactions contemplated hereby, shall be in writing or (unless otherwise specified) by telex, facsimile transmission, telegram or cable, and shall be mailed (with first class postage prepaid) or sent or delivered to each party (a) at the address set forth under its name on the signature pages hereof, or (b) at such other address as shall be designated by such party in a written notice to each other party delivered in accordance with the provision of this Section 9.3. Except as otherwise specified: (x) all notices or other communications sent by mail, if duly given, shall be effective three (3) Business Days after deposit into the mails, (y) all notices or other communications sent by a nationally recognized overnight courier service, if duly given, shall be effective one (1) Business Day after delivery to such courier service (with the relevant fees paid or duly provided for), and (z) all other notices or other communications, if duly given or made, shall be effective upon receipt.
- 9.4. Assignment and Participations.
- 9.4.1. Agreement Binding. This Agreement shall be binding upon and inure to the benefit of the parties and their respective Successors and permitted assigns, provided that the Borrower may not assign or otherwise transfer all or any part of its rights or obligations hereunder or under any other Loan Document without the prior written consent of the Agent and each Lender, and any such assignment or transfer purported to be made without such consent shall be ineffective.
- 9.4.2. Sale of Participations.
- (a) Any Lender may at any time sell participation interests in its Loans and Commitment to any other Person, provided that no such participation shall, without the consent of the Borrower, require the Borrower to file a registration statement with the Securities and Exchange Commission or apply to qualify such participation under the securities laws of any jurisdiction. Such sales may be made without the consent of the Agent, the Borrower or any other Lender; provided, however, that (i) the selling Lender shall have provided the Borrower and the Agent with prior written

notice of the proposed sale of any participation interest in any Loan or in such Lender's Commitment; and (ii) the selling Lender retains the right to vote as a Lender hereunder in respect of the interest sold without being bound to (A) obtain the consent of its participant or (B) exercise its rights in accordance with instructions received from its participant, except that the participant's consent may be required for (w) proposed increases in the Total Commitment; (x) proposed changes to the timing or amount of principal payments; (y) proposed changes to the timing, rate(s) or amount(s) of payments of interest or fees; or (z) proposed releases of any Collateral or Guaranties for which the consent of such Lender would be required hereunder.

- (b) All amounts payable by the Borrower hereunder or under any other Loan Document (including, without limitation, amounts payable to any Lender selling a participation under Section 2.7 or Section 2.13), shall be determined as if such Lender had not sold such participation; provided that the Borrower and each Lender hereby acknowledge and agree that, for purposes of Sections 9.5.1 and 9.6, (i) such participation shall give rise to a direct obligation of the Borrower to the participant, and (ii) such participant shall be considered to be a "Lender."
- (c) No Lender shall, as between (i) the Borrower, the Agent or any other Lender and (ii) such Lender, be relieved of any of its obligations hereunder or under any other Loan Document as a result of any such sale of a participation.
- 9.4.3. Assignments. Any Lender may at any time:
- (a) assign to any other Lender or to any Affiliate of any Lender or to a fund managed by the same investment manager as such Lender, or
- (b) after first obtaining the approval of (i) unless the assigning Lender is the Documentation Agent or an Affiliate thereof, the Agent, and (ii) unless any Event of Default has occurred and is continuing, the Borrower (in either case, which approval will not be unreasonably withheld),

assign, to any Eligible Assignee, all or a portion of its rights and obligations under this Agreement (including without limitation all or a portion of its Commitment and the Loans owing to it); provided that no such, assignment shall, without the consent of the Borrower, require the Borrower to file a registration statement with the Securities and Exchange Commission or apply to qualify such participation under the securities laws of any jurisdiction; and provided further that (v) (i) each such assignment shall be of a constant, and not a varying, percentage of the assigning Lender's rights and obligations under this Agreement and other Loan Documents, and the assignment shall cover the same percentage of

such Lender's Commitment and Loans, and (ii) the commitment of the Swing Line Lender to fund Swing Line Borrowings and the Swing Line Lender's interest in outstanding Swing Line Borrowings may not be assigned to any Person other than a Person who concurrently becomes both a successor Agent pursuant to Section 8.6 and the Swing Line Lender, (w) unless the Agent and (unless any Event of Default has occurred and is continuing) the Borrower otherwise consent, the aggregate amount of the Commitment of the assigning Lender being assigned pursuant to each such assignment to any Person other than a Person that is already a Lender (determined as of the date of the Assignment Agreement with respect to such assignment) shall in no event be less than \$5,000,000 and shall be an integral multiple of \$1,000,000, (x) after giving effect to such assignment (unless the assignment is of the assigning Lender's entire remaining interest in the Loans), the aggregate amount of the Commitment retained by the assigning Lender shall in no event be less than \$5,000,000, (y) the parties to each such assignment shall execute and deliver to the Agent (to the extent applicable, for its acceptance and the acceptance of the Borrower), an Assignment Agreement, and (z) the Agent shall receive from the assignor a processing fee of \$3,000. Unless the Agent or (where the Borrower's consent is required) the Borrower gives written notice to the assigning Lender that it objects to the proposed assignment (together with a written explanation of the reasons behind such objection) within ten (10) Business Days following receipt of the assigning Lender's written request for approval of the proposed assignment, the Agent or the Borrower, as the case may be, shall be deemed to have approved such assignment. Upon such execution, delivery, approval and acceptance, and upon the effective date specified in the applicable Assignment Agreement, (x) the assignee thereunder shall be a party hereto and, to the extent that rights and obligations hereunder have been assigned to it pursuant to such Assignment Agreement, have the rights and obligations of a Lender hereunder, and (y) the assigning Lender thereunder shall, to the extent that rights and obligations hereunder have been assigned by it pursuant to such Assignment Agreement, relinquish its rights (other than rights of indemnification) and be released from its obligations under this Agreement (and, if such assignment is of such Lender's entire remaining Commitment, shall cease to be a party hereto).

- 9.4.4. Assignments or Pledges to Federal Reserve Bank or Trustee.
- (a) Any Lender may pledge or assign all or any part of its interest under the Assignments or Pledges to Federal Reserve Bank. Any Loan Documents for security purposes to any Federal Reserve Bank pursuant to Regulation A of the Board of Governors of the Federal Reserve System and any operating circular issued by the pledgee Federal Reserve Bank; provided that (a) no Lender shall, by reason of such pledge or assignment, be relieved of any of its obligations hereunder or under any other Loan Document, and (b) in

no event shall such Federal Reserve Bank be considered to be a "Lender" or be entitled to require the pledging/assigning Lender to take, or omit to take, any action in respect of the Loans.

- (b) Any Lender which is a trust may pledge all or any part of its interest hereunder to its trustee for the benefit of the trust's beneficiaries.
- 9.4.5. Certain Effects of Assignments, Participations and Pledges.
- Upon its receipt of an Assignment Agreement (a) executed by an assigning Lender and an assignee, together with the processing fee referred to in Section 9.4.3 and any withholding documentation required to be delivered pursuant to Section 2.13, the Agent shall (subject to the provisions of Section 9.4.3 relating to the Agent's right of consent to assignments): (i) accept such Assignment Agreement by executing a counterpart thereof, (ii) record the information contained therein in the Agent's records relating to the Loans; and (iii) give prompt notice thereof to the Borrower (including, to the extent that the Borrower's consent thereto is required, the delivery of a counterpart thereof to the Borrower for its execution and return to the Agent for transmittal to the assigning Lender). The Agent shall maintain a copy of each Assignment Agreement delivered to it and accepted as provided herein.
- The assignee of any permitted sale or assignment (b) (including assignments for security and sales of participations) shall have the same rights and benefits against the Borrower and otherwise under the Loan Documents (except however, in the case of sales of participations, the right to grant or withhold consents or otherwise vote in respect thereof and the right to payment of certain increased costs and amounts in respect of withholding taxes, as provided in Section 9.4.2(b)), including the right of setoff, and in the case of any outright assignment (as distinguished from an assignment for security or the sale of a participation) the same obligations in respect thereof, as if such assignee were an original Lender. Except to the extent otherwise required by the context of this Agreement, the word "Lender" where used in this Agreement shall mean and include any holder of a Note originally issued to a Lender hereunder, and each such holder shall be bound by and have the benefits of this Agreement the same as if such holder had been a signatory hereto. Any outright assignment of a Lender's interest hereunder to another Lender made in conformance with the terms of this Section 9.4 shall result in a corresponding adjustment to the selling and purchasing Lenders' Commitments and Percentage Interests. At the request of any assignee of a Lender, the assigning Lender shall, upon the effectiveness of any assignment (or as promptly thereafter as practicable), surrender its Note to the Agent to be exchanged for a new Note(s) in the name(s) of the

assigning Lender (if it retains any interest hereunder) and its assignee, to reflect the Commitment(s) of the assigning Lender and its assignee after giving effect to such assignment.

- 9.4.6. Confidentiality. The Agent, and each of the Lenders, understands that some of the information and documents furnished to it pursuant to this Agreement or the other Loan Documents may be confidential, and agrees that it will keep all non-public information, documents and agreements so furnished to it confidential and will make no disclosure to other Persons of such information or agreements until it shall have become public, except (a) to the extent required in connection with matters involving operations under or enforcement or amendment of the Loan Documents; (b) in accordance with the Agent's or such Lender's obligations under law or regulations or pursuant to subpoenas or other legal process to make information available to governmental agencies and examiners or to others; (c) to any corporate parent or (if such Affiliate is a financial institution) other Affiliate of Agent or such Lender, solely for purposes of the administration or enforcement of the Loans, so long as such parent or other Affiliate agrees to accept such information or agreement subject to the restrictions provided in this Section 9.4.6; (d) to any participant of the Agent or such Lender that agrees to keep such information, documents or agreement confidential in accordance with the restrictions provided in this Section 9.4.6; (e) to the Agent or to any other Lender and such Agent's and such other Lender's respective counsel and other professional advisors so long as such Persons are instructed to keep such information confidential in accordance with the provisions of this Section 9.4.6; (f) to proposed assignees and participants that agree to keep such information, documents or agreements confidential in accordance with the restrictions provided in this Section 9.4.6; or (q) with the prior written consent of the Borrower.
- 9.4.7. Representations of Lenders. Each Lender represents and warrants to the Borrower, the Agent and each of the other Lenders as follows:
- (a) Such Lender is an Eligible Assignee described in clause (a) of the definition of "Eligible Assignee" (and each Guarantor shall be entitled to rely on such representation and warranty);
- (b) Such Lender has experience and expertise in the making of loans such as the Loans; and
- (c) Such Lender will make (or purchase, as the case may be) its Loans for its own account in the ordinary course of its business and without a view to distribution of such Loans within the meaning of the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended, or other federal securities laws

(it being understood that, subject to the provisions of the Loan Documents relating to the transfer of interests thereunder, the disposition of such Loans, or any interests therein, shall at all times remain within such Lender's exclusive control).

9.5. Indemnification by Borrower; Expenses.

Indemnification. Whether or not the 9.5.1. transactions contemplated hereby shall be consummated, the Borrower shall pay, indemnify and hold each Agent-Related Person, and each of the Documentation Agent, each Co-Agent and each Lender and their respective officers, directors, employees, counsel, agents and attorneys-in-fact (each, an "Indemnified Person"), harmless from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, charges, expenses or disbursements (including reasonable attorneys' fees and expenses incurred by the Agent or any of the Lenders), of any kind or nature whatsoever, that may at any time (including at any time following repayment of the Loans and the termination, resignation or replacement of the Agent or the replacement of any Lender) (collectively, "Indemnified Liabilities") be imposed on, incurred by or asserted against any such Indemnified Person in any way arising out of or in connection with or by reason of, the negotiation, performance, enforcement, preparation of a defense of any claim, litigation, investigation or proceeding relating to or arising out of or in connection with (a) this Agreement or any other Loan Document, (b) the High Yield Offering, (c) the transactions contemplated hereby or thereby, (d) the use of any Loan proceeds, or (e) any action taken or omitted by any such Indemnified Person under or in connection with any of the foregoing; provided that (x) the Borrower shall have no obligation hereunder to any Indemnified Person with respect to Indemnified Liabilities arising from the gross negligence or willful misconduct of such Indemnified Person; and (y) no Indemnified Person that is (i) an Initial Purchaser, as defined in the Purchase Agreement dated August 7, 1997 (the "Purchase Agreement") for the purchase and sale of the High Yield Notes, or (ii) an officer, employee or person who controls any Initial Purchaser within the meaning of the Securities Act and the Exchange Act (as such terms are defined in the Purchase Agreement) (any such Person, in its capacity as an Initial Purchaser or an officer, employee or person who so controls any Initial Purchaser (and not in its capacity as a Lender, Agent or Agent-Related Person, Documentation Agent or Co-Agent or an officer, director, employee, counsel, agent or attorney-in-fact of any of the foregoing), a "High Yield Offering Person"), shall be entitled to be indemnified under clause (b) of this Section 9.5.1 for Indemnified Liabilities imposed on, incurred by or asserted against such Indemnified Person solely in its capacity as a High Yield Offering Person, it being the intent that each such Person's rights of indemnity with respect to the High Yield Offering and the purchase and sale of the High Yield Notes, in such Person's capacity as a

High Yield Offering Person, shall arise solely under, and be governed exclusively by, the Purchase Agreement. All amounts owing under this Section 9.5.1 shall be paid promptly upon demand. election of any Indemnified Person, the Borrower shall defend such Indemnified Person in respect of any Indemnified Liabilities using legal counsel selected by the Borrower to such Indemnified Person at the sole cost and expense of the Borrower. Without limiting the generality of the foregoing, in the event that any Indemnified Person becomes involved in any capacity other than in such Person's capacity as a High Offering Person in any action, proceeding or investigation brought by or against any Person in connection with or as a result of the Loans or any matter referred to in this Agreement or any other Loan Document or the transactions contemplated hereby or thereby, the Borrower shall periodically reimburse such Indemnified Person for its legal and other expenses (including the cost of any investigation and preparation) incurred in connection therewith.

- 9.5.2. Expenses. The Borrower shall pay immediately upon demand (a) all reasonable costs, fees and expenses, including reasonable attorneys' fees and expenses, incurred by the Agent in connection with the preparation, review, execution and delivery of, and the exercise of its duties under, this Agreement and any of the other Loan Documents, and the preparation of amendments, supplements and waivers hereunder and thereunder; (b) all reasonable costs, fees and expenses, including reasonable attorneys' fees and expenses, incurred by the Agent or the Lenders in connection with the enforcement, preservation or protection (or attempted enforcement, preservation or protection) of any rights or remedies of the Agent or any of the Lenders under this Agreement or any other Loan Document (including in connection with any "workout" or restructuring relating to this Agreement or any bankruptcy or insolvency case involving Borrower or any of its Subsidiaries or other Affiliates); and (c) all reasonable costs, fees and expenses incurred by Agent and each of the Lenders for appraisals, audits, environmental inspections and reviews, searches and filings in connection with any of the foregoing. As used in this Agreement, the term "reasonable attorneys' fees and expenses" shall include, without limitation, reasonable allocable costs and expenses of the Agent's and the Lenders' in-house legal counsel, outside legal counsel and financial advisors and staff, and "reasonable costs, fees and expenses" shall include, without limitation, reasonable allocable costs, fees and expenses of the Agent's and the Lenders' internal appraisal, audit, environmental and other similar services, and reasonable fees and disbursements of expert witnesses and other outside consultants.
- 9.6. Setoff. In addition to all Liens upon, and rights of setoff against, the monies, securities or other property of the Borrower given to the Agent or any Lender by law, the Agent, and each Lender, shall have a lien upon and a right of setoff against

all monies, securities and other property of the Borrower now or hereafter in the possession of or on deposit with such Person, whether held in a general or special account or deposit or for safekeeping or otherwise, and, subject to any provision of this Agreement to the contrary, every such Lien and right of setoff may be exercised without demand upon or prior notice to the Borrower. Each Person exercising any such right of setoff shall notify the Agent and the Borrower thereof promptly thereafter, but no failure to give any such notice shall affect the validity of the relevant setoff. No Lien or right of setoff shall be deemed to have been waived by any act or conduct on the part of the Agent or any Lender, or by any neglect to exercise such right of setoff or to enforce such Lien, or by any delay in so doing, and every right of setoff and Lien shall continue in full force and effect until such right of setoff or Lien is specifically waived or released by an instrument in writing executed by the Person entitled thereto.

- 9.7. No Third Party Beneficiaries. This Agreement and each of the other Loan Documents are made and entered into for the sole protection and legal benefit of the Borrower, the Agents, the Lenders, the Arrangers and the Indemnified Persons and their respective permitted Successors and assigns, and, except as expressly provided (a) in Section 9.4.2(b) with respect to participants, or (b) in Section 9.4.7(a) with respect to permitted reliance thereon by the Guarantors, no other Person shall be a direct or indirect legal beneficiary of, or have any direct or indirect cause of action or claim in connection with, this Agreement or any of the other Loan Documents.
- 9.8. Severability; Independence of Covenants.
- 9.8.1. Severability. Any provision of this
 Agreement or any other Loan Document that is prohibited or
 unenforceable in any jurisdiction shall, as to such jurisdiction
 only, be ineffective to the extent of such prohibition or
 unenforceability, without (a) invalidating the remaining provisions
 hereof or (b) affecting the validity or enforceability of such
 provision in any other jurisdiction. To the extent permitted by
 applicable law, the parties waive any provision of law which renders
 any provision hereof prohibited or unenforceable in any respect.
- 9.8.2. Independence of Covenants. All covenants hereunder shall be given independent effect so that if a particular action or condition is not permitted by any one or more of such covenants, the fact that it would be permitted by an exception to, or be otherwise within the limitations of, another covenant shall not avoid the occurrence of an Event of Default or Default if such action is taken or such condition exists. Notwithstanding any other provision of any Loan Document, any action or circumstance otherwise permitted under the Loan Documents that requires any waiver, consent or amendment of any documentation governing an issuance of

Subordinated Debt shall be prohibited unless such waiver, consent or amendment is obtained prior to the taking or occurrence thereof.

- 9.9. Survival. The representations, warranties and indemnities of the Borrower in favor of the Agent and the Lenders, and the representations, warranties and indemnities of the Lenders in favor of the Agent, shall survive indefinitely and, without limiting the foregoing, shall survive the execution and delivery of this Agreement and each of the other Loan Documents, the making of any Loan, the expiration of the Total Commitment and the repayment of all Obligations and other amounts due hereunder.
- 9.10. Release. The Borrower and its Subsidiaries on behalf of each of them and for each of their direct and indirect affiliates, parent corporations, subsidiaries, subdivisions, successors, predecessors, shareholders, and assigns, and their present and former officers, directors, legal representatives, employees, agents, and attorneys, and their heirs, executors, administrators, trustee, successors and assigns (collectively, the "Releasors"), hereby release and forever discharge (this "Release") each Lender and the Agent and each of their respective direct and indirect affiliates, parent corporations, subsidiaries, subdivisions, successors, predecessors, shareholders, and assigns, and their present and former officers, directors, legal representatives, employees, agents, and attorneys, and their heirs, executors, administrators, trustees, successors and assigns (collectively, the "Releasees") of and from any and all claims, liabilities, demands, rights, obligations, damages, expenses, attorneys' fees and causes of action whatsoever from the beginning of the world to the date hereof, whether individual, class or derivative in nature, whether at law or in equity, whether based on federal, state or foreign law or right of action, foreseen or unforeseen, mature or unmatured, known or unknown, accrued or not accrued, which Releasors have or had against the Releasees, arising out of or relating to this Agreement or any other Loan Document (the "Released Claims"), and covenants not to institute, maintain, or prosecute any action, claim, suit, proceeding or cause of action of any kind to enforce any of the Released Claims. In any litigation arising from or related to an alleged breach of this Release, this Release may be pleaded as a defense, counterclaim or crossclaim, and shall be admissible into evidence without any foundation testimony whatsoever. The Releasors expressly covenant and agree that this Release shall be binding in all respects upon their respective successors, heirs, assigns and transferees, and shall inure to the benefit of the heirs, successors and assigns of Releasees. Notwithstanding the foregoing, (x) the Agent and the Lenders hereby expressly reserve any rights they have or may have, collectively and individually, in connection with the Wattles Loan and (y) the Borrower hereby reserves its rights to assert any defense it may have with respect to the Wattles Loan in a proceeding brought against the Borrower by the Agent or any Lender, provided that the

Borrower may not assert any such defense or defenses with respect to any of the Obligations.

- 9.11. Limitation on Liability. To the extent permitted by applicable law, no claim may be made by the Borrower, any Lender or any other Person against the Agent or any Agent-Related Person, the Documentation Agent, any Co-Agent or any Lender, or the Affiliates, directors, officers, employees, attorneys or agents of any of them, for any special, indirect, consequential or punitive damages in respect of any claim for breach of contract or any other theory of liability arising out of or related to the transactions contemplated by this Agreement, or any act, omission or event occurring in connection therewith; and the Borrower and each Lender hereby waive, release and agree not to sue upon any claim for any such damages, whether or not accrued and whether or not known or suspected to exist in its favor; provided, however, that if a Lender refuses to fund a Loan and a court of competent jurisdiction finds that such refusal was without justification and in bad faith, such Lender may be liable to the Borrower for the Borrower's reasonable and foreseeable damages resulting from such refusal to fund. any reason any Lender breaches its obligation to make any Loan hereunder, neither the Agent nor any other Lender shall be responsible to the Borrower for any damage or loss by reason thereof, nor shall the Borrower, or any other Lender, be excused from its own performance hereunder.
- 9.12. Entire Agreement; Amendment. This Agreement, together with the Exhibits and Schedules hereto and the letter agreement referred to in Section 2.11.2, comprise the entire agreement of the parties, and may not be amended or modified except by written agreement of the Borrower and the Majority Lenders executed in conformance with the terms of Section 8.1. No provision of this Agreement may be waived except in writing, and then only in the specific instance and for the specific purpose for which given.
- 9.13. WAIVER OF JURY TRIAL. THE PARTIES HERETO (a) WAIVE
 ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE,
 DETERMINE OR DEFEND ANY RIGHTS UNDER OR RELATING TO THIS AGREEMENT
 OR ANY OTHER LOAN DOCUMENT, OR ANY AMENDMENT, INSTRUMENT, DOCUMENT
 OR AGREEMENT DELIVERED OR WHICH MAY IN THE FUTURE BE DELIVERED IN
 CONNECTION HEREWITH OR THEREWITH, OR ARISING FROM ANY RELATIONSHIP
 EXISTING IN CONNECTION WITH THIS AGREEMENT OR ANY OTHER LOAN
 DOCUMENT, AND (b) AGREE THAT (i) ANY SUCH ACTION OR PROCEEDING SHALL
 NOT BE TRIED BEFORE A JURY, AND (ii) ANY PARTY HERETO MAY FILE AN
 ORIGINAL COUNTERPART OR COPY OF THIS AGREEMENT WITH ANY COURT AS
 WRITTEN EVIDENCE OF THE CONSENT OF THE PARTIES HERETO TO THE WAIVER
 OF THEIR RESPECTIVE RIGHTS TO A TRIAL BY JURY.
- 9.14. Headings. The headings of the various provisions of this Agreement are for convenience of reference only, do not constitute a part hereof, and shall not affect the meaning or

construction of any provision hereof.

- 9.15. Counterparts; Telecopied Signatures. This
 Agreement may be executed in any number of counterparts and by
 different parties hereto on separate counterparts, each of which
 when so executed shall be deemed to be an original, and all of which
 taken together shall constitute one and the same Agreement. This
 Agreement may be executed and delivered by telecopier or other
 facsimile transmission all with the same force and effect as if the
 same were a fully executed and delivered original manual
 counterpart.
- 9.16. FORUM SELECTION AND CONSENT TO JURISDICTION. THE FULLEST EXTENT PERMITTED UNDER APPLICABLE LAW, ANY LITIGATION BASED HEREON, OR ARISING OUT OF, UNDER, OR IN CONNECTION WITH, THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, OR ANY COURSE OF CONDUCT, COURSE OF DEALING, STATEMENTS (WHETHER ORAL OR WRITTEN) OR ACTIONS OF THE AGENT, THE LENDERS, OR THE BORROWER SHALL BE BROUGHT AND MAINTAINED IN THE COURTS OF THE STATE OF NEW YORK OR IN THE UNITED STATES DISTRICT COURT FOR SOUTHERN COURT OF NEW YORK; PROVIDED, HOWEVER, THAT ANY SUIT SEEKING ENFORCEMENT AGAINST ANY COLLATERAL OR OTHER PROPERTY MAY BE BROUGHT, AT THE AGENT'S OPTION, IN THE COURTS OF ANY JURISDICTION WHERE SUCH COLLATERAL OR OTHER PROPERTY MAY BE FOUND. THE BORROWER HEREBY EXPRESSLY AND IRREVOCABLY SUBMITS TO THE JURISDICTION OF THE COURTS OF THE STATE OF NEW YORK FOR THE PURPOSE OF ANY SUCH LITIGATION AS SET FORTH ABOVE AND IRREVOCABLY AGREES TO BE BOUND BY ANY JUDGMENT RENDERED THEREBY IN CONNECTION WITH SUCH LITIGATION. THE BORROWER IRREVOCABLY CONSENTS TO THE SERVICE OF PROCESS BY CERTIFIED MAIL, POSTAGE PREPAID, OR BY PERSONAL SERVICE WITHIN OR WITHOUT THE STATE OF NEW YORK. THE BORROWER HEREBY EXPRESSLY AND IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY OBJECTION WHICH IT MAY HAVE OR HEREAFTER MAY HAVE TO THE LAYING OF VENUE OF ANY SUCH LITIGATION BROUGHT IN ANY SUCH COURT REFERRED TO ABOVE AND ANY CLAIM THAT ANY SUCH LITIGATION HAS BEEN BROUGHT IN AN INCONVENIENT FORUM. TO THE EXTENT THAT THE BORROWER HAS OR HEREAFTER MAY ACQUIRE ANY IMMUNITY FROM JURISDICTION OF ANY COURT OF FROM ANY LEGAL PROCESS (WHETHER THROUGH SERVICE OR NOTICE, ATTACHMENT PRIOR TO JUDGMENT, ATTACHMENT IN AID OF EXECUTION OR OTHERWISE) WITH RESPECT TO ITSELF OR ITS PROPERTY, TO THE FULLEST EXTENT PERMITTED UNDER APPLICABLE LAW, THE BORROWER HEREBY IRREVOCABLY WAIVES SUCH IMMUNITY IN RESPECT OF ITS OBLIGATIONS UNDER THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their respective officers or agents thereunto duly authorized as of the date first written above.

BORROWER: HOLLYWOOD ENTERTAINMENT CORPORATION By: Name: Title: Address: 9275 S.W. Peyton Lane Wilsonville, Oregon 97070 Attn: Donald J. Ekman Telephone: (503) 570-1600 Facsimile: (503) 570-1680 AGENT: SOCIETE GENERALE, NEW YORK BRANCH By: Name: Title: Address: 1221 Avenue of the Americas New York, New York 10020 Attention: Elaine Khalil Telephone: (212) 278-6852 Facsimile: (212) 278-7997 LENDERS: SOCIETE GENERALE, NEW YORK BRANCH

By:
Name:
Title:

GOLDMAN SACHS CREDIT PARTNERS L.P., as Documentation Agent and as a Lender

By:

Title:
CREDIT LYONNAIS LOS ANGELES BRANCH, as a Co-Agent and a Lender
By: Name: Title:
DEUTSCHE BANK AG, NEW YORK BRANCH AND/OR CAYMAN ISLANDS BRANCH, as a Co-Agent and a Lender
By: Name: Title:
By: Name: Title:
KEYBANK NATIONAL ASSOCIATION, as a Co-Agent and a Lender
By: Name: Title:
U.S. BANK NATIONAL ASSOCIATION
By: Name: Title:

THE SUMITOMO BANK, LIMITED

Name:

By: Name: Title:
By: Name: Title:
UNION BANK OF CALIFORNIA, N.A.
By: Name: Title:
VAN KAMPEN PRIME RATE INCOME TRUST
By: Van Kampen Investment Advisory Corp.
By: Name: Title:
THE BANK OF NOVA SCOTIA
By: Name: Title:
By: Name: Title:
THE FUJI BANK, LIMITED

By:

Name: Title:
THE MITSUBISHI TRUST AND BANKING CORPORATION
By: Name: Title:
SUNTRUST BANK CENTRAL FLORIDA, N.A.
By: Name: Title:
TRANSAMERICA BUSINESS CREDIT CORPORATION
By: Name: Title:
CITY NATIONAL BANK
By: Name: Title:
BANQUE WORMS CAPITAL CORPORATION
<pre>By: Name: Title:</pre>

By: Name: Title:
GENERAL ELECTRIC CAPITAL CORPORATION
By: Name: Title:
FLEET BUSINESS CREDIT CORPORATION
By: Name: Title:
FLEET NATIONAL BANK (FORMERLY KNOWN AS BANKBOSTON, N.A.)
By: Name: Title:
THE TORONTO-DOMINION BANK
By: Name: Title:
BLACK DIAMOND INTERNATIONAL FUNDING, LTD
<pre>By: Name: Title:</pre>

```
Lessor/Assignor * Lease # * Effective Date * Maturity Date * Original
Cost * Lease Payment * Balance @ 6/30/00
CIT (KeyCorp Leasing) * 2/23/00-01 * 03/01/2000 * 02/28/2003 * 2,566,242
* 62,323 * 2,310,921
GE Capital (KeyCorp Leasing) * 2/23/00-02 * 03/01/2000 * 02/28/2003 *
5,434,010 * 131,968 * 5,038,104
Safeco Credit (KeyCorp Leasing) * 2/23/00-03 * 03/01/2000 * 02/28/2003 *
2,023,759 * 61,147 * 1,876,315
KeyCorp Leasing * 2/23/00-04 *
                                       03/01/2000 * 02/28/2003 * 2,517,851 *
49,148 * 2,370,406
TransAmerica * 1173-053-01 * 12/16/1999 * 12/16/2002 * 4,152,294* 100,841
* 3,547,246
TransAmerica * 1173-053-02 * 12/16/1999 * 12/16/2002 * 2,711,596 * 65,853
*2,382,331
TransAmerica * 1173-053-03 * 12/16/1999 * 12/16/2002 * 3,109,311 * 75,511
* 2,731,751
TransAmerica * 1173-053-04 * 12/23/1999 * 12/23/2002 * 2,956,308 * 71,796
 * 2,597,328
TransAmerica * 1173-053-05 * 12/23/1999 * 12/23/2002 * 2,555,372 * 62,059
* 2,245,077
TransAmerica * 1173-053-06 * 12/23/1999 * 12/23/2002 * 1,802,452 * 43,774
* 1,583,583
TransAmerica * 1173-053-07 * 12/23/1999 * 12/23/2002 * 1,608,012 * 39,051
* 1,412,753
TransAmerica * 1173-053-08 * 12/23/1999 * 12/23/2002 * 2,162,462 * 52,516
* 1,850,719
TransAmerica * 1173-053-09 * 12/23/1999 * 12/23/2002 * 1,677,130 * 40,730
* 1,473,478
TransAmerica * 1173-053-10 * 12/23/1999 * 12/23/2002 * 665,669 * 16,166 *
584,838
TransAmerica * 1173-053-11 * 12/23/1999 * 12/23/2002 * 2,444,961 * 59,377
* 2,148,073
TransAmerica * 1173-053-12 * 12/23/1999 * 12/23/2002 * 1,168,934 * 28,388
* 1,026,992
TransAmerica * 1173-053-13 * 12/23/1999 * 12/23/2002 * 684,628 * 16,626 *
601,495
TransAmerica * 1173-053-14 * 12/23/1999 * 12/23/2002 * 2,301,227 * 55,886
* 2,021,792
TransAmerica * 1173-016-01 * 01/01/2000 * 12/01/2003 * 10,311,987 *
186,493 * 9,193,027
Heller Financial * 1910092-09 * 03/01/1997 * 11/01/2000 * 5,058,350 *
121,578 * 1,212,720
```

<TABLE>

3.1.4.

<C>

Capital Leases

Heller Financial * 1910092-10 * 03/01/1997 * 11/01/2000 * 2,547,260 * 61,224 * 610,696 Heller Financial * 1910092-11 * 03/01/1997 * 11/01/2000 * 2,395,374 * 57,574 * 574,284 Vision Information Services * 05/15/1998 * 07/01/1998 * 06/01/2001 * 270,254 * 9,694 * 98,696 Creekridge Capital (CFC Funding) * 6/8/98-02 * 03/01/1999 * 02/28/2002 * 95,424 * 3,079 * 57,437

Total Balance at 6/30/00 was \$49,550,062.

</TABLE>

4.17. Subsidiary Assets

As of May 31, 2000, the assets held by Hollywood Management Company were approximately as follows:

Cash (as permitted per Section 6.6.4) \$ (14,822,804)
Receivables 5,794,898
Prepaid Marketing Expenses 6,992,349
Furniture, Fixture and Equipment 28,314,048

Total Assets \$ 26,278,491

6.2(a). Funded Debt

Indebtedness under the High Yield Notes, as of the
Closing Date \$ 250,000,000
Indebtedness under the Credit Agreement, as of
June 30, 2000 289,000,000

<TABLE>

<S> <C>

6.2(d). Capital Leases

Lessor/Assignor * Lease # * Effective Date * Maturity Date * Original Cost * Lease Payment * Balance @ 6/30/00

```
CIT (KeyCorp Leasing) * 2/23/00-01 * 03/01/2000 * 02/28/2003 * 2,566,242
* 62,323 * 2,310,921
GE Capital (KeyCorp Leasing) * 2/23/00-02 * 03/01/2000 * 02/28/2003 *
5,434,010 * 131,968 * 5,038,104
Safeco Credit (KeyCorp Leasing) * 2/23/00-03 * 03/01/2000 * 02/28/2003 *
2,023,759 * 61,147 * 1,876,315
KeyCorp Leasing * 2/23/00-04 *
                                      03/01/2000 * 02/28/2003 * 2,517,851 *
49,148 * 2,370,406
TransAmerica * 1173-053-01 * 12/16/1999 * 12/16/2002 * 4,152,294* 100,841
* 3,547,246
TransAmerica * 1173-053-02 * 12/16/1999 * 12/16/2002 * 2,711,596 * 65,853
*2,382,331
TransAmerica * 1173-053-03 * 12/16/1999 * 12/16/2002 * 3,109,311 * 75,511
* 2,731,751
TransAmerica * 1173-053-04 * 12/23/1999 * 12/23/2002 * 2,956,308 * 71,796
 * 2,597,328
TransAmerica * 1173-053-05 * 12/23/1999 * 12/23/2002 * 2,555,372 * 62,059
* 2,245,077
TransAmerica * 1173-053-06 * 12/23/1999 * 12/23/2002 * 1,802,452 * 43,774
* 1,583,583
TransAmerica * 1173-053-07 * 12/23/1999 * 12/23/2002 * 1,608,012 * 39,051
* 1,412,753
TransAmerica * 1173-053-08 * 12/23/1999 * 12/23/2002 * 2,162,462 * 52,516
* 1,850,719
TransAmerica * 1173-053-09 * 12/23/1999 * 12/23/2002 * 1,677,130 * 40,730
* 1,473,478
TransAmerica * 1173-053-10 * 12/23/1999 * 12/23/2002 * 665,669 * 16,166 *
584,838
TransAmerica * 1173-053-11 * 12/23/1999 * 12/23/2002 * 2,444,961 * 59,377
* 2,148,073
TransAmerica * 1173-053-12 * 12/23/1999 * 12/23/2002 * 1,168,934 * 28,388
* 1,026,992
TransAmerica * 1173-053-13 * 12/23/1999 * 12/23/2002 * 684,628 * 16,626 *
601,495
TransAmerica * 1173-053-14 * 12/23/1999 * 12/23/2002 * 2,301,227 * 55,886
* 2,021,792
TransAmerica * 1173-016-01 * 01/01/2000 * 12/01/2003 * 10,311,987 *
186,493 * 9,193,027
Heller Financial * 1910092-09 * 03/01/1997 * 11/01/2000 * 5,058,350 *
121,578 * 1,212,720
Heller Financial * 1910092-10 * 03/01/1997 * 11/01/2000 * 2,547,260 *
61,224 * 610,696
Heller Financial * 1910092-11 * 03/01/1997 * 11/01/2000 * 2,395,374 *
57,574 * 574,284
Vision Information Services * 05/15/1998 * 07/01/1998 * 06/01/2001 *
270,254 * 9,694 * 98,696
Creekridge Capital (CFC Funding) * 6/8/98-02 * 03/01/1999 * 02/28/2002 *
95,424 * 3,079 * 57,437
```

Total Balance at 6/30/00 was \$49,550,062. </TABLE>

A. VESTING OF STOCK OPTIONS

- 1. Executive Vice Presidents, the President and the Chief Executive Officers
- -Unvested options will vest 100% upon a Change of Control
- 2. Senior Vice Presidents
- -Unvested options will vest 25% upon Change of Control

B. SEVERANCE PAYMENTS

Senior Vice Presidents, Executive Vice Presidents, the President and the Chief Executive Officer shall receive 2 years of base pay together with bonus and medical premiums if within 2 years of the Change of Control the (i) officer's employment is terminated for any reason, other than for Cause (as defined herein) or (ii) the officer resigns for Good Reason (as defined herein).

C. OFFICERS LOANS

Outstanding loans from the Company to Executive Vice Presidents, the President and the Chief Executive Officer shall be forgiven upon a Change of Control.

D. DEFINITIONS

- "Change of Control" shall mean: (Subject to revisions by Company's outside counsel)
- (1) The shareholders of Company approve of the following:

("Approved Transactions"):

(i) Any merger or statutory plan of exchange involving Company ("Merger") in which Company is not the continuing or surviving corporation or pursuant to which Common Stock would be converted into cash, securities or other property, other than a Merger involving Company in which the holders

of Common Stock immediately prior to the Merger have the same proportionate ownership of Common Stock of the surviving corporation after the Merger; or

- (ii) Any sale, lease, exchange, or other transfer (in one transaction or a series of related transactions) of all or substantially all of the assets of Company or the adoption of any plan or proposal for liquidation or dissolution;
- Common Stock (or securities convertible into
 Common Stock) and such offer results in a portion
 of those securities being purchased and the
 offeror after the consummation of the offer is
 the beneficial owner (as determined pursuant to
 Section 13(d) of the Securities Exchange Act of
 1934, as amended (the "Exchange Act")), directly
 or indirectly, of securities representing at
 least 20 percent of the voting power of
 outstanding securities of Company;
- (3) Company receives a report on Schedule 13D of the Exchange Act reporting the beneficial ownership by any person of securities representing 20 percent or more of the voting power of outstanding securities of Company, except that if such receipt shall occur during a tender offer or exchange offer described in (2) above, a Chang of Control shall not take place until the conclusion of such offer; or
- individuals who at the beginning of such period constituted a majority of the Board of Directors cease for any reason to constitute a majority thereof unless the nomination or election of such new directors (each an "Approved Director") was approved by a vote of at least two-thirds of the directors then still in office who were directors at the beginning of such period or who are Approved Directors.

"Cause" shall mean termination from the Company because of (1) the officer's conviction for, or guilty plea to, a felony or a crime involving moral turpitude; (2) the officer's commission of an act of substantial personal dishonesty which is monetarily injurious to the Company in connection with his employment by the Company; (3) the

officer's willful and repeated failure, after written notice, to follow the written policies of the Company as established by the board of directors of the Company, and as applicable to all employees of the Company.

"Good Reason" shall mean the resignation of the officer after (1) notice in writing is given to him/her of his or her relocation, without the officer's consent, to a place of business more than 25 miles outside of the Wilsonville, Oregon area, (2) the actual relocation of offices for the Company more than 25 miles outside of Wilsonville, Oregon, (3) a reduction in the officer's base pay, or (4) a substantial alteration occurs in the nature or status of the officer's responsibilities from those in effect on the date of the Change of Control, disregarding changes in title.

Exhibit A

<ARTICLE> 5

<LEGEND> THIS SCHEDULE CONTAINS SUMMARY FINANCIAL INFORMATION EXTRACTED FROM
THE CONSOLIDATED FINANCIAL STATEMENTS FOR THE NINE MONTHS ENDED SEPTEMBER 30,
2000.

<MULTIPLIER> 1,000

<\$>	<c></c>
<period-type></period-type>	3-MOS
<period-start></period-start>	Jul-01-2000
<fiscal-year-end></fiscal-year-end>	Dec-31-2000
<period-end></period-end>	Sep-30-2000
<cash></cash>	10,089
<securities></securities>	0
<receivables></receivables>	41,329
<allowances></allowances>	0
<inventory></inventory>	74,198
<current-assets></current-assets>	140,042
<pp&e></pp&e>	590 , 525
<pre><depreciation></depreciation></pre>	191,813
<total-assets></total-assets>	1,119,836
<current-liabilities></current-liabilities>	407,059
<bonds></bonds>	250,000
<preferred-mandatory></preferred-mandatory>	0
<preferred></preferred>	0
<common></common>	365,440
<other-se></other-se>	(132,269)
<total-liability-and-equity></total-liability-and-equity>	1,119,836
<sales></sales>	306,532
<total-revenues></total-revenues>	306,532
<cgs></cgs>	114,449
<total-costs></total-costs>	288,093
<other-expenses></other-expenses>	0
<loss-provision></loss-provision>	0
<interest-expense></interest-expense>	17,235
<income-pretax></income-pretax>	1,204
<income-tax></income-tax>	488
<pre><income-continuing></income-continuing></pre>	716
<discontinued></discontinued>	0
<extraordinary></extraordinary>	0
<changes></changes>	0
<net-income></net-income>	716
<eps-basic></eps-basic>	0.02
<eps-diluted></eps-diluted>	0.02